## **General Disclaimer**

## One or more of the Following Statements may affect this Document

- This document has been reproduced from the best copy furnished by the organizational source. It is being released in the interest of making available as much information as possible.
- This document may contain data, which exceeds the sheet parameters. It was furnished in this condition by the organizational source and is the best copy available.
- This document may contain tone-on-tone or color graphs, charts and/or pictures, which have been reproduced in black and white.
- This document is paginated as submitted by the original source.
- Portions of this document are not fully legible due to the historical nature of some of the material. However, it is the best reproduction available from the original submission.

Produced by the NASA Center for Aerospace Information (CASI)



NASA CR-72598 GA-9480



## FINAL REPORT REEVALUATION AND ANALYSIS OF NEUTRON SPECTRA IN LIQUID HYDROGEN

by

G. D. Trimble and W. E. Selph

### GULF GENERAL ATOMIC

prepared for

### NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

NASA Lewis Research Center Contract NAS 3-11228 R. L. Danilowicz, Project Manager

(THRU) (OODE) CATEGORY)

ACILITY

NASA CR-72598 GA-9480 ente atatu alitati en la diferencia.

and a the fighter of the state of the state of the fighter of the state of the stat

#### FINAL REPORT

· • •

## REEVALUATION AND ANALYSIS OF NEUTRON SPECTRA IN LIQUID HYDROGEN

by

G. D. Trimble and W. E. Selph

GULF GENERAL ATOMIC P. O. Box 608 San Diego, California 92112

prepared for

### NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

August 4, 1969

CONTRACT NAS 3-11228

NASA Lewis Research Center Cleveland, Ohio R. L. Danilowicz, Project Manager Chemical and Nuclear Rocket Procurement Section

## CONTENTS

1.	INTR	ODUCTION	•	•	•	•	1
2.	THE	DRETICAL ANALYSIS OF THE FAST AND					
	INTE	RMEDIATE FLUXES	•	•	•	•	4
	2.1	Introduction	•	•	•	•	4
	2.2	Description of the O5R Problem	٠	•	•	•	4
	2.3	Analysis of the O5R Collision Histories .	•	•	•	•	16
	2.4	Results of the Fast and Intermediate Energ	У				
		Neutron Calculation	•	•	•	•	23
	2.5	Description of the Thermal Neutron Calcul	atio	ns	•	•	24
3.	REEV	VALUATION OF THE FAST NEUTRON SPEC	CTR	AL	ı		
	DATA	A	•	•	٠	•	31
	3.1	Reasons for Reevaluation of Fast Neutron					
		Spectral Data	•	•			31
	3.2	Factors which Influence Reduction of the F	ast				_
		Neutron Spectral Data		•			31
	3.3	Other Factors which Influence Reduction of	i Fas	st			
		Neutron Data • • • • • • • • • •	•	•	•	•	50
٨	סדדי	AL HATION OF THE INTERMEDIATE NEIL	ጥወር	N			
ч.	SDEC	TRAL DATA	Inc			_	66
			•	•	•	•	00
	4.1	Introduction	•	•	٠	•	66
	4.2	Intermediate Neutron Energy Detector Effi	cien	су	•	•	66
	4.3	Other Factors which Influence Reduction of	í.				
		Intermediate Neutron Energy Data	•	•	•	•	66
5.	REE	VALUATION OF THE THERMAL NEUTRON					
	SPEC	CTRAL DATA	•	•	٠	•	71
	5.1	Introduction	•	•			71
	5.2	Thermal Neutron Energy Detector Efficien	cv	•			71
	5.3	Other Factors that Influence Reduction of	,				
		Thermal Neutron Data $\cdot$ · · · · ·	•	•	•	•	71
6		USSION OF COMPARISON OF CALCULATION	NS				
0.		FYPERIMENT					77
	AND		•	•	•	•	11

Page

# Page

6.1	M	onitor d Cal	s and	Nor	mali	zati	on o	f Ex	per	ime	ents	3				77
6.2	: Di	scuss	ion of	E Ree	valu	ated	Dat	ta ar	nd C	alc	ula	tio	ns	•	•	78
REFERENC	ES		•••	•	••	•	•	•••	•	•	•		•	•	•	112
APPENDIX .	A - C E	GEOM BETW	ETRY EEN '	' INF THE	UT SOU	TO C RCE	05R. 2 AN	(13 ID C	INC OL	HE LIN	S C 1A]	OF : COR	LH ()	2.	•	A-1
APPENDIX	B - I	KINNY	PRO	GRA	M L	ISTI	NG	• •	•	٠	•	•	•	•	٠	B-1
APPENDIX	c - s	UBRC	UTIN	ie so	OURC	CE I	lst	ING	•	•	•	•	•	•	•	C-1
APPENDIX	D - A	ACTIF	K US	ER R	.OUT	INE	s	• •	•	•	•	•	•	•	•	D-1
APPENDIX	E - J N	TABUI MEASU	LATE JREM	D DA	ATA	OF '	THE	SP	EC:	rr/	AL	•	•	•	•	E-1

## ILLUSTRATIONS

Figure		Pag	e
1	O5R geometry	•	5
2	Hemisphere-integrated flux spectrum		9
3	Cumulative distribution for neutrons	. 13	1
4	O5R Monte Carlo results for neutrons penetrating the empty liquid hydrogen dewar	. 29	5
5	O5R Monte Carlo results for neutrons penetrating a 2.5-in. liquid hydrogen slab as viewed by a collimated detector	. 26	6
6	O5R Monte Carlo results for neutrons penetrating a 4.5-in. liquid hydrogen slab as viewed by a collimated detector	. 2'	7
7	O5R Monte Carlo results for neutrons penetrating a 13.0-in. liquid hydrogen slab as viewed by a collimated detector	. 28	8
8	Density of neutrons degraded through 1 eV as a function of position along the dewar centerline. The abscissa show the distance from the probe tube base. Curves for different thicknesses of liquid hydrogen in front of the probe base are shown	21	0
٩	Typical proton recoil spectrum (NE-211)	• 50	2
10	Typical Compton electron spectrum	• • •	, 7
11	Light output versus energy	• 20	י ב
12	Efficiency for 2 x 2.5-in. detector biased at 0.02746	• 57	, ,
13	Comparison of the detector efficiencies used to reduce the data of 7-31-64 on the first and present contracts	. 43	3
14	Efficiency for the 5 x 5-in. vertical detector biased at 0.032 standard light units	. 44	ł
15	Efficiency of the 5 x 5-in. vertical detector by two different methods	. 45	5

## ILLUSTRATIONS (Continued)

.

n Bhail Phan ann an S

Figure		Page
16	Efficiency of the 5 x 5-in. horizontal detector biased at 0.040 "cobalt units"	. 51
17	Efficiency for the 2 x 2.5-in. detector biased at 0.04233 standard light units	. 52
18	Comparison of the fast neutron spectra measured using two different detectors for the water-cooled source	. 53
19	Comparison of neutron spectra from the air-cooled and water-cooled fast neutron source for the first and second programs	. 54
20	Comparison of neutron spectra from the air-cooled and water-cooled fast neutron source	. 59
21	Fast neutron source spectrum measured with and without gamma filter in flight path	. 61
22	Comparison of the spectral data for the 2.5-in. thickness of $LH_2$ at 37 <sup>°</sup> taken under the first and second programs	. 63
23	Comparison of the spectral data for the 4.5-in. thickness of $LH_2$ at 37 <sup>0</sup> taken under the first and second programs	. 64
24	Comparison of the spectral data for the 10.5-in. thickness of LH <sub>2</sub> at 37 <sup>°</sup> taken under the first and second programs	. 65
25	Efficiency of the intermediate neutron energy detector .	. 67
26	Efficiency of the thermal neutron detector	. 72
27	Mean emission time used to reduce the thermal $LH_2$ data $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	. 73
28	Fast neutron spectrum for a 2.5-in. thickness of $LH_2$ at 0°	. 79
29	Fast neutron spectrum for a 4.5-in. thickness of $LH_2$ at 0°	. 80
30	Fast neutron spectrum for a 7.0-in. thickness of $LH_2$ at 0 <sup>o</sup>	. 81
31	Fast neutron spectrum for a 13.0-in. thickness of LH <sub>2</sub> at $0^{\circ}$	. 82

indiministration (NUM) have also field at property.

## ILLUSTRATIONS (Continued)

Figure		Page
32	Intermediate neutron spectrum for a 2.5-in. thickness of $LH_2$ at 5°	• 83
33	Intermediate neutron spectrum for a 4.5-in. thickness of $LH_2$ at 5°	• 84
34	Intermediate neutron spectrum for $3, 0$ -in. thickness of LH <sub>2</sub> at 5° · · · · · · · · · · · · · · · · · ·	• 85
35	Intermediate neutron spectrum for a 10.5-in. thickness of $LH_2$ at 5°	. 86
36	Intermediate neutron spectrum for a 13.0-in. thickness of $LH_2$ at 5°	• 87
37	Neutron spectrum for a 2.5-in. thickness of $LH_2$ at 15 <sup>°</sup> .	• 88
38	Neutron spectrum for a 4.5-in. thickness of $LH_2$ at 15°.	. 89
39	Neutron spectrum for a 7.0-in. thickness of $LH_2$ at 15°.	. 90
40	Neutron spectrum for a 10.5-in. thickness of $LH_2$ at 15 <sup>o</sup>	. 91
41	Neutron spectrum for a 13.0-in. thickness of $LH_2$ at $15^{\circ}$	. 92
42	Neutron spectrum for a 2.5-in. thickness of LH $_2$ at 37 $^{\circ}$ .	• 93
43	Neutron spectrum for a 4.5-in. thickness of $LH_2$ at 37 <sup>°</sup> .	• 94
44	Neutron spectrum for a 7.0-in. thickness of $LH_2$ at $37^{\circ}$ .	• 95
45	Neutron spectrum for a 10.5-in. thickness of $LH_2$ at $37^{\circ}$	. 96
46	Neutron spectrum for a 13.0-in. thickness of $LH_2$ at $37^{\circ}$	. 97
47	Neutron spectrum for a 2.5-in. thickness of LH <sub>2</sub> at $53^{\circ}$ .	. 98
48	Neutron spectrum for a 4.5-in. thickness of LH <sub>2</sub> at 53 $^{\circ}$ .	. 99
49	Neutron spectrum for a 7.0-in. thickness of $LH_2$ at 53 <sup>0</sup> .	.100
50	Neutron spectrum for a 10.5-in. thickness of $LH_2$ at 53 <sup>°</sup> .	.101
51	Neutron spectrum for a 13.0-in. thickness of $LH_2$ at 53 <sup>°</sup> .	.102
52	Neutron spectrum for a 2.5-in. thickness of $LH_2$ at 78 <sup>°</sup> .	.103
5 <b>3</b>	Neutron spectrum for a 4.5-in. thickness of $LH_2$ at 78 $^{\circ}$ .	.104

## ILLUSTRATIONS (Continued)

Figure		Page
54	Neutron spectrum for a 7.0-in. thickness of $LH_2$ at 78 $^{\circ}$ .	. 109
55	Neutron spectrum for a 10.5-in. thickness of $LH_2$ at 78 $^{\circ}$ .	. 106
56	Neutron spectrum for a 13.0-in, thickness of LH <sub>0</sub> at $78^{\circ}$ .	. 10

## TABLES

• • • • • • • •

÷

Table				Page
1	Summary of data taken under Contracts NAS3-4214 and NAS3-6217	•	•	3
2	Dewar material compositions	•	•	7
3	Dewar centerline materials (between the source and precollimator)	•	•	8
4	Energy biasing for 13 inch LH <sub>2</sub> analysis	٠	•	12
5	Tabulated efficiency for the $2 \times 2.5$ in. NE-211 liquid scintillator biased at 0.02746 standard light units .	•	•	46
6	Tabulated efficiency for the 5 x 5 in. vertical NE-211 liquid scintillator biased at 0.032 standard light units	•	•	48
7	Tabulated efficiency for the 5 x 5 in. horizontal NE-211 liquid scintillator biased at 0.04 "cobalt units"	•	•	55
8	Tabulated efficiency for the $2 \times 2.5$ in. NE-211 liquid scintillator biased at 0.04233 standard light units .	•	•	56
9	Absolute detector efficiency for the intermediate neutron energy detector		•	68
10	Tabulated efficiency of the thermal neutron energy . detector	•	•	74
11	Normalization factors for the experimental data	•	•	108
1 <b>2</b>	Normalization factors for the calculations	•	•	109

.

#### ABSTRACT

الكمار العواريقي

Under Contracts NAS3-4214 and NAS3-6217, angular neutron spectra were measured from 15 MeV to 0.0005 eV at various angles for liquid hydrogen thicknesses of 2.5, 4.5, 7.0, 10.5, and 13.0-in. Since considerable time had elapsed between the two contracts, it was felt desirable to reevaluate and update all the spectral data in a consistent manner using the latest available techniques.

This report describes those methods and techniques used to reevaluate and update all the spectral data generated under the abovementioned contracts. Calculations were performed using the O5R Monte Carlo and 1DF discrete ordinates codes for comparison with selected spectral measurements.

The reevaluated and updated spectral data and calculations are tabulated in the report.

#### 1. INTRODUCTION

This final report describes the work performed on a research program under Contract NAS3-11228 with the Nuclear Systems Division, NASA-Lewis Research Center.

The purpose of this program was to reevaluate and update all spectral data generated under Contracts NAS3-4214 and NAS3-6217 using the latest available techniques. The methods and techniques whereby the neutron spectral measurements were made, under the above mentioned contracts, have been described in detail in References 1 and 2, and a detailed listing of these measurements is given in Table 1. The data taken were reduced using the best efficiency and background subtraction techniques available at the time. However, since there had been over a fouryear lapse since the data were taken on the first contract and a two-year lapse since the second contract to the initiation of the present program, it was felt desirable to reevaluate and update all the spectral data in a consistent manner using the latest available techniques.

The spectral data were taken in three separate energy regions: fast, intermediate, and thermal. The reason for this separation is that each energy region required a different neutron detector.

This final report has been separated into these same groupings. Section II, describes the calculations, Section III discusses the reevaluation of the fast neutron spectral data, Section IV gives the reevaluation of the intermediate neutron spectral data, Section V discusses the reevaluation of the thermal neutron spectral data, and Section VI compares calculations and experiments for all the data generated under Contracts NAS3-4214 and NAS3-6217. Appendix A contains the geometry input to O5R,

#### 1. INTRODUCTION

This final report describes the work performed on a research program under Contract NAS3-11228 with the Nuclear Systems Division, NASA-Lewis Research Center.

The purpose of this program was to reevaluate and update all spectral data generated under Contracts NAS3-4214 and NAS3-6217 using the latest available techniques. The methods and techniques whereby the neutron spectral measurements were made, under the above mentioned contracts, have been described in detail in References 1 and 2, and a detailed listing of these measurements is given in Table 1. The data taken were reduced using the best efficiency and background subtraction techniques available at the time. However, since there had been over a fouryear lapse since the data were taken on the first contract and a two-year lapse since the second contract to the initiation of the present program, it was felt desirable to reevaluate and update all the spectral data in a consistent manner using the latest available techniques.

The spectral data were taken in three separate energy regions: fast, intermediate, and thermal. The reason for this separation is that each energy region required a different neutron detector.

This final report has been separated into these same groupings. Section II, describes the calculations, Section III discusses the reevaluation of the fast neutron spectral data, Section IV gives the reevaluation of the intermediate neutron spectral data, Section V discusses the reevaluation of the thermal neutron spectral data, and Section VI compares calculations and experiments for all the data generated under Contracts NAS3-4214 and NAS3-6217. Appendix A contains the geometry input to O5R,

Appendix B lists the KINNY program and input instructions, Appendix C gives the subroutine SOURCE listing, Appendix D provides the ACTIFK user subroutines, and Appendix E is a detailed listing of all spectral data generated under both Contracts NAS3-4214 and NAS3-6217, and the O5R and IDF calculations made under the present contract, NAS3-11228.

للمخطيفة ويعودون ويوج الارتجاب الرابط المراجع

4.11

#### Table 1

### SUMMARY OF DATA TAKEN UNDER CONTRACTS NAS3-4214 AND NAS3-6217

#### Contract NAS3-4214

friðskilgan.

A. Fast Neutron Spectrum Measurements

Angular neutron flux from 15 to 0.5 MeV for angles of  $0^{\circ}$ , 15°, 37°, 53°, and 78° and liquid hydrogen thicknesses of 2.5, 4.5, 7.0, 10.5, and 13.0 in. and with the liquid hydrogen dewar empty.

#### Contract NAS3-6217

A. Fast Neutron Spectrum Measurements

Angular neutron flux from 15 to 0.5 MeV at  $0^{\circ}$  for liquid hydrogen thicknesses of 2.5, 7.0, and 13.0 in. and for the empty dewar; and at  $37^{\circ}$  for liquid hydrogen thicknesses of 2.5, 4.5, 10.5 and for the empty dewar.

#### B. Intermediate Neutron Spectrum Measurements

Angular neutron flux from 1.5 MeV to 500 eV for angles of  $5^{\circ}$ ,  $15^{\circ}$ ,  $37^{\circ}$ ,  $53^{\circ}$  and  $78^{\circ}$  and liquid hydrogen thicknesses of 2.5, 4.5, and 7.0 in. Empty dewar measurements were made at  $5^{\circ}$ ,  $15^{\circ}$ ,  $37^{\circ}$ , and  $78^{\circ}$ .

#### C. Thermal Neutron Spectrum Measurements

Angular neutron flux from 10 eV to 0.0005 eV for angles of  $37^{\circ}$  and  $78^{\circ}$  and liquid hydrogen thicknesses of 2.5, 4.5, 7.0, 10.5 and 13.0 in.

#### 2. THEORETICAL ANALYSIS OF THE FAST AND INTERMEDIATE FLUXES

#### 2.1 INTRODUCTION

The Monte Carlo method exemplified by the code O5R<sup>(3)</sup> was selected for the analysis of the liquid hydrogen experiment because of its flexibility in describing the source and material geometry and because of its ability to consider separately the direct and scattered components. This allows the formulation of estimators which approximate the collimated detector.

The experiment was performed for thicknesses of 2.5, 4.5, 7.0, 10.5, and 13 inches of liquid hydrogen between the source and precollimator. For the analysis reported here representative thicknesses of 2.5, 4.5, and 13 inches were selected by mutual agreement with the contracting agency.

Described in Section 2.1 below are the procedures used in the implementation of the O5R Program in calculating collision histories in the fast and intermediate energy region. Analysis of these histories is discussed in Section 2.3 and the results are presented in Section 2.4.

#### 2. 2 DESCRIPTION OF THE O5R PROBLEM

#### 2.2.1 GEOMETRY OF MATERIALS

Figure 1 shows the sector boundaries used in the mathematical description of the dewar and the materials assigned to each of these sectors. This is a plan view of the cylindrically symmetric arrangement. Not all space within the enclosing parallelopiped is accounted for in this projection. Between the outer cylinder, which forms the central part of the dewar, and the corners of the parallelopiped there are four segments of space assigned as external void. The OSR listing of this description of the geometry is given in Appendix A for the 13-inch configuration.





Figure 1. O5R geometry.

The material compositions are presented in Table 2 in terms of atomic density.

Hydrogen vapor is assumed to fill all of the internal compartments of the dewar not filled with liquid hydrogen. The LH<sub>2</sub> vapor at atmospheric pressure and at the boiling point temperature has a density which is 1.57%of the liquid density. A summary of the materials encountered along the dewar centerline in Table 3 shows that the vapor adds an equivalent of .736 cm of LH<sub>2</sub> to the 2.5-inch LH<sub>2</sub> configuration and an equivalent 3.28 cm to the 13-inch LH<sub>2</sub> configuration.

#### 2. 2. 2 CROSS SECTIONS

Since in the dewar geometry there is a total thickness of 1.431 cm of stainless steel between the source and the precollimator it was decided that inelastic events should be treated exactly by the code. This decision greatly complicated the cross section preparation (roughly doubled the amount of cross section input) and the coding required to track particles and analyze collisions. The approach was deemed necessary, however, because the probability of successive collisions was very sensitive to the energy lost in the first collision. The inelastic events are a significant fraction of the total cross section in all of the primary constituents of stainless steel at energies above 3 MeV.

All of the more important cross sections were taken from the ENDF/B files. These include hydrogen, iron, nickel, and chromium. Cross sections for silicon, aluminum, and boron (constituents of the insulation material) were either not available or had insufficient detail in ENDF. Consequently, these were taken from the Gulf General Atomic standard cross section library. In the case of silicon, these cross sections were evaluated at Gulf General Atomic<sup>(4)</sup> for submission to ENDF.

The ENDF cross sections and Legendre coefficients were input to the O5R cross section handling programs to generate the PHI tape used directly in the calculations. Some difficulties were encountered, however,

	Material	$\rho(g/cm^3)$	Atom Density (atoms/cm <sup>3</sup> x 10 <sup>-24</sup> )				
1	Liquid Hydrogen	. 071	H .042425				
2	Stainless Steel	7.9	Fe .05901				
			Ni .008529				
			Cr .01851				
3	Insulation plus	.7063	B $2.78 \times 10^{-4}$				
	Stainless Steel		A1 5.65 x $10^{-4}$				
	Homogenized		Si $3.26 \times 10^{-4}$				
			Fe 4.815 x $10^{-3}$				
			$Cr 1.51 \times 10^{-3}$				
			Ni $6.955 \times 10^{-4}$				
4	Hydrogen Vapor	. 0011147	H $6.6607 \times 10^{-4}$				
N	Note on Material 3: 90 layers of borosilicate glass, each $1.71 \ge 10^{-3} \text{ g/cm}^2$						
		90 layers of Al foil, each	h 2.144 x $10^{-3}$ g/cm <sup>2</sup>				
2 layers of Stainless Steel, each 2.55 g/cm <sup>2</sup> homogenized over a depth of 7.62 cm							

#### Table 2

tores ....

### DEWAR MATERIAL COMPOSITIONS

in the case of the inelastic cross sections. The lowest energy point used in reporting some of the cross sections was at some point above the inelastic threshold scattering rather than being at the threshold. This, coupled with the method of interpolation in O5R occasionally resulted in finite cross sections for an event at energies below the threshold. To overcome this difficulty an auxiliary program was written to redefine the ENDF data at points near the threshold.

An additional complication in the cross section preparation arose because of deficiencies in the ENDF/B data for Ni. There was no scattering

#### Table 3

Dewar Configuration	LH <sub>2</sub> Depth	Vapor Depth (cm)	Equi	iv. LH <sub>2</sub>	Dewar Materials (g/cm <sup>2</sup> )			
Name	(cm)		In.	cm	g/cm <sup>2</sup>	Steel	Insulation	
2.5 in.	6.19	46.99	2.72	6.926	0.491	11.52	0.35	
4.5	11.11	42.07	4.63	11.77	0.831			
10.5	26.03	27.15	10.42	26.456	1.88			
13.0	32.22	20.96	12.8	32.548	2.31	<b>Y</b>		

### DEWAR CENTERLINE MATERIALS (Between the Source and Precollimator)

energy law given for the (n, 2n) reaction and the law given for inelastic scattering was found to be in error. The assistance of Dr. Marvin Drake was obtained to formulate nuclear temperature models for these two reactions. The formulation is the standard ENDF/B scattering energy law<sup>(5)</sup> (LF = 7) with a nuclear temperature given by  $\theta = 422 \text{ E}^{0.5}$  for inelastic scattering and by  $\theta = 302 \text{ E}^{0.5}$  for the (n, 2n) reaction.

The order of anisotropy included in the neutron scattering angular distributions was P9 except for B (P6), A1 (P8), and Si (P8).

#### 2.2.3 SOURCE

The mathematical description of the neutron source was modeled after the 7.62-cm water-cooled depleted uranium target used in the second series of liquid hydrogen experiments.  $^{(2)}$  The hemisphere-integrated spectrum given off by this target is shown in Fig. 2. $^{(6)}$  The spectrum and intensity of the source were assumed invariant with angle within the approximately 65-degree half-angle conical intercept of the dewar.

To reduce the computer running time required to obtain good averages of the scattered flux, the source was biased both in angle and in energy. The energy biasing for the 2.5- and 4.5-inch slabs was obtained by generating neutrons with equal probability in each of the source energy



Figure 2. Hemisphere-integrated flux spectrum.

groups. This should provide approximately equal convergence of each of the groups for a thin shield. The cumulative distribution for neutrons above  $4 \times 10^3$  eV is shown in Fig. 3 for both the unbiased and biased sources. These plots refer only to the relative number of source particles generated. The weights are adjusted so that the weight generated in any group is the same in either the biased or unbiased condition.

In the 13-inch  $LF_2$  configuration an adjoint calculation was performed using the 1DF transport code<sup>(7)</sup> as a means of estimating the optimum source biasing. The solution of the 1DF adjoint problem gives only the approximate importance because of the inadequacies of a onedimensional representation of the experimental geometry. The calculation is nevertheless worthwhile since it can add considerably to the efficiency of the Monte Carlo runs. The hydrogen slabs were represented in 1DF as being infinite and a neutron source was input with an energy distribution which is the inverse of the distribution used in the experiment. The energy distribution of the penetrating radiation is then an indication of the optimum input source for the Monte Carlo calculations. Slight adjustments were made in the O5R energy groups to conform to the 1DF fine group structure boundaries. The result of this calculation indicated use of the cumulative probability and weighting factors given in Table 4.

Source particles below 0.2 MeV were ignored in the O5R calculations of the thick slab because their impact on leakage through the slab would not be significant. Angular biasing was used to force more of the neutrons into the important central portions of the dewar. The extent of the distortion from an isotropic distribution and the associated weighting are summarized below:

Polar	Unbiased	Biased	Weight
Angle	Ρ(θ)	Ρ(θ)	Adjustment
10 <sup>0</sup>	0.02631	0.1	0.2631
20 <sup>0</sup>	0.104454	0.3	0.3907
65 <sup>0</sup>	1.0	1.0	1.278





E Upper (eV)	<b>P(E)</b> <sup>*</sup>	Weight Adjustment Factor
2.0 E5	0.02194	7.34
4.0 E5	0.04127	10.14
6.0 E5	0.06411	5.55
8.0 E5	0.08605	4.44
1.0 E6	0.10797	3.476
1.5 E6	0.15185	2.656
2.0 E6	0.19573	1.255
3.0 <b>E</b> 6	0.25812	1.048
4.0 <b>E</b> 6	0.32211	5.0704 x $10^{-1}$
6.0 E6	0.42274	$3.159 \times 10^{-1}$
8.0 E6	0.52337	9.678 x $10^{-2}$
1.0 E7	0.624	$3.285 \times 10^{-2}$
1.2 E7	0.812	5.185 x $10^{-3}$
1.4 E7	1.0	$2.93 \times 10^{-3}$

## Table 4

### ENERGY BLASING FOR 13 INCH LH, ANALYSIS

\*P(E) is the cumulative probability of neutrons being generated with energy less than, or equal to, the energy of the group upper bound.

The probability allotment for each increment is interpolated linearly with the cosine of the angle within the bounding values. Although neutrons are not generated outside the  $65^{\circ}$  cone the answers are all normalized to one  $4\pi$  isotropic neutron at the source. To prevent roundoff problems at the boundary passing through the source, the initial position of the particles was set on the surface of a 1-cm sphere.

#### 2.2.4 IMPLEMENTATION OF O5R

There are a variety of routines which the O5R user must provide to match his specific problem requirements. Only two of these special routines, KINNY and SOURCE, were required in this calculation although a variety of other minor modifications were made to increase operating efficiency on the GGA Univac 1108 computer.

KINNY is the subroutine which treats nonelastic events. The version used in this analysis treats discrete-level inelastic scattering, unresolved level scattering using an evaporation model, and (n, 2n) and (n, 3n) processes. The program listing and input specifications are given in Appendix B. The KINNY subroutine receives information on the precollision neutron parameters, the species collided with, and the point of collision. It then calculates postcollision parameters based on the collision energetics for the specific collision type. Scattered angular distributions may be either isotropic or anisotropic in the center of mass system in accordance with options selected in the O5R input. Use is made of standard random number routines in selecting the polar and azimuthal scattering angle. When the incident energy exceeds the threshold, a given level may be excited. For this option postcollision parameters are calculated by subroutine LEVEL. Un esolved level scattering is handled in subroutine EVAPMD.

Subroutine SOURCE was written to generate source neutrons at a point from input cumulative distributions in energy and angle. Weighting factors are input corresponding to each bin so that energy and angle biasing may be employed. This listing is given in Appendix C. Input quantities are: Card A (6E10.2) x,  $\gamma$ , z, NMEM, NMED, NREG

- 1-3 x, y, and z coordinates of the source point.
- 4 The number of neutron histories to be generated.
- 5 The number of the medium corresponding to the location of the source point.

6 The number of the region corresponding to the location of the source point.

Card B (2110) MTHET, MEGR

- 1 The maximum number of boundaries of the angle bins used in the problem  $\leq 21$ .
- 2 The maximum number of energy bin boundaries  $\leq 41$ .

#### Card C (7E10.2) THET(I)

The angle bounds beginning with the bound on or nearest the x axis in the negative x direction. Units are in radians.

Card D (7E10.2) PTHET(I)

The cumulative probability of generating a neutron at angles between the x axis (or the bound nearest the x axis) and the bound THET(I). These probability values should be adjusted to reflect the biased condition if angular biasing is used.

Card E (7E10.2) WTHET(I)

The (MTHET-1) weight constants which apply to the polar angle bins. All WTHET(I) = 1 if biasing is not used.

Card F (7E10.2) EBNDS(J)

The (MEGR) bounding values of the energy groups beginning with the lowest energy. Units are eV.

Card G (7E10.2) POFE(J)

The cumulative probability of neutrons being generated with an energy less than EBNDS(J). There are MEGR of these values which always begin with 0 and end with 1.

Card H (7E10.2) EWT(J)

The (MEGR-1) weights associated with the energy bins listed with the lowest energy bin first.

The implementation of this subroutine to obtain particular source biasing conditions is discussed in the preceding section.

In assembling and checking out the O5R program for these calculations it became apparent that reductions in run time would be required if all of the calculations were to be performed within the budgeted computer time. One of the basic constraints was the small amount of core space for storing the parameters of neutrons currently being tracked. The program flow is set up so that neutrons are generated in batches and each batch is traced successively through the energy supergroups until all neutrons are degraded below the minimum energy of interest in the problem. Thus cross section data for only one supergroup has to be in the core. Collision parameters can be stored on tape as generated but the postcollision particle parameters must be maintained in the core for every particle in the batch.

つびり きりまた

Sec. a start

After some mapping of the program it was possible to store the parameters for 1000 particles. Calculations could then be performed in 1000 particles batches if no splitting was allowed but this number would have to be reduced considerably if the splitting option was used because each split resulted in two or more neutrons which must be tracked. The liquid hydrogen calculation was first modeled with the intent of using the splitting and russian roulette options in O5R but it was found that the batch size had to remain so small that no real computing economy could be realized. On a larger computer the variance of the calculated values could have been reduced by judicious use of weight limiting regions. The program was found to be inefficient on the GGA Univac 1108 machine since a significant fraction of the program run time is taken up with input/output (I/0) functions. Most of these were handled by FORTRAN reading to and from tape using a 256-word buffer size. Some of the options that were tried which netted small, but not really significant gains, were: (1) transferring from tape to drum using FORTRAN and a 256-word buffer; and (2) using NTRAN with a 256-word buffer on drum storage. A third option which did result in significant run time savings was changing the buffer size from 256 to 10,000. The O5R PHI and cross section tapes were rewritten (by program modification) to conform with this buffer size and were then transferred to drum. This operation allowed much quicker transfer rate between the tape and drum and between the drum and core.

The result was about a factor of two reduction in run time per history. This allowed about two times as many histories to be run per unit time, or a factor of  $1/\sqrt{2}$  reduction in the uncertainty of the answer.

In larger machines where I/O functions are billed at a much lower rate, and central processor time is billed separately, the savings described above would not be realized. In the Gulf General Atomic UNIVAC machine however, the central processor is tied up during I/O operations and any improvement in these is reflected directly in the run time and computing costs.

NTRAN is a special feature of UNIVAC systems that allows for variable buffer lengths and parallel I/O and computing operations. The variable buffer size is used in the input and the parallel feature is advantageous in writing collision history information while calculations continue.

Checkout of the program consisted of small batch runs where collision parameters and intermediate quantities are printed out to determine that:

- 1. Source particles are being properly selected and weighted
- 2. Cross section values are being correctly determined
- 3. Path length is selected properly.

In addition the PICTURE routine was run to check that the geometry was being interpreted by GEOM as was intended. This proved to be a useful operation in that small regions between the outer cylinder of the dewar and the enclosing paralellopiped which had not been described were later included in the model.

#### 2.3 ANALYSIS OF THE O5R COLLISION HISTORIES

#### 2.3.1 THE ACTIFK PROGRAM

Analysis of the O5R collision tapes was performed using the  $ACTIFK^{(8)}$  code. ACTIFK, as distributed by the radiation shielding information center, will only calculate neutron dose at a point, but by

suitable modification, it can be made to perform a wide variety of analysis tasks. The code package includes the same geometry routine as in the O5R code and is similar to O5R in the provisions for cross section handling. Existing routines perform many of the coding details connected with tape reading, computation of storage requirements for input data, transmission of data to the appropriate subroutines at the proper time, and other tasks.

Modifications of the code were of two basic types: some of the routines were modified to fit the different format for data storage in our modified version of O5R; and subroutines connected with analysis of the histories were removed and replaced in their entirety. Included in the latter category were:

STBATCH - A routine which is called at the beginning of each batch of neutron histories to initialize quantities being calculated. In addition, it reads input associated with the user routines on the first call.

SDATA - A routine which calculates any quantity desired from the source particle parameters. In the LH<sub>2</sub> analysis, SDATA was used to calculate the uncollided flux.

RELCOL - A routine which calculates any desired quantity from real collision parameters. This is the heart of the ACTIFK program as used in this project. All last flight estimators were calculated here as well as the calculation of an input source for the thermal calculations.

ENDBATCH - At the end of each batch of neutrons this routine is called to do any desired calculations such as averaging or normalizations which are more economical to do here rather than including them in the calculation of the contribution from each collision.

ENDRUN - A routine which is called after all collisions have been processed. Final answers and statistical quantities are calculated here and printed out.

Each of these routines may call upon a variety of built-in routines such as the geometry package, routines for calculating attenuation along the flight path and for calculations of total or differential scattering cross sections. The listing of these routines is given in Appendix D.

#### 2.3.2 FLUX ESTIMATORS

The mathematical formulations of the various estimators used in the collision tape analysis are described below.

#### 2.3.2.1 Energy-Angle Flux

In estimating the energy-angle flux, a unit detector was assumed to be positioned at the center of the entrance to the precollimator and an estimate was made of the contribution to the flux at this point resulting from each collision point of interest. It will be recalled from the geometry description that the precollimator begins at the x = 0 plane and extends in the negative x direction. Thus only collisions whose x coordinate was positive were sampled.

The first steps in determining the energy-angle flux estimate are to calculate the distance, R, from the collision point to the detector and the angle of scatter,  $\theta$ , between the precollision particle direction and the collision point-detector axis. The next parameters determined are the energy after scatter for the particular scatter angle and scattering species, the differential angle scattering cross section per steradian about the  $\theta$  direction, and the attenuation,  $e^{-b_1}$ , of the neutron after it scatters. The postcollision attenuation coefficient,  $b_1$ , is simply the integral of the total cross section times the path length along R. The contribution to the flux is then given by

$$\mathbf{F}_{s} = \frac{WT_{1}}{R^{2}} \left( \frac{d\sigma_{s}}{d\Omega} \right)_{\mathbf{E},\theta} e^{-b_{1}}$$

where  $WT_1$  is the weight of the incident particle reduced by  $\sigma_s(E)/\sigma_T(E)$ at the energy of the incident neutron. The contributions to flux at the detector are then sorted into an input energy and angle bin structure. Contributions accumulate in each bin and are averaged and normalized at the end of the batch. Normalizations of the flux in a given bin includes dividing by the number of neutrons processed and multiplying by a source solid angle normalization factor which accounts for the fact that source particles are generated only over a portion of space but normalization is per source particle in any direction. Other normalizations account for the desired units on the calculated quantities. These involved multiplying by  $\overline{E}/\Delta E$  where  $\overline{E}$  is the bin average energy and  $\Delta E$  is the bin width, and dividing by the solid angle increment of the angle bins. Thus, units of the flux were  $E \cdot \Phi(E)$  (eV \* neutrons \* cm<sup>-2</sup> \* ster<sup>-1</sup> \* eV<sup>-1</sup>).

#### 2.3.2.2 Normally Exiting Collided Flux

Special problems are encountered in attempting to analyze time-offlight experiments because of the narrow collimation of the detector. The behavior of the scattered flux with angles greater than a few degrees is generally a smoothly varying function which can be approximated by a reasonable number of angular bins. The scattering flux exiting per unit solid angle near the slab normal, however, is generally a more rapidly varying function. Thus to be sure of analytically duplicating the drift tube angular resolution one would need an angular bin only a few minutes wide. To obtain a sufficient number of collisions inside the material within a cone this narrow would require prohibitive run times. To increase the efficiency for estimating the near normal flux a new type of estimator was formulated and incorporated into the analysis routine.

The geometry of the slab and flight path involved in this estimator is given below.



The assumption is made that the distribution of scattered radiation exiting the slab does not vary significantly for any location on the surface within a radius of 5 cm from the geometry centerline. Each collision lying inside the cylinder formed by the 5-cm radius is then sampled to determine its contribution per unit solid angle to the flux exiting normally through the circle on the surface formed by the 5-cm radius. This exiting flux is then normalized by dividing by the area of the circle. It should be noted that the estimate is not made to a point detector on the surface but rather on a per unit solid angle basis. The assumption made here is that the depth of the shield is negligible with respect to the distance to the detector. Thus the estimator for the normal scattered flux would be

$$F_{ns} = \frac{WT_{1}}{25\pi} \left( \frac{d\sigma_{s}}{d\Omega} \right)_{\theta, E} e^{-b_{1}}$$

where  $\theta$  is the angle between the incident direction and the slab normal,  $b_1$ , is summed along a normally exiting ray, and the other terms are as defined above. If desired, a weighting factor may be used to account for the difference in solid angle intercept of the drift tube detector due to depth within the slab. If the drift tube is 50 meters long and the collision occurs at a depth x from the exit face then the weighting factor would be  $\left(\frac{50}{50+x}\right)^2$ . For a slab depth of 30 cm this factor deviates from 1 by about 1% at most.

#### 2.3.2.3 Special Angle Flux Estimator

The technique described above for determining the normally exiting collided flux also proved useful in estimating the flux at exact angles. Better statistics were obtained using the estimator than using the angular increment approach (2.3.2.1 above) in some cases. In this case collisions were sampled if they occurred inside a region within a 5 cm radius (measured in the x plane) from a line inclined at the angle of interest and passing through the center of the exit face. The collision is sampled only if its x coordinate is positive, as in the previously discussed estimator.

A modified version of ACTIFK was written which incorporated or ''' a  $37^{\circ}$  and  $78^{\circ}$  angular estimator of this type. Scattered fluxes calculated with this program were found to be in agreement with the point detector values at the same angle for high energies and were found to be converged at lower energies than for the alternate estimator. Some insight as to why this trend exists may be obtained from considering the problem of calculating the flux within a given angle increment at the two points A and B as shown below.



For the same angular increment a larger portion of the shield is viewed from point B and therefore more collisions will contribute to the angular flux. This becomes increasingly important at lower energies where collisions lying very near the surface are of importance. A higher collision density would be required to get adequate data at point A because, as one approaches the surface inside the angle increment, the available collision volume disappears rapidly. This is similar to the problem of attempting to estimate the flux at very small exit angles and the solution worked out is similar to the normally exiting estimator.

#### 2.3.2.4 Source for Thermal Calculations

In the O5R program neutron histories were terminated when the energy fell below .075 eV. An analysis program was written to obtain an input source for the  $S_n$  calculation of thermal flux based on the Monte Carlo collision histories. Each collision was tested to determine whether

the energy before collision is greater than  $1 \, \text{eV}$  and whether the energy after collision is below  $1 \, \text{eV}$ . If either of these tests fail the collision is rejected. If both conditions are met the position of the collision and the neutron energy after collision are stored. By summing the postcollision weights of such particles over appropriate increments of space and energy, a source term was formulated for input to the 1DF S<sub>n</sub> transport code. Although this source is distributed in energies below 1 eV the total is equal to the spatial density of particles falling through 1 eV and this is the only source of neutrons at lower energies.

#### 2.3.2.5 Uncollided Flux

For each source particle noted on the O5R collision tape the ACTIFK routine calls SDATA and calculates the uncollided contribution to the flux at the detector centered on the exit face of the LH<sub>2</sub> slabs.

In normalizing the uncollided and the normally exiting collided flux it became apparent that the detector at the end of the drift tube views the exiting flux in a different manner than a unit detector positioned at the entrance to the precollimator (as approximated by the last flight estimator). There are two important factors in the normalization. The first one is that the collimator allows the detector to view only a portion of the source so only that fraction should be considered as the source for the uncollided flux. The second factor is that the scattered flux is calculated on a per unit solid angle basis so that the direct radiation must be in these same units before the two components can be added to get the total normally exiting flux.

In the first case an adjustment factor of 0.0841 was included to account for the portion of the source viewed directly. The second adjustment (due to angularity) may be made by multiplying the normal uncollided estimator by the ratio

 $\frac{R_{S-E}^{2}R_{E-D}^{2}}{2}$ 

where the values of R are separation distances from source to exit face, exit face to detector, and source to detector, respectively.

Thus the uncollided flux estimator

$$\Phi_{\rm u} = \frac{{\rm w}}{4\pi {\rm R}_{\rm X-E}} e^{-{\rm b}_{\rm 1}}; {\rm b}_{\rm 1} = \sum \Sigma_{\rm i} t_{\rm i}$$

is changed to

$$\Phi_{nu} = \frac{\Psi(0.0822)}{4\pi} e^{-b} = W \cdot 6.55 \cdot 10^{-3} \cdot e^{-b} 1$$

The normalized uncollided flux as calculated with this estimator may be added numerically to the normally exiting collided flux to get the total normally exiting flux as seen by a detector at the end of the flight path.

It is important that these normalization be noted by anyone desiring to compare other calculations with these on an absolute basis.

#### 2.3.3 STATISTICS

There is provision in the ACTIFK program for printing onto a "statistical tape" the contribution to each quantity being calculated due to each source neutron. This tape can then be utilized to calculate the standard distribution or variance of the data. This provision was deleted in our analysis because of the added requirements for core storage and run time which it imposes. Instead a variance of individual batch averages about the entire run average was calculated. While this quantity is not a true variance of the data it does serve to indicate the degree of convergence of the batch averages.

#### 2.4 RESULTS OF THE FAST AND INTERMEDIATE ENERGY NEUTRON CALCULATION

Fast and intermediate energy calculations were performed for the empty dewar and for LH<sub>2</sub> thicknesses of 2.5, 4.5 and 13 inches preceding

the precollimator. Ten thousand histories were run for the empty dewar calculation; twenty thousand histories were run for the 2.5- and 4.5-inch configurations; and thirty thousand histories were run for the 13-inch case. First flights were stretched in the 13-inch case and contracted in the empty dewar case - both with appropriate adjustments in particle weight.

Results for these four configurations are shown in Figs. 4 through 7.

### 2.5 DESCRIPTION OF THE THERMAL NEUTRON CALCULATIONS

It is possible in the O5R calculations to continue tracing the neutrons until they reach thermal energy. One could then allow for upscattering and diffusion at thermal energy or switch to a single velocity model upon reaching the thermal region. This first and more exact treatment was ruled out because of the excessively long computing times it requires. The second approach was ruled out because the results would be in terms of an integral over the thermal group which is not useful for the intended purpose of comparison with the experimental spectra.

It was decided instead to trace the neutrons down to 0.75 eV and use the spectrum of neutrons degraded through 1 eV as the source term for a one-dimensional  $S_n$  treatment. This approach proved to be successful in that the source obtained from the O5R calculations did provide thermal spectra which are essentially in agreement with the experiment.

In the implementation of the 1DF  $\operatorname{program}^{(7)}$  for calculating the angular thermal flux along the dewar centerline use was made of the cross sections resulting from the theoretical work of Koppel and Young. <sup>(9)</sup> Scattering kernels calculated by Naliboff<sup>(10)</sup> for liquid para-hydrogen and liquid ortho-hydrogen were prepared as a cross section library tape by the WTFG3<sup>(11)</sup> code to make them compatible with the format used by the GGC-3<sup>(12)</sup> cross section averaging program. GGC-3 was then used to prepare the group-averaged cross-section data employed in 1DF. This


Figure 4. O5R Monte Carlo results for neutrons penetrating the empty liquid hydrogen dewar.



Figure 5. O5R Monte Carlo results for neutrons penetrating a 2.5-in. liquid hydrogen slab as viewed by a collimated detector.



Figure 6. O5R Monte Carlo results for neutrons penetrating a 4.5-in. liquid hydrogen slab as viewed by a collimated detector.



Figure 7. O5R Monte Carlo results for neutrons penetrating a 13.0-in. liquid hydrogen slab as viewed by a collimated detector.

approach is similar to that used in a previous analysis<sup>(2)</sup> of the hydrogen data for obtaining group-averaged data for the S<sub>n</sub> transport code GAPLSN. <sup>(13)</sup> GAPLSN, which was in use at GGA at the time of the other study, uses a cross section input format which differs from that in 1DF; consequently, group averaged data generated in the prior study could not be used directly.

In the input to the 1DF c. Iculations the angular distribution of the source term below 1 eV was considered to be isotropic. Spectral intensities corresponding to the plots in Fig. 8 were input at closely spaced mesh intervals. An infinite slab approximation was used. A radial buckling of . 0019 cm<sup>-2</sup> was used in all three cases. The calculations used a  $P_1$  expansion of cross sections with 46 energy groups and an  $S_8$  quadrature. The results of these calculations are tabulated in Appendix E and shown in Figs. 42, 43, 46, 52, 53 and 56.



Figure 3. Density of neutrons degraded through 1 eV as a function of position along the dewar centerline. The abscissa show the distance from the probe tube base. Curves for different thicknesses of liquid hydrogen in front of the probe base are shown.

#### 3. REEVALUATION OF THE FAST NEUTRON SPECTRAL DATA

#### 3.1 REASONS FOR REEVALUATION OF FAST NEUTRON SPECTRAL DATA

As may be seen in Table 1, some measurements made during the first contract period were repeated during the second one. There were inconsistencies in these data since some were measured with different detectors, reduced with different efficiencies from different references and utilized different neutron sources. Therefore, it was felt worthwhile to reexamine details of the data reduction process, use the latest efficiencies, and reduce all the fast neutron spectral data in a consistent manner.

#### 3.2 FACTORS WHICH INFLUENCE REDUCTION OF THE FAST NEUTRON SPECTRAL DATA

#### 3.2.1 INTRODUCTION

The spectral measurements are made by the time-of-flight technique. In this technique a pulse of neutrons are injected into the liquid hydrogen dewar. A probe tube samples the neutron spectrum at the chosen position by allowing those neutrons with the proper angle to escape down the probe tube. The neutrons are timed over a flight path of fixed length and are detected at the end of their flight by various neutron detectors selected for each energy range of interest. The fiducial point or reference time for the flight time is the gamma-flash or bremsstrahlung burst produced when the electron beam strikes the high-Z metal target producing a burst of photoneutrons or photofission neutrons.

#### 3.2.2 DETECTOR EFFICIENCIES FOR FAST NEUTRON DETECTORS

#### 3.2.2.1 Description of Detector and Method of Operation

Three different fast neutron detectors were used on the two programs. They are primarily characterized by the volume of the scintillator liquid and the surface upon which the neutron beam impinges. These detectors are listed below:

- 1. Fast Neutron Detector (2.0-in. diameter by 2.5-in. high) NE-211 liquid scintillator mounted with the axis perpendicular to the neutron beam.
- 2. Fast Neutron Detector 5.0-in. diameter by 5.0-in. high) NE-211 liquid scintillator mounted with the axis perpendicular to the neutron beam.
- 3. Fast Neutron Detector (5.0-in. diameter by 5.0-in. long) NE-211 liquid scintillator mounted with the axis parallel to the neutron beam.

Fast detectors (1) and (2) were used on the first program and detectors (1) and (3) on the second program.

The fast neutron detector is composed of a liquid scintillator viewed by a 14-stage photomultiplier tube. The scintillator, (NE-211\*), consists of xylene, activators, and POPOP as a wave shifter and is contained in a right cylindrical glass container. The composition of the NE-211 is stated by the manufacturer to be  $CH_{1,24}$ .

The neutrons interact with the hydrogen nuclei in the xylene producing recoil protons in the scintillator. Recoil carbon nuclei, and alpha particles from  $(n, \alpha)$  reactions in carbon also occur. The energy spectrum of the recoil protons extends from zero to a maximum energy equal to that of the incident neutrons. The theoretical shape of the proton recoil spectrum is shown in Fig. 9; the experimentally obtained pulse height distribution is shown on a linear proton energy scale. The energy from the

Manufactured by Nuclear Enterprises, Ltd., Winnipeg, Canada



Figure 9. Typical proton recoil spectrum (NE-211).

recoil protons is transferred to the electronic states of the molecules in the scintillator. De-excitation of the molecule takes place by the emission of ultraviolet light. The POPOP wave shifter absorbs the ultraviolet light and re-emits it in the frequency range appropriate for detection by a photomultiplier tube. The recoil carbon nuclei and alpha particles from reactions in carbon can also cause scintillations, although the light output per unit energy deposited is less than for protons.

#### 3.2.2.2 Fast Neutron Detector Efficiency

#### 3.2.2.2.1 Batchelor's Efficiency

The efficiency of any neutron detector is the number of counts per neutron incident on the detector, - front surface in our case. The relative efficiency of the scintillation detector versus neutron energy must be known to convert the counting rate to neutron flux.

For the first program, a 2-in. by 2-1/2 in. detector was chosen originally, because a detector of the same size and composition had been calibrated by Batchelor, et al,  $^{(14)}$  and was the best available efficiency at the time. The efficiency was both calculated by Monte Carlo methods and measured on a Van de Graaf accelerator. Once the efficiency of the 2-in. by 2-1/2 in. detector was known, the efficiency of the 5 in. x 5 in. vertically mounted detector was obtained by direct comparison.

The minimum pulse height accepted by the electronic circuitry is determined by the discriminator bias. At neutron energies near the bias energy, the detector efficiency is changing rapidly and is quite sensitive to the bias (it would be desirable to set the bias as low as possible and use the detector only for energies well above it). In order to use the Batchelor's efficiency the bias has to be the same as was used in his efficiency measurements. Originally, we attempted to adjust the bias so that the efficiency went to zero at the same energy as a linear extrapolation of Batchelor's curve would give, about 220 neutron keV. According to Batchelor's equations this was equivalent to 48.7 electron keV.

To determine the bias setting at 48.7 electron keV, reference was made to the Compton edge of the pulse height distribution taken with a  $^{137}$ Cs source. A typical pulse height distribution for a Compton spectrum is shown in Fig. 10. In terms of the gamma-ray energy E<sub>y</sub>, the energy of the maximum Compton electron energy E<sub>y</sub> is determined as follows:

$$E_{c} = E_{\gamma} \left[ 1 + \frac{m_{0}c^{2}}{2E_{\gamma}} \right]^{-1}$$

where  $m_0 c^2 = 0.511$  MeV. The electron energy  $E_c$  (was chosen) for these early measurements, as the pulse height at which the intensity of the pulses had fallen to approximately two-thirds of its value at the "Compton peak".

The efficiency given by Batchelor, et al.,  $(^{(14)})$  had a number of deficiencies, however. Pulse height resolution was not taken into account and the measured and calculated efficiencies deviated drastically below 1 or 2 MeV. The method of setting the bias was never explained satisfactorily even after private communications with one of the authors. The Van de Graaf calibrations were suspect because of sizeable collimator scattering corrections.  $^{(14)}$  3.2.2.2.2 Efficiency due to Verbinski, et al.

In 1964, shortly after the final report for the first program was published, Verbinski, et al.,  ${}^{(15)}$  published efficiency calculations for a 2 x 2 in. and 5 x 5 in. NE-213 liquid scintillator detector. These scintillators were adopted as standards since the calculations had been checked against measurements on a Van de Graaf accelerator.  ${}^{(16)}$  In addition, a reproducible bias was defined for these calculations.

Verbinski's calibration combined the best features of a series of measurements and of Monte Carlo calculations of pulse-height spectra for monoenergetic neutrons incident on an organic liquid scintillator.

In his calculations the half-height of the abrupt edge near the maximum pulse height (labeled H in Fig. 10) was assumed to correspond to the maximum hydrogen recoil energy and a trial function of proton light L versus neutron energy was constructed. This relationship was then used in a Monte Carlo calculation and the calculated pulse-height distributions were convoluted with a Gaussian smearing function until they matched the measured pulse-height distributions. The Monte Carlo calculations were then repeated with the correct  $L_{p}(E)$  function especially for the low-energy neutrons where the assumption that the half-height corresponds to the maximum recoil energy is not accurate. Values for similar functions for alpha particles and carbon-recoil ions  $L_{\rho}(E)$  and  $L_{\rho}(E)$  were also calculated but with less accuracy than  $L_{p}(E)$ . The final Monte Carlo calculations utilized the corrected  $L_{p}(E)$ ,  $L_{q}(E)$  and  $L_{c}(E)$  and the experimental pulseheight spectra were normalized to the calculated curves in the protonrecoil plateau, the experimental data thereby being converted to absolute differential efficiency. Integration of the absolute differential efficiency above the bias level represents the counting efficiency at the given bias setting. Details of the efficiency calculations and efficiency tables have been published in Refs. 15 and 17.

To use these tables, it is essential that the bias be defined in terms of a reproducible, standard pulse height or "light unit" chosen as the extrapolated end point of the Compton edge from the 1.17 MeV and 1.33 MeV gammas of Co-60 shown as "E" in Fig. 10. This unit is approximately equal to the light produced by 1.25-MeV electrons according to the prescription of Flynn, et al., (18) or 13 percent above the 1.28-MeV Compton edge half-height for Verbinski's system. The latter relationship, which uses the <sup>22</sup>Na gamma-ray pulse height spectrum, was chosen to allow reproduction of the experimental measurements and communication in the calibration procedures to other laboratories. <sup>(17)</sup>

The bias in terms of a fraction of the standard light unit is dependent upon the resolution of the detector. The pulse height resolution of two



Figure 10. Typical Compton electron spectrum.

scintillation detectors is almost never the same. The resolution being a function of the photoelectron statistics, which are dependent on the light output of the scintillator, light collection efficiency, and quantum efficiency of the photocathode. The use of the 1.28-MeV Compton edge half-height removes somewhat the sensitivity of the bias upon the resolution of the detector as opposed to the extrapolated end point (labeled E in Fig. 10).

#### 3.2.2.2.3 Detector Bias

The standard light unit has fluctuated somewhat, and during the second contract was defined as the half-height of the Compton edge from the 1.17 MeV and 1.33 MeV gammas of Co-60. <sup>(2)</sup> The efficiency of the detector is a sensitive function of the bias especially near the bias energy. Therefore, it was necessary to reevaluate the data from the second program as well as the first based upon the present definition of the standard light unit. To implement this it was necessary to obtain the relationship between the actual bias based upon the Compton edge from Cs-137 and the standard light unit related to the Compton edge from Cs-137 and the standard light unit related to the Compton edge from Na-22 as mentioned above.

No measurements were taken and therefore none are available for the experimental relationship between the Compton edges from Cs-137 and Na-22. However, the relationship between pulse height and energy has been measured for a liquid scintillator by Flynn, et al., <sup>(18)</sup> and extended by Horrocks. <sup>(19)</sup> The half-height of the upper edge of the pulse height distribution was identified with the Compton energy (actually the half-height is some  $4 \pm 1\%$  higher than  $E_c$ ). For energies below ~100 keV photopeaks are observed, but at a few hundred keV and above, gamma-ray sources give essentially a Compton distribution. For gamma-ray energies above about 800 keV, Flynn and Horrocks were able to fit their data by a linear expression

L = 
$$\frac{E_{c}^{(1)}cV)-18}{0.34}$$
 (E ≥ 80 keV)

where L is the relative light output and  $E_c$  is the maximum Compton electron energy. If we substitute  $E_c = 1061 \text{ keV}$  for  $^{22}\text{Na}$ , then  $L_{22} = 3068$  and if we substitute  $E_c = 478 \text{ keV}$  for  $^{137}\text{Cs}$  we get  $L_{137} = 1353.9$ . The ratio of the relative light output for the two gamma rays is

$$\frac{L_{22}}{L_{137}} = \frac{3067.6}{1353.9} = 2.266$$

The bias level based on  $^{137}$ Cs would then be

$$L_{B} = \frac{C_{B}}{C_{137}}$$

where  $C_B$  is the bias channel and  $C_{137}$  is the channel number for the halfheight of the Compton energy for the Cs-137 gamma ray. The light output definitions are shown schematically in Fig. 11 as would be measured by a pulse height analyzer. The channel for the standard light unit,  $C_{SLU}$ , would be 
$$C_{SLU} = (C_{137}) (2.266) (1.13) = 2.870 C_{137}$$

and the bias level based on the standard light unit would be

$$L_{B} = \frac{C_{B}}{2.870 C_{137}} = 0.3484 \frac{C_{B}}{C_{137}}$$

Although the relative light output is undoubtedly a sensitive function of the resolution of the detector, the ratio of the relative light output for the two gamma rays is not nearly as sensitive and any error resulting in this ratio is certain to be a second order effect.

In the same manner, if we substitute  $E_c = 1038 \text{ keV}$  for  ${}^{60}$ Co, then  $L_{60} = 3000 \text{ and } L_B$  based on the standard light unit instead of the halfheight of the Compton edge would be

$$\frac{L_{22}}{L_{60}} = \frac{3068}{3000} = 1.022$$



Figure 11. Light output versus energy.

and

$$C_{SLU} = (C_{60}) (1.022) (1.13)$$

so that

$$L_{B} = \frac{C_{B}}{(C_{60})(1.022)(1.13)} = \frac{C_{B}}{C_{60}(1.155)}$$

 $L_{B} = 0.8658 \frac{C_{B}}{C_{60}}$ .

#### 3.2.2.2.4 Efficiencies for the Three Fast Neutron Detectors

In summary, the efficiencies for the three detectors noted in Section 3.1 were obtained as follows. The efficiency for the fast neutron Detector 1 was taken from the tabulated values by Verbinski, et al., (15)after determining the bias level in terms of the standard light unit. The detector is 2-1/2 in. high whereas Verbinski's standard detector was 2 in. This, however, changes the amplitude of the efficiency and not the shape, i.e., the relative efficiency remains the same, and since experiment and calculation are normalized to each other only the relative efficiency is necessary. The efficiency for Detector 3 was taken from tabulated values published earlier by Verbinski, et al., (15) calculated for a similar detector after determining the bias in terms of the standard light unit. A small correction was necessary to correct for the difference in the light curves which had been changed since the earlier tabulations were published.<sup>(20)</sup> The efficiency of Detector 2 was determined by two separate methods: The first method was by comparison with Detector 1 using data taken during the first contract. The second method used the fact that Detectors 2 and 3 were similar in that the liquid scintillator was the same but they differed insomuch as the neutron beam impinges perpendicular to the scintillator axis in one case and parallel to the scintillator axis in the other. Nevertheless, recent experiments conducted at Gulf General

Atomic<sup>(21)</sup> indicated that this detector was isotropic when biased at 0.25light units which corresponds to about 1.3 MeV neutron energy, and therefore the efficiency is independent of the direction the neutron impinges upon the scintillator. Therefore, the efficiency can be obtained as for Detector 3 from the tabulated values published by Verbinski, <u>et al</u>.<sup>(17)</sup>

The efficiency for Detector 1, which is tabulated in Table 5 and shown in Fig. 12, was 0.02746 standard light units and was used to reduce the data taken on 7-31-64. Figure 13 compares the two detector efficiencies used to reduce the zero-degree spectral data of 7-31-64. The data, due to Batchelor, et al., <sup>(14)</sup> were used during the first contract period. The data, due to Verbinski, et al., (17) corresponding to a bias value of 0.02746 standard light units, were used under the present program. The two curves have been amplitude normalized at 1 MeV. The efficiencies agree in shape from 0.7 to 2.5 MeV and differ by only 18 percent at 15 MeV. The efficiency for Detector 2, which is tabulated in Table 6 and shown in Fig. 14, was 0.032 standard light units and was used to reduce data taken on 8-1-64, 8-3-64, 8-5-64, and 8-7-64. The detector was maintained at the same bias on all four days and therefore the same efficiency could be used. A comparison of the efficiencies for this detector by the two different methods described above is shown in Fig. 15. The first method is based on the efficiency of the  $2 \ge 2.5$  in. detector biased at 0.02746 standard light units and multiplied by the efficiency ratio of the  $5 \times 5$  in. to the  $2 \times 2.5$  in. detectors. The second method is based on a bias of 0.032 standard light units and taken from Verbinski's tabulations corrected for the change in light tables. The agreement is very good down to about 1 MeV verifying the isotropy of the detector down to this energy. Even at 0.8 MeV the disagreement is only about 30 percent. The efficiency labeled 0.032 standard light units was used since the uncertainty in the ratio of the two detectors increases at the lower energies and the uncertainty in the efficiency becomes so large at energies below 0.8 MeV that it is not very useful. Detector 3 which was used for measuring the data on 2-24-66 was

 $\mathbf{42}$ 

















STANDARD LIGHT UNITS						
Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency	
. 247	.01	. 62	. 470	3.75	. 35	
.260	.0155	. 64	. 475	4.0	. 34	
.270	. 022	. 66	.48	4.25	. 33	
.280	.03	. 68	.485	4.5	. 323	
. 290	.04	. 70	.49	4.75	. 315	
. 300	.054	.74	.495	5.0	. 308	
. 310	.072	. 78	. 498	5.25	. 300	
. 320	.105	. 82	. 50	5.5	. 293	
. 340	. 140	. 86	. 50	5.75	. 285	
.36	. 190	. 90	. 50	6.0	. 28	
. 37	.210	. 95	. 50	6.4	. 27	
.38	.230	1.0	.50	6.8	. 259	
. 39	. 250	1.1	. 495	7.2	.248	
.40	.270	1.2	. 495	7.4	.245	
.41	.285	1.4	. 477	7.6	. 242	
.43	. 32	1.6	. 462	7.8	.239	
.45	. 35	1.8	. 450	8.0	.235	
.47	. 375	2.0	. 437	8.4	.230	
. 50	.405	2.25	. 424	8.8	.225	
. 52	.423	2.5	.405	9.2	.22	
. 54	. 44	2.75	. 392	9.6	.217	
.56	. 45	3.0	. 38	10.0	. 214	
.58	.458	3.25	. 371	10 5	. 212	
. 60	.465	3.5	. 359	11.0	. 21	

# TABULATED EFFICIENCY FOR THE 2 x 2.5 IN. NE-211 LIQUID SCINTILLATOR BIASED AT 0.02746

47

بالمتهاجين

## TABULATED EFFICIENCY FOR THE 2 x 2.5 IN. NE-211 LIQUID SCINTILLATOR BLASED AT 0.02746 STANDARD LIGHT UNITS (cont)

Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency
11.5	. 206	13.5	. 194	15.5	. 188
12.0	.205	14.0	. 192	16.0	. 190
12.5	.200	14.5	. 190	16.5	. 190
13.0	. 197	15.0	. 189	17.0	. 192
		1			

Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency
. 13	.0103	.55	.573	. 88	.720
. 14	.0141	.56	.587	.90	.725
.16	.0206	. 57	.600	. 92	.73
. 18	.0288	.58	. 610	. 94	.733
. 20	.0386	. 59	. 620	. 98	.738
. 22	.0505	. 60	. 628	1.00	.745
.24	.065	. 61	. 635	1.1	.750
.26	.081	. 62	. 640	1.2	.753
.28	.101	. 63	.645	1.4	.757
. 30	. 123	. 64	. 65	1.6	.750
. 32	. 147	. 65	. 655	1.8	.740
. 34	. 174	. 66	. 66	2.0	.732
.36	.205	. 67	. 663	2.2	.725
. 38	.238	. 68	. 666	2.4	.72
.40	. 275	. 69	. 67	2.6	.705
.42	. 318	.70	. 675	2.8	.700
. 44	. 362	.72	. 68	3.0	. 689
.46	.407	.74	. 69	3.2	. 68
.48	.448	.76	. 695	3.4	. 665
.50	.488	.78	.700	3.6	. 65
.51	.504	. 80	.702	3.8	.645
.52	.525	. 82	.707	4.0	.632
.53	.540	. 84	.715	4.2	. 625
.54	.558	.86	.718	4.4	.617

### TABULATED EFFICIENCY FOR THE 5 x 5 IN. VERTICAL NE-244 LIQUID SCINTILLATOR BIASED AT 0.032 STANDARD LIGHT UNITS

# TABULATED EFFICIENCY FOR THE 5 x 5 IN. VERTICAL NE-211 LIQUID SCINTILLATOR BIASED AT 0.032 STANDARD LIGHT UNITS (cont)

Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency
4.6	.605	7.4	.510	10.5	. 445
4.8	. 60	7.6	. 502	11.0	.438
5.0	. 592	7.8	.498	11.5	.436
5.2	.583	8.0	.493	12.0	.432
5.4	. 577	8.2	.488	12.5	.430
5.6	.569	8.4	.482	13.0	.430
5.8	.560	8.6	.478	13.5	.430
6.0	.553	8.8	.472	14.0	.430
6.2	. 548	9.0	.468	14.5	.430
6.4	.540	9.2	.463	15.0	.430
6.6	.533	9.4	.460	15.5	.430
6.8	.528	9.6	.458	16.0	.430*
7.0	. 522	9.8	.453	16.5	.430
7.2	. 515	10.0	.451		

0.040 "cobalt units", and this was accomplished by splitting the 60 keV gamma-ray from  $^{241}$  Am. Shortly thereafter a similar detector was calibrated using a Van de Graaf accelerator, and the best fit to the data points was for an efficiency whose bias was 0.040 "cobalt units".  $^{(16)}$  This efficiency is given in Table 7 and shown in Fig. 16.

The efficiency for Detector 1 was based on a bias of 0.04233 standard light units and was used to reduce the data on 2-26-66. This efficiency is given in Table 8 and shown in Fig. 17.

#### 3.3 <u>OTHER FACTORS WHICH INFLUENCE REDUCTION OF FAST</u> NEUTRON DATA

After the data were reduced with the correct efficiency, there still remained an inconsistency in the data taken during the first and second contracts. As a first step in determining the reasons for this inconsistency, two fast neutron spectra for the water-cooled source were compared. These spectra were from the same neutron source but were taken by two different detectors on successive days. The data labeled 1-29-66, T4, and tabulated in Appendix A were taken with the same detector used to make spectral measurements on both the first and second programs. The data labeled 1-30-66, T2, were reported by Profio, et al., <sup>(6)</sup> and were taken with an early 2-in. x 2-in. detector, modeled after Verbinski, et al. (15)The spectra shown in Fig. 18 are the same down to about 1.5 MeV, below which the difference is related to bias uncertainties. The data labeled 1-29-66, T4 were reduced with an efficiency corrected for the new standard light unit definition whereas the other data were not corrected. Both sets of data are hemisphere integrated spectra, i.e., the collimation was large enough to view the entire surface of the spherical source. The spectra in Fig. 19, which are tabulated in Appendix E, are a comparison of the fast neutron spectra for the air-cooled and water-cooled sources. Here the collimator was only 0.87 in. in diam and the source was viewed through the



Efficiency of the 5 x 5-in. horizontal detector biased at 0.040 "cobalt units". Figure 16.



2.

DETECTOR ABSOLUTE EFFICIENCY



Figure 18. Comparison of the fast neutron spectra measured using two different detectors for the water-cooled source.





Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency
21.0	.410	1.25	.717
20.0	.412	1.0	. 690
19.0	.414	0.8	. 600
18.0	.416	0.6	.410
17.0	.418	0.5	.230
16.0	.420	0.4	.100
15.0	.422	0.35	.060
14.4	.422	0.30	.034
11.0	.427	0.25	.018
9.0	.461	0.20	.0077
7.0	.515	0.18	.0051
5.0	.583	0.17	.0041
4.0	. 618	0.16	.0032
3.0	. 665	0.15	. 0026
2.0	.709	0.10	.0011
1.5	.717		

# TABULATED EFFICIENCY FOR THE 5 x 5 IN. HORIZONTAL NE-211 LIQUID SCINTILLATOR BIASED AT 0.04 "COBALT UNITS"

# TABULATED EFFICIENCY FOR THE 2 x 2.5 IN. NE-211 LIQUID SCINTILLATOR BLASED AT 0.04233 STANDARD LIGHT UNITS

-	Absolute	Enoray	Absolute Detector	Energy	Absolute Detector
(MeV)	Efficiency	(MeV)	Efficiency	(MeV)	Efficiency
. 37	.0116	. 61	.270	1.1	.410
. 38	.015	. 62	.278	1.2	.411
.39	.0195	. 63	.285	1.4	.407
.40	.0253	. 64	.290	1.6	. 597
.41	.0315	. 65	.297	1.8	.388
.42	.0400	.66	.303	2.0	. 379
.43	.05	. 67	. 307	2.2	. 369
.44	.063	. 68	.314	2.4	. 360
.45	.079	.69	. 318	2.6	.352
.46	.096	.70	. 324	2.8	. 344
.47	. 114	.72	. 342	3.0	.337
.48	. 129	.76	.350	3.2	. 329
.49	. 144	.78	.359	3.4	. 322
.50	.158	.80	.364	3.6	.315
.51	.173	. 82	,370	3.8	.308
. 52	. 185	. 84	. 377	4.0	.302
,53	.198	.86	, 381	4.2	.296
.54	.209	.88	. 385	4.4	.290
.55	.218	.90	.390	4.6	.284
.56	.228	. 92	. 394	4.8	.279
. 57	.238	. 94	. 398	5.0	.274
.58	. 247	.96	.400	5.2	.268
, 59	.254	. 98	.402	5.4	, 274
.60	.263	1.00	.404	5.6	.258
		1		1	

# TABULATED EFFICIENCY FOR THE 2 x 2.5 IN. NE-211 LIQUID SCINTILLATOR BIASED AT 0.04233 STANDARD LIGHT UNITS (cont)

Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency	Energy (MeV)	Absolute Detector Efficiency
.254	8.4	.205	12.5	. 181
.250	8.6	.202	13.0	. 178
.245	8.8	.201	13.5	. 175
.240	9.0	. 199	14.0	. 173
.235	9.2	.197	14.5	. 172
,230	9.4	. 196	15.0	. 171
. 2.27	9.6	. 194	15.5	. 171
.223	9.8	. 193	16.0	. 172
.219	10.0	. 192	16.5	. 173
.216	10.5	.190	17.0	. 176
.213	11.0	.188	17.5	.178
.210	11.5	. 187	18.0	. 177
.208	12.0	. 185	18.5	. 176
	Absolute Detector Efficiency . 254 . 250 . 245 . 240 . 235 . 230 . 227 . 223 . 219 . 216 . 213 . 210 . 208	Absolute Energy   Detector Energy   Efficiency (MeV)   .254 8.4   .250 8.6   .245 8.8   .240 9.0   .235 9.2   .230 9.4   .227 9.6   .223 9.8   .219 10.0   .216 10.5   .213 11.0   .210 11.5   .208 12.0	Absolute DetectorEnergy (MeV)Absolute Detector.2548.4.205.2508.6.202.2458.8.201.2409.0.199.2359.2.197.2309.4.196.2279.6.194.2239.8.193.21610.5.190.21311.0.188.21011.5.187.20812.0.185	Absolute Detector EfficiencyEnergy (MeV)Absolute Detector EfficiencyEnergy (MeV).254 $8.4$ .205 $12.5$ .250 $8.6$ .202 $13.0$ .245 $8.8$ .201 $13.5$ .240 $9.0$ .199 $14.0$ .235 $9.2$ .197 $14.5$ .230 $9.4$ .196 $15.0$ .227 $9.6$ .194 $15.5$ .223 $9.8$ .193 $16.0$ .219 $10.0$ .192 $16.5$ .210 $11.5$ .188 $17.5$ .208 $12.0$ .185 $18.5$

empty liquid hydrogen dewar. The data have been corrected for the transmission of the dewar. The main difference between these spectra and those in Fig. 18 is that the hemisphere integrated spectrum is softer in the lower energy region due to the inclusion of large angle scatterings from the spherical surface. Figure 19 shows a reasonable agreement between the source spectra measured during the first and second programs. The data were taken one and a half years apart, as noted by the dates on the figure. For the 2-26-66, T4 data, there was a 3.144-cm thick gammaray filter of uranium which was not present for the 7-31-64, T6 data. As shown later, the uranium filter does not alter the spectral shape. The fast neutron sources were different on the two program periods. The physical difference in the sources for the two programs was the use of air cooling for the first and water-cooling for the second. The data shown in Fig. 20 provide a comparison of the hemisphere integrated fast neutron spectra for the air-cooled and water-cooled sources. The data for the air-cooled source, 5-23-64, T4, which are tabulated in Appendix E, are some of the first taken at this facility for the 3.0-in. diam uranium source. Though the statistics are not good, it can still be seen that the agreement in spectral shape between the air- and water-cooled sources is reasonable. Therefore, it can be safely concluded that the sources for the two contracts were identical in the fast neutron region, the same conclusion stated in the final report of the second contract.<sup>(2)</sup>

Since it was now obvious that source differences could not cause the observed spectral disagreements in the two data sets, other factors in the data reduction were closely examined.

The fast neutron flight path was approximately fifty meters long and although evacuated contained various materials such as mylar and aluminum windows, air, stainless steel from the walls of the dewar, and helium contained in the probe tube. These materials caused at the most a 20 percent correction due to their transmission and therefore even if the cross sections had been in error by 10 percent this would have accounted



Figure 20. Comparison of neutron spectra from the air-cooled and water-cooled fast neutron source.

\*\*\*\*\*\*
for at most a 2-percent transmission error. The thicknesses of the flight path materials were also examined to determine if the proper thickness had been used to reduce the data.

One of the major differences between the first and second programs was the insertion of 3.144-cm of uranium into the flight path. The purpose of the uranium was to act as a filter since the uranium suppresses the bremsstrahlung flash with only about a 40 percent correction for neutron transmission. To determine if the uranium filter influenced the data reduction techniques, the same source spectrum was measured with and without the uranium filter. As can be seen in Fig. 21, the filter did not influence the spectral shape for this test case.

The bremsstrahlung burst produced when the pulsed electron beam strikes the target is quite intense and if nothing were done to prevent its influence a loss of important data would result at early times and high neutron energies. However, since the photomultiplier is offgated during this period of time and the gain reduced by about a factor of a hundred, the effect is considerably reduced. Even so, some of this burst is detected since the detector is gamma sensitive, and appears as gamma rays decaying in an exponential manner. The technique for removal of these gamma rays is to fit the decay to an exponential and subtract it as a time-dependent background. The extent of this effect is determined by the electron beam current and pulse width. However, in this case the effect was small and influenced the spectra, if any, above about 10 MeV neutron energy. Only the data from the first contract had to be corrected since the uranium filter eliminated this problem in the second program.

The efficiency of the fast neutron detector decreases to zero around 0.2 MeV (see Figs. 12 to 15). The multichannel analyzer, however, continues to collect data after the neutron detector efficiency of the detector has stopped. The data obtained in the channels after this point, contain the ambient background of the detector as well as any time-dependent and





time-independent gamma rays. This background is removed from the signal prior to analysis by the data reduction code.

The uncertainty in the bias affects the spectral data below about 0.8 MeV and the uncertainty in removing the short lived gamma rays influences the spectral data above 10 MeV. Another effect, referred to as the fiducial point (zero time) can affect the spectral data above about 6 MeV. The fiducial point is determined by allowing the detector to see the bremsstrahlung flash and subtracting the flight time of the gamma rays. This requires that the offgate on the detector be removed during the burst since the purpose of the gate is to help suppress this flash. (16) This procedure is straightforward as long as the uranium filter is not present in the flight path as was the case for the first program. It becomes somewhat more complicated when uranium is in the flight path since the uranium also suppresses the bremsstrahlung flash making the exact channel number of the flash more difficult to determine. We had previously shown that the spectra were the same for the air- and water-cooled sources and therefore the source spectra through the LH, dewar for the two programs should be the same. Examination of the early-time behavior of these two source spectra indicated that the fiducial channel had been measured incorrectly.

The difference amounted to a shift in the fiducial point of four channels or 125 nsec. This shift has been incorporated into the zero-degree data of 2-26-66 and for the 37-degree data of 2-24-66. The resulting data for the 37-degree data are compared in Figs. 22 to 24. The two sets have been shape normalized. This correction brings the second set of data into agreement with the first and therefore makes both sets of data consistent.







Figure 23. Comparison of the spectral data for the 4.5-in. thickness of  $LH_2$  at 37<sup>°</sup> taken under the first and second programs.



Figure 24. Comparison of the spectral data for the 10.5-in. thickness of  $LH_2$  at 37<sup>o</sup> taken under the first and second programs.

### 4.0 REEVALUATION OF THE INTERMEDIATE NEUTRON SPECTRAL DATA

#### 4.1 INTRODUCTION

The intermediate neutron energy detector  $^{(22)}$  consisted of a NE-226 liquid scintillator counting capture gamma rays from a disk of boron carbide loaded with paraffin wax. The bias was determined as a fraction of the Compton energy for  $^{60}$ Co and  $^{137}$ Cs; it was 150 keV. The biasing proce-dure was similar to that for the fast neutron detector.

#### 4.2 INTERMEDIATE NEUTRON ENERGY DETECTOR EFFICIENCY

The detector and its efficiency calibration are discussed in Ref. 22 and the efficiency used in the LH<sub>2</sub> data reduction is tabulated in Table 9 and shown in Fig. 25. Because of the response to transmitted high-energy neutrons via inelastic scattering in fluorine, and uncertainties in the boron cross section and branching ratio at high energies, the efficiency of the detector is not expected to be very accurate above 2 MeV.

## 4.3 <u>OTHER FACTORS WHICH INFLUENCE REDUCTION OF INTER-</u> MEDIATE NEUTRON ENERGY DATA

The intermediate neutron flight path was approximately sixteen meters long and evacuated. The intermediate neutron detector was sensitive to thermal neutron captures consequently, a 0.175-in-thick boron filter was used to prevent pulsing overlap. The flight-path windows were mylar and a one-inch thickness of lead was placed into the flight path to minimize the effect of capture gammas. Both mylar and lead were chosen for use in the flight path since they contain a minimum of resonances in the intermediate neutron region. The fiducial point (zero time) was determined in a manner similar to that for the fast neutron





## Table 9

Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency
$1 \times 10^2$	.00636	$3.0 \times 10^5$	. 00753
$2 \times 10^2$	.00638	$3.3 \times 10^5$	.00754
$5 \times 10^2$	.00640	$3.50 \times 10^5$	.00755
$1 \times 10^3$	.00643	$3.60 \times 10^5$	.00754
$3 \times 10^3$	.00645	$3.70 \times 10^{5}$	.00753
$5 \times 10^3$	.00647	$3.75 \times 10^5$	.007525
$1 \times 10^4$	.00649	$3.80 \times 10^5$	.00752
$3 \times 10^4$	.00652	$3.85 \times 10^5$	.00745
$4 \times 10^4$	.006525	$3.90 \times 10^5$	.00738
$5 \times 10^4$	.006535	$4.0 \times 10^5$	. 00727
$5x5 \times 10^4$	.00654	$4.1 \times 10^5$	.00705
$6.0 \times 10^4$	.00655	$4.15 \times 10^5$	.00698
$6.5 \times 10^4$	.00656	$4.2 \times 10^5$	.00691
$7.0 \times 10^4$	.00658	$4.25 \times 10^5$	.00690
$8.0 \times 10^4$	.00663	$4.3 \times 10^5$	.00692
$9.0 \times 10^4$	.00669	$4.35 \times 10^5$	.00696
$1.0 \times 10^5$	.00675	$4.4 \times 10^5$	.00702
$1.1 \times 10^{5}$	.00681	$4.6 \times 10^5$	.00728
$1.4 \times 10^{5}$	.00702	$4.8 \times 10^5$	.00750
$1.7 \times 10^5$	.00722	$5.0 \times 10^5$	.00780
$1.9 \times 10^5$	.00730	$5.2 \times 10^5$	.00798
$2.0 \times 10^5$	.00735	$5.3 \times 10^5$	.00801
$2.3 \times 10^5$	. 00743	$5.4 \times 10^5$	.00803
$2.5 \times 10^{5}$	.00747	$5.5 \times 10^{5}$	.00805
$2.7 \times 10^{5}$	.00750	$5.6 \times 10^5$	.00804
$2.9 \times 10^5$	.00752	$5.7 \times 10^5$	.008035

## ABSOLUTE DETECTOR EFFICIENCY FOR THE INTERMEDIATE NEUTRON ENERGY DETECTOR

Energy	Absolute Detector Efficiency	Energy	Absolute Detector Efficiency
$5.8 \times 10^{5}$	.008025	$9.0 \times 10^5$	.00742
$6.0 \times 10^5$	.008000	$9.4 \times 10^5$	.00737
$6.2 \times 10^{5}$	.00797	$9.8 \times 10^5$	.00729
$6.6 \times 10^{5}$	.00789	$1.20 \times 10^6$	.00694
$7.0 \times 10^{5}$	.0078	$1.4 \times 10^6$	.00659
$7.4 \times 10^{5}$	.00772	$1.6 \times 10^6$	. 00621
$7.8 \times 10^{5}$	.00765	$1.8 \times 10^{\circ}$	.00588
$8.2 \times 10^{5}$	.00756	$2.0 \times 10^{6}$	.00558
$8.6 \times 10^{5}$	.00749	$2.1 \times 10^{\circ}$	.0055

## ABSOLUTE DETECTOR EFFICIENCY FOR THE INTERMEDIATE NEUTRON ENERGY DETECTOR (cont)

Table 9

detectors. In this case, however, the one-inch of leau did not suppress the bremsstrahlung flash as much as did the 1.238-in. of uranium nor was there any real need since data above 2 MeV were not very accurate as explained earlier.

1.18

The background removal was broken down into two parts. The first part was background composed of ambient, time independent gamma-rays or long lived gamma-rays and both were referred to as ambient. The second part was a time-dependent gamma-ray background produced by the capture of thermal neutrons in the  $LH_2$ . The energy of the capture gamma ray was 2.1 MeV and decayed exponentially with the thermal neutron dieaway of the  $LH_2$  in the dewar. Both backgrounds were removed from the signal during the analysis by the data reduction code.

## 5. REEVALUATION OF THE THERMAL NEUTRON SPECTRAL DATA

#### 5.1 INTRODUCTION

The thermal neutron energy detector consisted of a bank of 32  $BF_3$  counters mounted with their axis perpendicular to the beam.

## 5.2 THERMAL NEUTRON ENERGY DETECTOR EFFICIENCY

Details of the detector and its efficiency are discussed in Ref. 23. The detector efficiency used in the  $LH_2$  data reduction is tabulated in Table 10 and shown in Fig. 26.

## 5.3 OTHER FACTORS THAT INFLUENCE REDUCTION OF THERMAL NEUTRON DATA

The mean emission time or the mean time for a neutron of a particular energy to be emitted plays an important part in the data reduction. The mean emission time is the amount of time a neutron spends in the  $LH_2$ assembly before it is emitted. Since the fiducial point for the flight time is the bremsstrahlung burst the flight time appears longer than it actually is and must be corrected for the mean emission time. The current data reduction code simply subtracts the mean emission tim.<sup>3</sup> from the flight time. This is only an approximation<sup>(24)</sup>, however, an improved method could not be calculated due to the two-dimensional geometry of the  $LH_2$ . The mean emission times used to reduce the thermal  $LH_2$  data are shown in Fig. 27. The value of the asymptotic mean emission time or dieaway was obtained from the time-dependent capture gammas since their decay time is equal to the dieaway.



W. S. System



-11

Resident

## Table 10

Energy (eV)	Efficiency	Energy (eV)	Efficiency
$1 \times 10^{-4}$	. 6255	$3.0 \times 10^{-2}$	.6510
$1.5 \times 10^{-4}$	. 6682	$3.5 \times 10^{-2}$	. 6323
$2.0 \times 10^{-4}$	. 6941	$4.0 \times 10^{-2}$	. 6150
$3.0 \times 10^{-4}$	.7247	$4.5 \times 10^{-2}$	. 5995
$4.0 \times 10^{-4}$	.7423	$5.0 \times 10^{-2}$	.5862
$6.0 \times 10^{-4}$	.7617	$5.5 \times 10^{-2}$	. 5728
$8.0 \times 10^{-4}$	.7721	$6.0 \times 10^{-2}$	.5600
$1.0 \times 10^{-3}$	.7784	$7.0 \times 10^{-2}$	. 5393
$1.5 \times 10^{-3}$	.7864	$8.0 \times 10^{-2}$	.5213
$2.0 \times 10^{-3}$	.7897	$9.0 \times 10^{-2}$	.5048
$3.0 \times 10^{-3}$	.7909	$1.0 \times 10^{-1}$	.4897
$4.0 \times 10^{-3}$	.7885	$1.5 \times 10^{-1}$	.4327
$4.5 \times 10^{-3}$	.7865	$2.0 \times 10^{-1}$	.3935
$5.0 \times 10^{-3}$	.7736	$3.0 \times 10^{-1}$	.3411
$5.5 \times 10^{-3}$	.7771	$4.0 \times 10^{-1}$	.3064
$6.0 \times 10^{-3}$	.7736	$6.0 \times 10^{-1}$	.2616
$6.5 \times 10^{-3}$	.7624	$8.0 \times 10^{-1}$	.2329
$7.0 \ge 10^{-3}$	.7660	$1.0 \times 10^{0}$	.2123
$8.0 \times 10^{-3}$	.7629	$1.5 \times 10^{0}$	. 1787
$9.0 \times 10^{-3}$	.7579	$2.0 \times 10^{0}$	. 1576
$1.0 \times 10^{-2}$	.7519	$3.0 \times 10^{0}$	. 1316
$1.3 \times 10^{-2}$	.7284	$4.0 \times 10^{0}$	.1156
$1.6 \times 10^{-2}$	.7078	$6.0 \times 10^{0}$	.09591
$1.9 \times 10^{-2}$	. 6884	$8.0 \times 10^{0}$	.08388
$2.2 \times 10^{-2}$	. 6777	$1.0 \times 10^{1}$	.07554
$2.5 \times 10^{-2}$	.6673	$1.5 \times 10^{1}$	.06235

Contraction of the second second

# TABULATED EFFICIENCY OF THE THERMAL NEUTRON DETECTOR

25.00

Energy (eV)	Efficiency	Energy (eV)	Efficiency
$2.0 \times 10^{4}$	.05435	$1.0 \times 10^4$	.002533
$3.0 \times 10^{4}$	.04472	$1.5 \times 10^4$	.002069
$4.0 \times 10^{1}$	.03891	$2.0 \times 10^4$	.001792
$6.0 \times 10^{1}$	.03195	$3.0 \times 10^4$	.001464
$8.0 \times 10^{4}$	.02776	$4.0 \times 10^4$	.001268
$1.0 \times 10^2$	.02489	$6.0 \times 10^4$	.001035
$1.5 \times 10^2$	.02039	$8.0 \times 10^4$	.0008967
$2.0 \times 10^2$	.01770	$1.0 \times 10^5$	.0008021
$3.0 \times 10^2$	.01449	1.5 x 10 <sup>5</sup>	.0006550
$4.0 \times 10^2$	.01257	$2.0 \times 10^{5}$	.0005673
$6.0 \times 10^2$	.01028	$3.0 \times 10^{5}$	.0004632
$8.0 \times 10^2$	.008911	$4.0 \times 10^{5}$	.0004012
$1.0 \times 10^{3}$	.007976	$6.0 \times 10^{5}$	.000327598
$1.5 \times 10^3$	.006520	$8.0 \times 10^{5}$	.000283710
$2.0 \times 10^3$	.005650	$1.0 \times 10^{6}$	.000207197
$3.0 \times 10^3$	.004617	$1.5 \times 10^{6}$	.000179459
$4.0 \times 10^{3}$	.004001	$2.0 \times 10^{6}$	.000146509
$6.0 \times 10^3$	.003268	$3.0 \times 10^{6}$	.000126880
$8.0 \times 10^3$	.002821	$4.0 \times 10^{6}$	.000103617

# TABULATED EFFICIENCY OF THE THERMAL NEUTRON DETECTOR (cont)

Table 10

The background was measured by placing a B-10 filter over the entrance of the collimator and this was subtracted from the signal during data reduction.

WENELS, Which is dealed which have a second s

The flight path materials were one wall of the stainless steel dewar at liquid hydrogen temperature of  $20.4^{\circ}$ K (.0726 in. thick) one wall of the stainless steel dewar at room temperature (.0789 in. thick) helium in the probe tube at  $20.4^{\circ}$ K, mylar flight path windows, and air. The thermal neutrons were timed over an evacuated 16-meter flight path. These cross sections are fairly well known except for the stainless steel where the cross section is not known below 0.01 eV and very certainly not for temperatures of  $20.4^{\circ}$ K. It was clear from the experimental data that there was significant Bragg scattering by the stainless steel walls. Equally clear, however, was the fact that such Braff scattering would not be well described by the sum of the Bragg scattering of the stainless steel constituents, --iron, chromium, and nickel. The procedure used was therefore as follows:

- 1. The stainless steel was initially assumed to consist of the sum of its constituents which were described as 1/v absorbers with free atom scattering.
- 2. Several test cases of LH<sub>2</sub> data were reduced using the cross sections of Item 1. Where peaks occurred due to Bragg scattering, the flux above the peak was extended to just below it and the difference related to a constant change in cross section. These differences were obtained for each Bragg peak and averaged over the several test cases.
- 3. The cross sections for stainless steel, Item 1, were modified by the subtraction of their respective energies and below of the cross section differences from Item 2. All the  $LH_2$  data were then reduced with this same cross section set.

#### 6. DISCUSSION OF COMPARISON OF CALCULATIONS AND EXPERIMENT

## 6.1 MONITORS AND NORMALIZATION OF EXPERIMENTS AND CALCULATIONS

Each set of data, e.g., fast, intermediate, and thermal, was normalized to the neutron source intensity as measured by adjacent aluminum foil monitors. However, since each neutron energy region used a different detector system, the fast neutron data does not directly normalize to the intermediate energy region nor the intermediate to the thermal neutron energy region. Also, the solid angle to the front surface of the LH<sub>2</sub> varied as the thickness of LH<sub>2</sub> was varied due to the way in which the thickness changes were made. In view of these changes in geometry, it would be more appropriate to renormalize for each thickness over the entire energy range since the aluminum foil monitors do not track for different thicknesses. This method was attempted and discarded after some inconsistencies arose.

Another method of normalization which is perhaps not as good, is to normalize the fast and intermediate neutron energy range in the energy region where there is generally an adequate overlap. There is a gap between the intermediate and thermal energy regions, and the normalization in these energy regions was through their connection by a 1/E flux which appears to be established from 1 eV to  $10^4$  eV.

It is also necessary to normalize the calculations with the experiment and since fast and intermediate energy regions were calculated as one, these were normalized to the experiment by shape normalization. In the thermal region, the calculations and experiment were normalized in the 1/E region. In summary, the fast and intermediate experimental data

were normalized in their overlap region and the calculation shape normalized to the experiment. The thermal experimental neutron data were normalized to their calculations in the 1/E energy region. The thermal experimental data and calculations were normalized to the fast and intermediate experimental data and calculations in the 1/E energy region. All of the experimental data taken under both contracts and the calculations made under the second contract are given in Appendix E. In all cases, the fast neutron experimental data were the invariant and all other data normalized to them. The normalization factors are given for the experiments and calculations in Tables 11 and 12, respectively. To use these two tables, multiply the data listed under the column labeled "description" and tabulated in Appendix E by the normalization factor to match the corresponding fast neutron energy region.

#### 6.2 DISCUSSION OF REEVALUATED DATA AND CALCULATIONS

All of the reevaluated data have been tabulated in Appendix E. They are also displayed graphically in Figs. 28 through 56. For the  $0^{\circ}$  angle, measurements were made for only the fast neutron energy region, and at  $5^{\circ}$  for the intermediate neutron energy region. For 15, 37, 53, and  $78^{\circ}$ , measurements were made for both the fast and intermediate regions. For 37 and 78<sup>°</sup> angles, measurements were made from 10 MeV to .0005 eV. A gap in the data for these angles usually extends from 10 eV to  $10^{3}$  or  $10^{4}$ eV. This gap exists because no neutrons could be observed statistically above the time-dependent capture gammas. The  $0^{\circ}$  data for the 7.0, 10.5, and 13.0 in. thicknesses for the first program (7-31-64) are unsatisfactory: all of them give the same result. These were the first three measurements made and could have been caused by inexperience in transferring the LH<sub>o</sub> from compartment to compartment.

In a discussion of the experimental data, it is important to note some of the details surrounding the measurements. The fast neutron source intensity was limited since it could only accept 250 watts of electron beam power dictated by the air-cooling of the uranium target. In many cases, especially at large angle and thickness the signal-to-background

























Figure 35. Intermediate neutron spectrum for a 10.5-in. thickness of  $LH_2$  at 5°.











Figure 38. Neutron spectrum for a 4.5-in. thickness of  $LH_2$  at 15<sup>o</sup>.





Figure 40. Neutron spectrum for a 10.5-in. thickness of  $LH_2$  at  $15^{\circ}$ .



Figure 41. Neutron spectrum for a 13.0-in. thickness of  $LH_2$  at  $15^\circ$ .




















































Table 11

### NORMALIZATION FACTORS FOR THE EXPERIMENTAL DATA

#### A. Intermediate Neutron Energy Region

Descript	ion	Normalization Factor First Program	Normalization Factor Second Program
2.5-in.	15 <sup>0</sup> 37 <sup>0</sup> 53 <sup>0</sup> 78 <sup>0</sup>	9.34 x 10 <sup>-4</sup> 6.06 x 10 <sup>-4</sup> No Data 3.96 x 10 <sup>-4</sup>	$8.4 \times 10^{-2}$
4.5-in.	15° 37° 53° 78°	$8.84 \times 10^{-4} \\ 5.95 \times 10^{-4} \\ 8.0 \times 10^{-4} \\ 3.57 \times 10^{-4}$	$7.2 \times 10^{-2}$
7.0-in.	15 <sup>0</sup> 37 <sup>0</sup> 53 <sup>0</sup> 78 <sup>0</sup>	$6.45 \times 10^{-4}$ $5.00 \times 10^{-4}$ $5.78 \times 10^{-4}$ $5.56 \times 10^{-4}$	
10.5-in.	15 <sup>0</sup> 37 <sup>0</sup> 53 <sup>0</sup> 78 <sup>0</sup>	7.1 $\times$ 10 <sup>-4</sup> 4.76 $\times$ 10 <sup>-4</sup> 5.26 $\times$ 10 <sup>-4</sup> 4.76 $\times$ 10 <sup>-4</sup>	$7.57 \times 10^{-2}$
13.0-in.	15 <sup>0</sup> 37 <sup>0</sup> 53 <sup>0</sup> 78 <sup>0</sup>	9.7 $\times$ 10 <sup>-4</sup> 6.37 $\times$ 10 <sup>-4</sup> 4.17 $\times$ 10 <sup>-4</sup> 3.85 $\times$ 10 <sup>-4</sup>	

#### B. Thermal Neutron Energy Region

Descript	ion	Normalization Facto First Program	)r	Normalization Factor Second Program
2.5-in.	37 <sup>0</sup> 78 <sup>0</sup>	$3.8 \times 10^{-3}$ $3.8 \times 10^{-3}$		$5.5 \times 10^{-1}$
4.5-in.	37 <sup>0</sup> 78 <sup>0</sup>	$4.9 \times 10^{-3} \\ 4.9 \times 10^{-3}$	• •	$6.0 \times 10^{-1}$
7.0-in.	37 <sup>0</sup> 78 <sup>0</sup>	$2.5 \times 10^{-3}$ 2.5 x 10 <sup>-3</sup>		
10.5-in.	37 <sup>0</sup> 78 <sup>0</sup>	$2.4 \times 10^{-3}$ $2.5 \times 10^{-3}$		$4.0 \times 10^{-1}$
13.0-in.	37 <sup>0</sup> 78 <sup>0</sup>	$3.0 \times 10^{-2}$ $3.0 \times 10^{-2}$ 109		

#### Table 12

#### NORMALIZATION FACTORS FOR THE CALCULATIONS

Descrip	tion	Normalization Factor First Program	Normalization Factor Second Program
2.5-in.	0 <sup>0</sup>	$1.00 \times 10^4$	9.26 x 10 <sup>5</sup>
	37 <sup>0</sup>		$3.82 \times 10^{6}$
	78 <sup>0</sup>	9.53 x $10^4$	
4.5-in.	0 <sup>°</sup>	$1.03 \times 10^4$	
	15 <sup>0</sup>	$6.76 \times 10^4$	
	37 <sup>0</sup>		$3.80 \times 10^6$
	78 <sup>0</sup>	$3.3 \times 10^5$	
13.0-in.	0 <sup>0</sup>		$7.87 \times 10^5$
	37 <sup>0</sup>	$4.08 \times 10^5$	: 
	78 <sup>0</sup>	$1.56 \times 10^{6}$	

#### A. Intermediate Neutron Energy Region

#### B. Thermal Neutron Energy Region

Descrip	tion	Normalization Factor	r
2.5-in.	37 <sup>0</sup>	6.54	
	78 <sup>0</sup>	8.85	
4.5-in.	37 <sup>0</sup>	7.0	
	78 <sup>0</sup>	10.4	
13.0 <i>-i</i> .a.	37 <sup>0</sup>	29.0	
	78 <sup>0</sup>	40.0	

ratio was low and in general the better data are therefore at small angles and thicknesses. During the second program, a water-cooled uranium target was developed and the fast intensity increased by a factor of six since the source could accept 1500 watts of electron beam power. This increase in neutron intensity is reflected in the comparison of the fast neutron data for the two contracts.

The measurements were made on a one-shot basis due to the tremendous detail and preparations required to handle the 150 gallons of liquid hydrogen within the LINAC facility under limited funding. It was not possible to review the data until after the experiment was completely finished. It would certainly have been desirable to have reviewed preliminary measurements before taking the complete data set. Only a few measurements were remeasured and these were all in the fast neutron energy region.

In considering the massive amount of experimental data and comparison of some of the data with calculations, a few general observations can be made. First, the Monte Carlo calculations suffer from poor statistics in precisely the same way as do the measurements. Since in the fast neutron energy region the flux is in the forward direction, few neutrons are scattered at large angles and the source intensity becomes quite degraded by the thickness of LH<sub>2</sub>, resulting in even less neutrons at large thicknesses and also larger angles. The agreement between calculation and experiment, in shape at 37° is good for all the thicknesses compared, even out to the 13.0-in. thickness. The agreement is not nearly as favorable at 78° at these same thicknesses. The data taken at 13.0-in. and 78 degrees are so unsatisfactory that they have not been presented. In at least one situation, there is a very good comparison between calculation and experiment; in the fast-intermediate neutron energy region this is for the 2.5-in. 37° data. Here the agreement is good between  $10^6$  to  $10^3$  eV. This is also a situation where the statistics are the best for both experiment and calculation. The agreement between the thermal neutron calculations and

experiment are favorable in half of the cases compared but vary considerably in the remainder. These disagreements occur at an energy where the transport cross section changes drastically from about 20 to 1 barn in the energy region from .04 to .01. This also leaves open the question of the applicability of a one-dimensional calculation below 0.01 eV since the transverse leakage becomes large. The buckling or source distribution for LH<sub>2</sub> was not measured and is therefore not well known. Due to the small transport cross section below 0.01 eV the calculations are quite sensitive to the buckling and therefore small changes in the buckling result in large changes in the flux.

We have recommended that the 2.5- and 4.5-in. thicknesses of  $LH_2$ for the 37° angle be submitted as a benchmark problem to the ANS-6 Standards Committee, American Nuclear Society. The reason for this selection was that the statistics for the Monte Carlo calculations were best for the 37° and 78° angles, and calculations were performed for only 2.5, 4.5, and 13.0-in. data. Of these, the calculational and experimental statistics were best for the 2.5- and 4.5-in. thicknesses at the 37° angle.

#### REFERENCES

- G. D. Trimble, G. K. Houghton, and J. H. Audas, "Measurement of Neutron Spectra in Liquid Hydrogen, "NASA Report NASA CR-54230, General Atomic Division, General Dynamics Corporation (1964).
- 2. G. D. Trimble, G. K. Houghton, and A. E. Profio, "Measurement of Neutron Spectra in Liquid Hydrogen," NASA Report NASA CR-72143, General Atomic Division, General Dynamics Corporation (1966).
- D. C. Irving, R. M. Freestone, Jr., F.B.K. Kam, "O5R, A General-Purpose Monte Carlo Neutron Transport Code," ORNL 3622, Feb. 1965.
- 4. M. K. Drake, "Neutron and Gamma Ray Production Cross Sections for Silicon," DASA Report No. 2099, Gulf General Atomic Inc., May 1, 1968.
- 5. BNL 50066 (T-467) "ENDF/B Specifications for an Evaluated Nuclear Data File for Reactor Applications," by Henry Honeck, May 1966 Revised July 1967 by S. Pearlstein.
- 6. A. E. Profio, R. M. Hackney, G. D. Trimble, H. M. Antunez and J. L. Russell, Jr., "Angular Flux Spectrum and Other Characteristics of the 7.62-cm Diameter Uranium Target," Gulf General Atomic Report GA-7409 (1966).
- 7. K. D. Lathrop, 'DTF-IV, A FORTRAN-IV Program for Solving the Multigroup Transport Equation with Anisotropic Scattering,'' Los Alamos Scientific Laboratory Report LA-3373 (1965).
- 8. F. B. K. Kam, K. D. Franz, "ACTIFK, A General Analysis Code for O5R," ORNL-3856, Sept. 1966.
- 9. J. U. Koppel and J. A. Young, "Slow Neutron Spectra in Molecular Hydrogen at Low Temperatures," General Atomic Report GA-6003 (1965).

- Y. D. Naliboff, "LHK An IBM 7044 Code to Calculate Cross Sections and Kernels for Liquid Para- and Ortho-Hydrogen," NASA Report NASA-CR-54227, General Dynamics Corporation, General Atomic Division, July 1964.
- 11. M. K. Drake, C. V. Smith and L. J. Todt, "Description of Auxiliary Codes Used in the Preparation of Data for the GGC-3 Code," General Atomic Report GA-7158, August 7, 1967.
- 12. J. Adir, S.S. Clark, R. Froehlich and L. J. Todt, "Users' and Programmers' Manual for the GGC-3 Multigroup Cross Section Code," General Atomic Report GA-7157, July 1967.
- J. H. Alexander, G. W. Hinman and J. R. Triplett, "GAPLSN, A Modified DSN Program for the Solution of the One-Dimensional Anisotropic Transport Equation," General Atomic Report GA-4972, March 1964.
- 14. R. Batchelor, W. B. Gilboy, J. B. Parker and J. H. Towle, <u>Nucl</u>. Instr. Methods, 13, 70-82 (1961).
- V. V. Verbinski, J. C. Courtney, W. R. Burns, and T. A. Love, "The Response of Some Organic Scintillators to Fast Neutrons," in ANS-SD-2, Proceedings of the Special Session on Fast Neutron <u>Spectroscopy</u>, American Nuclear Society 1964 Winter Meeting, Shielding Division Report ANS-SD-2 (February 15, 1965).
- 16. A. E. Profio, P. L. Phelps, J. W. Riggs, and J. C. Young, Design and Efficiency Calibration of the 5-cm and 13-cm Fast Neutron Detectors, General Atomic Report GA-7483, (1966).
- V. V. Verbinski, W.R. Burrus, T. A. Love, W. Zobel, N. W. Hill, and R. Texter, "Calibration of an Organic Scintillator for Neutron Spectrometry", ORNL-TM-2183, June 10, 1968.
- K. F. Flynn, L. E. Glendenin, E. P. Steinberg, and P. M. Wright, Nucl. Instr. Methods, 27, 13-17 (1964).
- 19. D. L. Horrocks, Nucl. Instr. Methods, 30, 157-110 (1964).
- 20. Private communication from V. V. Verbinski.
- 21. Private communication from L. Harris and H. Kendrick.

- 22. A. E. Profio, "The Boron Capture Detector For Time-of-Flight Measurements," General Atomic Report GA-7578 (1966).
- J. R. Beyster, et al., "Integral Neutron Thermalization, Annual Summary Report, October 1, 1965 through September 30, 1966," USAEC Report GA-7480, General Dynamics Corporation, General Atomic Division, November 1966.
- 24. P. d'Oultremont, "A Method for Correcting the Time-of-Flight Spectra for the Neutron Emission Time," Gulf General Atomic Incorporated Report GA-9269, June 1969.

#### APPENDIX A

# GEOMETRY INPUT TO 05R (13 INCHES OF $LH_2$ BETWEEN THE SOURCE AND COLLIMATOR)

-	2		MA	LE																	
201	EX	UN HN	DS DC	-	•60'	960 960	+02 103	•	•	7041 6001	80+( 60+(	12 12									
ZON	Ēz	BN	DS	-	.61	960	+02			609	5Ü+I	52									
ZON	E	-	1		1	1	*00				20	_			200			704		-	
HLU	CKY	HN.	05		•00' •60'	960	+0# ∔02		اية.	6090	60+1	າວ່		0.21	2020	)TU <b>CI</b>	. (	6 7 0 4 1	50702		
BLU	CKZ	BN	DS	-	.60	960	+02	•	•	6090	50+	22									
BLU	CK		1	,	1	1			•												
SUH	FAC	FS			1.		21		U												
SEL	TOR	-1	0																		
SEC	TOR	1	-1																		
SEL	TOR	Ű	1																		
BLO	CK		2		1	1															
MEU	IA				<b>`</b> 1,	-	2,		1.		2.		1+		21	1	•	21	1.	2,	
_	3,	_	0										_								
SUL	FAC	ES			3,		4,		5+		61		7+	•	A+	9	• :	10,	11.	121	
SEL	TOH	-1	0	-1	ก	0	0	0	0	â	<b>=</b> 0	⇔∩									
SEC	TOR	-1	ō	i	-1	ō	ō	ŭ	ō	ŏ	~Õ	-0									
SEC	TOK	-1	Ö	0	1	-1	0	0	0	0	-0	-0									
SEC	TOR	-1	0	0	0	1	1	0	0	0	-0	-0									
SEL	TOP	-1	0	0	0	0	<u> </u>	-1	-1	0	-0	-0									
SEL	TOK	-1	ō	Ö	Ö	ŏ	ö	ō	< ī	-1	-0	-ŭ									
SEL	TOR	-1	0	0	0	0	0	0	0	1	-1	-0									
SEL	TON	-1	0	0	0	Û	0	0	0	0	1	-1									
SEG	TOR	. <b>-1</b>		0	0	0	0	0	0 0	0	0	1									
SEC	TOR	ō	ī		ő	õ	ŏ	ŏ	ŏ	ŏ	ŏ	ö									
					_																
BLU	CK		3		1	1	а.		7.	1.00	۱ <b>۰</b> .	1.00	h.	\$ 44							
SUR	FAC	ES			3.		4,		14.	100	007 157	100	16	- <b>4</b> 01	00						
SEC	TOR	1	0	-1	Ū,	0	• •														
SEC	TOR	-1	0	1	-1	0									•						
SEC	TOR	1	-1	0	0	-1															
SEC	TOR	1	-1	0	0	1															
SEC	TOR	ō	: ī	ñ	Ö	ō							•								
			_					_													
	16	000	11 040	OTAI VXCI	เรเ	JKF	ACE	5 100	0/1+1	<b>1 1 Y</b>	50			. 10	0004	L0175	à	-	. 28452	<b>↓</b> ∩µ	e
	.1	000	0+0	1×5	9		:	100	00+	01.45	SQ			.10	0004	01Z5	ġ.		.37161	+04	Ś
	•1	000	0+0	115	9		•	100	n0+(	12	50		-	.28	4521	104	5				
	•1	000	0+0	175	9		•	100	(\$1)+(	01Z9	50	-	-	.37	1614	+04	5				
	- 1	000	0+U. 0+0:	1 X 1 X			-	635	00+1	01 01		2									
		000	0+0	1X			· #	112	70+	ĎŽ		ŝ									
	+1	000	0+0	1X				114	30+	02		\$									
	•1	000	0+01	1X			-,	176	20+0	20		\$									
	•1	000	0+0) 0+0'	LX VY			7	177	80+(	12		5									
	4 د الم	000	0+02 0+02	ÈX.			-	200) 266'	107( 76∔1	, e 12		s									
	•1	600	0+0	IX.			-	329	60+(	2		5									
	•1	000	0+0	125	9		*	100	00+0	1175	50			10	0004	0125	3		1066A	+0 <b>3</b> X	
	8	672) 672)	4+04 0+0'	t I Xer	3	6		100	0014	n 1 v 4				.1.0	000-		0		10647		
	•• ••1	036	640!	יהיי 5		5	٠	100	UVTI		a QF			+0	VUU4	0143			1 Y AADH.	4 B <b>Q</b> %	
	.1	000	0401	โม		-		462	80÷1	12		•									

and the second se

¢

à.

A-1

### APPENDIX B

小学のないないとな

1.000

## KINNY PROGRAM LISTING

1.		SUBROUTINE EVAP(E0,EP,THETA,ELO)
۷.	С	ELO EQUALS LOWEST EDERGY
3.	с	PICK FROM X+LXP(-X) BY KAHNS METHOD, EXAMPLE 15
4.	с	INSTEAD OF PICKING FROM EXR(-X), X INFINITE, PICK
5.	с	FROM EXP TRUNCATED AT XMAX. REJECT VALUE X IF THE SUM OF
0.	С	OF THE THO VALUES IS GREATER THAN XMAX.
7.		XMAX=(EO-ELO)/THETA
+UIAGN	*STIC*	THE TEST FOR EQUALITY HETWEEN NON-INTEGERS MAY NOT BE MEANINGFUL.
٥.		IF (XMAX.LE.0.0) CALL ERPOR
9.	2	R1=AMOD(EXPRIF(R),XMAX)
10.		R2=AMOD(EXPRINE (R), XMAX)
11.		X=R1+R2
12.		IF (X-XMAX)1+1+2
15.	1	EP=X * THE TA
14.		RETURN
15.		END
1.		FUNCTION CHOS(R+P+Q+NPQ)
٤.		DIMENSION P(1), Q(1)
3.	с	INTERPOLATES FROM HIGH TO LOW RANGES ONLY
4.		DO 10 $I=2,NPQ$
*DIAGN	OSTIC*	THE DU-INDEX, I, IS DEFINED WITHIN THE LOOP.
5.		I=I
*DIAGN	OSTIC*	THE TEST FOR EQUALITY PETWEEN NON-INTEGERS MAY NOT BE MEANINGFUL.
0.		IF(P(1).LE.R) GO TO 20
7.	10	CONTINUE
ö.		CALL ERROR
9.	20	CHOS=u(1)+(R-P(1))*(u(1)-0(1-1))/(P(1)-P(1-1))
10.		IF(CHOS.LT.0.0) CALL ERROR
·1•		RETURN
14.		
1.		SUBROUTINE ERROR
۷.		COMMON / INELC/ SPIUP, VP, PP, EJS, NINESIEN
3.		A = (A + (1) + ((3) - (3) + (1) + (3) + (2) + (3) +
4.		3 HQ(10/)ELAW(2/10/)NLAV(2/10/)NENG(1/1/NC(1/1/)
5.		$ = N \in [ADI(1,1), F \in ADI(1,1), F \in ADI(1,1), ADI(1,1)$
		= [M(2,1)] = [M(2,1)] = [M(2,1)] + [M(2,1)
· ·		
		WRITE (6.0000)
1	8000	FORMAT (20HIAN FREOR HAS OCCURRED)
11	0000	WRITE (6.301) SPUP. VP. P.E.S. IN
1.		RETURN
13.	301	FORMAT(1P6E12.5, I10)
14.		END

```
1.
               SUBROUTINE ELAWS (SPDSQ. WATE)
        С
   4.
               COMMON /INELL/ SPOUP, VE ... P.E.S.NINLS, IN.
   j.
              A WR(10),6(+,5),NE(+,5),LL(125,8,5),SIGL(125,8,5),
   4.
              1) (10), ELAN(2,10), MLA: (2,10), NENG(1), HG(1,1),
   5.
              C .: [ FAU1(1,1), ETAR1(1,1,1), PTAB1(1,1,1),
   0.
   1.
              U LETAJ2(1), ETA42(1,1), PTAB2(1,1),
              L FM(2,10), FA(2,10), LU(10), TA(10), TB(10), TC(10),
   0.
   4.
              F LAPP(10), MALDE(2,10), MELE(2,10)
  10.
               IL=1
  11.
               IF (NU(III).EQ.1) GU TU 200
               145=HU(II.)
  12.
               00 105 IL=1+1.5
  1..
*UIAGNOSTIC*
               THE DU-INDEX, IL, IS DEFINED WITHIN THE LOUP.
  14.
               IL=IL
+DIAGNOSTIC+
               THE TEST FOR EQUALITY PETWEEN NON-INTEGERS MAY NOT BE PEANINGFUL.
  15.
               IF (ELAW (IL, IN) . LE.E) GO 10 200
  10.
          105 CONTINUE
  11.
               CALL LARVER
  10.
          200 NL=NLAW (1L, II.)
               60 TO (11,22,33,44), NL
  19.
  20.
        С
                  TAUULATED VALUES
  21.
            11 HELSNETIG(IN)
  22.
               00 10 IE=1,NLL
#UIAGNOSTIC*
               THE DU-INDEX, IE, IS DEFINED WITHIN THE LOUP.
               IL=IE
  23.
+UIAGNOSTIC+
               THE TEST FOR EQUALITY HETWEEN NON-INTEGERS MAY NOT HE MEANINGFUL.
  24.
               IF (ENG(IE, IN) .LE.F) GO TO 30
  25.
            10 CONTINUE
               CALL ERRUR
  20.
  21.
            30 P=FLTRN(R)
           21 EP=CHUS(P, PTAB1(1, IE, IN), ETAB1(1, JE, IN), NETAB1(IE, IN))
  20.
  24.
               GO TO 100
  30.
            22 P=FLTKN(K)
               EP=CHUS(P, PTAB2(1, IN), ETAB2(1, IN), NETAD2(IN))*E
  31.
               GO TO 100
  32.
  33.
        С
                  EVAPORATION
  34.
           33 THETA=TA(IN)+TB(IN)+SQPT(E)+TC(IN)+E
  35.
               CALL EVAP(E, EP, THETA, ELAN(IL, IN))
  30.
               GO TO 100
           44 JJJ=1
  31.
               THE TEST FOR EQUALITY PETWEEN NON-INTEGERS MAY NOT BE MEANINGFUL.
*UIAGNOSTIC*
  30.
               IF (FLIRH (DUM).GE.U.5) JJJ=2
  34.
               EP=LA(JJJ, IN)
  40.
          100 SP=SQRT(EP+1.9132201E+12)
          300 IF(SP.GT.S) CALL ERROR
  41.
               WATE= ,ATE +FM(IL, IN)
  42.
  43.
               RI TURIA
  44.
               END
```

1.		SUBROUTINE LEVELS (SPDSQ. BETA)
۷.	с	
5.		COMMON /INELC/ SP.UP, VP, WP, E, S, NINLS, IN,
4.	A	NR(10),G(3,5),NE(6,5),EL(125,8,5),SIGL(125,8,5),
5.	E	3 NU(1U), ELAW(2,10), NLAW(2,10), NENG(1), ENG(1,1),
ο.	C	NETAD1(1,1), ETAB1(1,1,1), PTAB1(1,1,1),
7.	C	NETA52(1).EIA52(1.1).PTA52(1.1).
<b>d</b> .	E	FM(2,10), FA(2,10), EB(10), TA(10), TB(10), TC(10),
4.	Ē	NAPP(10), NMEDE(2,10), NELE(2,10)
10.		DIMENSION PTAB(10)
11.		
12		IE(NP(TH), EQ.1) GO TO PO
13		
14		
15	c	
16	U U	
17		
1		
10		F = A = (1, 1, 1, 1, 1) OD E LT EL (NEM+1+1N) GO TO EQ
21		$\Gamma (c, G) \in L(1, 1, 1, N), GR \in C^{+}(C, 1, 1, 1, N), N \in (T, 1, N))$
20.		
21.	6.0	
-01-	UCTIC	VUNTINUE COR FOUNTIN DETWEEN NON-INTECERS MAY NOT DE VERNIMMENT
TUTAG	105110+	THE TEST FOR EQUALITY BETWEEN NON-INTEGERS MAT NOT BE MEANINGFUL.
23.		
24.		
25.		
20.	60	CONTINUE
27.		D0 65 1=2,N5
20.		PTAB(I) = PTAB(I) + PTAB(I-1)
29.	65	CONTINUE
30.	C	CHOOSE LEVEL
31.	100	R=FLTRN(R)
32.		DO 70 IL=1,NS
*DIAG	NOSTIC*	THE DU-INDEX, IL, IS DEFINED WITHIN THE LOOP.
33.		ILEIL
*DIAGN	NOSTIC*	THE TEST FOR EQUALITY PETWEEN NON-INTEGERS MAY NOT BE MEANINGFUL.
34.		IF(R.LE.PTAB(IL)) GO TO BO
35.	70	CONTINUE
Зυ.		CALL ERROR
37.	С	CHECK FOR INSUFFICIENT CMS ENERGY
30.	80	IF(SPUSQ.LT.Q(IL.IN)/BETA) GO TO 90
39.		SP=BETA*SQRT(SPDSQ-Q(1L,IN)/BETA)
40.		RETURN
41.	90	CALL ERROR
42.		GU TO 100
43.		END

1.		SUBROUTINE DCS(U,V,W)
2.	С	COMPUTES NEW DIRECTIONS COSINES
3.		COMMON /INELC/ SP,UP,VP,WP,E,S,NINLS,IN,
4.		A NR(10),0(8,5),NE(8,5),EL(125,8,5),SIGL(125,8,5),
5.		B NU(10),ELAW(2,10),NLAW(2,10),NENG(1),ENG(1,1),
6.		C NETAD1(1,1),ETAB1(1,1,1),PTAB1(1,1,1),
7.		D NETA62(1),ETA82(1,1),PTA82(1,1),
ы.		E FM(2,10),EA(2,10),EB(10),TA(10),TB(10),TC(10),
9.		F NAPP(10), NMEDE(2,10), NELE(2,10)
10.		COMMON/SNGLES /BLZON, EROT, ECUT, EGROUP, EINC, EMONO, ESOUR,
11.		1ETAPE, ETA, ETATH, ETAUSD, ETOP, FONE, FTOTL, FWATE, ITERS, ITSTR,
12.		2LELEM, LF, MARK, MAXGP, MEDIA, MGPREG, MFISTP, MXREG, N, NCOLPR,
13.		3N#PCOL, NCONT1, NCONT2, NCONTP, NEWNM, NFISH, NFONE, NFPT, NGEOM,
14.		4NGROUP,NGWT,NHISMX,NHISTR,NINC,NFINC,NPINC,NITS,NKILL,
15.		5NLAST,NGLAST,NSIGL,NPLAST,NLEFT,NMEM,NMOST,NOEL,NPCOF,
16.		6NPTAPE,NQUIT,NROOM,NSOUR,NSPLT,NSTAPE,NSTRT,NTHERM,NTHRML,
17.		7NTYPE,OLDWT,PSIE,SPOLD,THETM,TNUC,UINP,UOLD,VINP,VOLD,
16.		BWATEF,WINP,WOLD,WTAVR,WTHIR,WTLOR,WTRED,WTSTRT,XOLD,XSTRT,
19.		9YOLD, YSTRT, ZOLD, ZSTRT, UNUSED (10)
20.	С	CALL GETMU FOR SCATTERING ANGLE
21.	-	CALL GETMU(COSPOL)
22.	С	GETMU RETURNS LF (SCATTERING DISTRIBUTION)
23.	-	IF(LF.NE.0) GO TO 10
24.	С	ISOTROPIC SCATTERING
25.		CALL GTISO(UP,VP,WP)
26.	~	RETURN
21.	C	$\frac{10}{5} \frac{10}{5} \frac$
20.		
27.		$\nabla I = I = -0.5$
30.		$\frac{1}{1} = \frac{1}{1} = \frac{1}$
30		
33		
34.		G0 T0 2
35.		1 ST=S*STHETA
36.		COSPHI=V/ST
37.		SINPHI=W/ST
38.		2 CALL AZIRN(SINAZI, COSAZI)
39.		C1=COSAZI*SINPOL
40.		C2=SINAZI*SINPOL
41.		C3=STHETA*COSPOL
42.	С	DIRECTION COSINES
43.		UP=CTHETA*COSPOL-C1*STHETA
44.		VP=C3*COSPHI+CTHETA*COSPHI*C1-SINPHI*C2
45.		WP=C3*SINPHI+CTHETA*SINPHI*C1+C0SPHI*C2
40.		RETURN
47.		END

54.		READ(5,103) (ETAB1(L,K,I),PTAB1(L,K,I),L=1,NTAB1)
55.		WRITE(6,309) ENG(K,I),(ETAB1(L,K,I),PTAB1(L,K,I),L=1,NTAB1)
50.	17	CONTINUE
57.	-	60 10 20
54	22	READ(5.102) EM(J.I).NETAB2(I)
50		
57.		
60.		$NTAD \ge NTAD > (TAD > 1 + TAD > 1 + T) + DTAD > 1 + T + NTAD > 1$
01.		
02.		WRITE(6,510) (ETAB2(L,1)) FTAB2(L) //L=1,NTAB2)
63.		GO TO 20
64.	33	
05.		WRITE(6,308) FM(0,1)
66.		READ(5,104) TA(1), TB(1), TC(1)
67.		WRITE(6,311) TA(1),TB(1),TC(1)
68.		GO TO 20
69.	44	READ(5,104) FM(J,I)
70.		WRITE(6,308) FM(J,I)
71.		READ(5,104) EA(J,I),EB(I)
72.		WRITE(6,312) EA(J,I),EB(I)
73.	20	CONTINUE
74.	100	CONTINUE
75.		RETURN
70.	10	FORMAT (546,415)
77.	90	FORMAT(615)
78.	102	FORMAT(E10.4, I10)
79.	103	FORMAT(6E12.5)
80.	104	FORMAT (3E10.4)
81.	301	FORMAT(///12X,18H**SCATTERER NUMBER , 13, 2H, 546, 2H**
82.	3011	FORMAT ( THOMEDIUM . 12, 9H, ELEMENT , 12
83.	303	FORMAT (//12X.18HDISCRETE LEVEL(S) /
84.	304	FORMAI (3400 ( .12, 1H) .1PE12.5, 3H EV
85.	305	FORMAT (SOHOE (FV), SIGMAS (BARNS) IN PAIRS, E (HIGH TO LOW) /
80.		(3(192512.5.23))
87	306	FORMAT (//12X.22HNON-UISCRETE LEVEL(S) /
84	307	FORMAT (2010) FLAW LOWER LIMIT . 19E12.5, 3H EV .6X. 4H LAW .12.
80	507	
90	308	FORMAT(14) WEIGHT FACTOR . 1PE12.5 )
01	300	FORMAT (FHORELOW . 10F12 5. 3H EV . 6X. 22HP(F.) CONSTANT OVER E /
21.	309	A ACHOR ( CT), D(F) IN DATOS, FI (HIGH TO LOW)
2.		
93.	310	SOLIPZEIZ, SIZATI SOLIPZEIZ, SIZATI SOLIPZEIZ, SIZATI SOLIPZEIZ, SIZATI
04.	310	
95.		EXPLATIONER - 10512 5. TH EV . 34-148 - 12-5. 104 (54)++1/2
90.	311	FURMATIZINA (IPEIZ, 5) 50 EV / SALAND (EZCIS) ING (EVITEI/Z )
97.		A 3X11HC / E12.5
90.	312	FURMALISHUAL FIREL2.5F SH EV F SAFENDL FEL2.5
44.		E ND

1.		SUBROUTINE INELIN
2.	C	READS INPUT FROM TAPE 5 (STANDARD)
3.		COMMON /INFLC/ SPILP, VP, WP, E, S, NINLS, IN.
4		A NR(10) . Q(0.5) . NE (0.5) . FL (125.8.5) . SIGL (125.8.5) .
5.		B NU(10) .FLAW(2.10) .NLAW(2.10) .NFNG(1) .FNG(1.1).
5.		C NETAH1(1,1), ETAB1(1,1,1), DTAB1(1+1+1),
7		0  NETABO(1)  ETABO(1,1)  OTABO(1,1)  OTABO(1
<b>!</b> •		F = F(0, 10) = F(0, 10) = F(10) = TP(10) = TP(
0.		E = FM(2)(0), EA(2)(0), EB(10), F(2, 10)
9.		F NAPP(10/,NMEDE(2/10/,MELE(2/10/
10.		DIMENSION COMM(12), NALAW(5)4/, CADEL(5)
1.		DATA (NALAW(1,1),121,5)/ JOH TABULATED EV VS P(EV)
12.		DATA (NALAW(1,2)) 1=1,5)/ JUH TABULATED EV/E VS P(E)/E)
13.		DATA (NALAW(1,3),1=1,5)/ JOH EVAPORATION MODEL
14.		DATA (NALAW(1,4/)IE1,5// SON EVEALTBIE
15.		READ (5,101) COMMININES
16.		WRITE (6,201) COMMININES
17.	101	FORMAT(11AG, A4, 110)
18.	201	FORMAT(1H1, 36X, 11A6, A4 //
19.		A 36X, 30HNUMBER OF INELASTIC SCATTERERS (13
20.		DO 100 I=1,NINLS
21.		READ(5,10) LABEL, NR(I), NU(I), NAPP(I)
22.		
23.		READ(5,90) (NMEDE(J)I), NELE(U)I), OTINAP
24.		WRITE(6,301) I,LABEL
25.		
26.		WRITE(6,3011) NMEDE(J,I), NELE(J,I)
27.	23	
28.		IF (NR(1), EQ.0) GO TO 15
29.		
30.		
3		PEAD(b, 100)  o(1, T)  NE(1, T)
34		
33.		WRIE(0)047 0700777
34.		
33.		
37		RE-NE(011) READ(5.103) (EL(K.1.1), ETGL(K.1.1), K-1.KE)
3		WDITE(6.305) (EL(K,J,I),SICL(K,J,I),K=1,KE)
30.	5	
<b>U9</b>	5	
40.	15	
4.2	10	WOITE(4.304)
42.		
4 <b>.</b> .		$READ(5,102) = FLAW(J,I) \cdot NLAW(J,I)$
45		
46		WEITE(6.307) FLAW(.1.1).NL. (NALAW(L.NL).L=1.5)
40.		GO TO (11, 22, 33, 44). Ni
4.8	11	READ(5,102) EM(J,I), $NENG(I)$
49	**	WRITE(6,308) FM(J,I)
50		NEL=NENG(I)
51.	16	DO 17 K=1.NEL
52.		READ(5,102) ENG(K,I), NETAB1(K,I)
53.		NTAB1=NETAB1(K,I)

1. 2. 3. 5. 5. 7. 8. 9. C 10. 20	SUBROUTINE KINNY(SPDSQ,U,V,W,WATE,NMED,LELEM) COMMON /INELC/ SP,UP,VP,WP,E,S,NINLS,IN, A NR(10),Q(8,5),NE(8,5),EL(125,8,5),SIGL(125,8,5), B NU(10),ELAW(2,10),NLAW(2,10),NENG(1),ENG(1,1), C NETAB1(1,1),ETAB1(1,1,1),PTAB1(1,1,1), D NETAB2(1),ETAB2(1,1),PTAB2(1,1), E FM(2,10),EA(2,10),EB(10),TA(10),TB(10),TC(10), F NAPP(10),NMEDE(2,10),NELE(2,10) ALPHA, BETA ) CALL AB(ALPHA,BETA,LELEM,NMED)
11. C	CHUOSE SCATTERER INDEX
+DIACNOSTIC+	DO 30 INCLININGS
13.	INE DO-INDEX, INVIS DEFINED WITHIN THE LOOF.
14.	NAP=NAPP(IN)
15.	D0 25 J=1,NAP
16.	IF (NMEDE (J, IN), EQ. NMED, AND, NELE (J, IN), EQ. LELEM) GO TO 40
17. 25	CONTINUE
18. 30	CALLEBRED
20 40	CALL ERROR
21.	S=S0RI(SPUS0)
22. C	NEW DIRECTION COSINES
23.	CALL DCS(U, V, W)
24.	IF(NR(IN), EQ.0) GO TO 150
25. C	RESOLVED LEVELS
26.	CALL LEVELS(SPDSQ, BETA)
27.	U=ALPriA+U+UP+SP
28.	VEALPHIA*V+VP*SP
29.	
31 6	
32. 150	CALL FLAWS(SPDSQ+WATE)
33.	U=UP*SP
34.	V=VP+SP
35.	w=wP*SP
36. 300	SPUSQ=U**2+V**2+W**2
*DIAGNOSTIC*	THE TEST FOR EQUALITY RETWEEN NON-INTEGERS MAY NOT BE MEANINGFUL.
57.	IF (SPDSQ.EQ.U.O) CALL ERROR
30.	E LURIN E LUR
59.	
1.	SUBROUTINE AB(A/B/L/N) Common (UDE) EN ( AL BUA(Boun)) BETA(30, 8) ( E1(30, 8)
2.	COMMON/MULLEM/ ALPHA(30)8/, DETA(30)8/, LT1(30)8/
5.	B-AUC/RETA(L.N))
5.	RETURN
5.	END

Input to the inelastic routine is given in the following table:

Card	Format	Symbol	Description
1	15	NINLS	Number of inelastic scatterers (separate nuclides - each nuclide can have several inelastic processes).
Repeat	Cards 2-6	for <i>i</i> = 1, NINLS	
2	615	I	Scatterer number
		NID(I, 1)	Nuclide number (LELM) in which I <sup>th</sup> scatterer is found.
		NID(I, 2)	Medium number in which [ <sup>th</sup> scatterer is found.
		NID(I, 3)	Nuclide number in which I <sup>th</sup> scatterer is found.
		NID(I, 4)	Medium number in which I <sup>th</sup> scatterer is found.
			(Two pairs of nuclide and medium numbers are read in case the same nuclide is found in several places.)
		NR(I)	Number of resolved levels
		NU(I)	Number of unresolved processes
		NE(I)	Number of energies in level cross section tables (same energy mesh for all processes of same nuclide) NE(I) = 0 for 1 level
3	7E10.4	(EVAP(I, J)),	Energy for switching from one evap- oration-like process to another, both represented by the same level cross section table.
	ТА(І, Ј ТВ(І, Ј ТС(І, Ј	), (cm/sec) <sup>2</sup> ), (cm/sec) <sup>2</sup> ∕eV ), (cm/sec) <sup>2</sup> /eV	Coefficients for expressing the energy dependence of the nuclear temperature, $T = TA + \sqrt{E} \cdot TB + E \cdot TC$ .
	F	M(I, J)	Factor multiplying weight of neutron produced in J <sup>th</sup> process.

Card	Format	Symbol	Description
		EA(I, J), EB(I, J),	Coefficients for expressing the final energy in terms of the initial energy for process J when initial energy is less than EVAP(I, J), $E_{final} = EA + E_{initial} \cdot EB$
		(J=1, NU(I)+NR(I))	
4	7E10.4	(Q(I, J), J=1, NR(I) +NU(I))	Q values for level J. Q is read for both resolved and unresolved pro- cesses, with $Q = 0$ to signal an unresolved process.
5	7E10.4	(EL(K, I), K=1 NE(I))	Energy table for cross section tables for scatterer I. (omit if NE(I) = 0).
6	7E10.4	(SIGL(K, J, I), K=1, NE(I)) (J=1, NR(I)+NU(I))	Level cross section tables (omit if NE(I) = 0)

## APPENDIX C

# SUBROUTINE SOURCE LISTING

00101	1.		SUBROUTIN SUBRCE (SPOSG.U.V.W.X.T.Z.WATENINMEMINEDIVEC
00103			CINENSION THET(21), PTHET(21), WTHET(20), ERNOS(41), POPE(41)
00103	3.		1E T (40), COSTH(21), DELMOP (20), DELLOP (40)
00104	4.		
00106	5.		
00110			2. An 145 . X1. Y1. 71
00116			READ 150. THET. MEGH
001.2	4.		HEGKP = HEGR-1
00123	10.		WTHEP="THET".
00124	11.		READ 145, (THET(I), 1=1, MTHET)
00132	14.		READ 145. (PIHET(1). I=1.MTHET)
00140	15.		READ 145, (WIHET(I) 1=1, MTHEP)
00146	14.		READ 145. (EBNDS(J). JEI, MEGR)
00154	15.		READ 145, (POFE(J), JE1, MEGHD)
00162	10.		REAU 145, (CAT (U), DEI, MEORE)
00170	1		C(CTH(1) = C(C(TH(T(1))))
00174	14	10	CONTINUE
00176	2		10 11 I=2.MIHET
00201	21.		UELAUP(I)=(CUSTH(I)-COSTH(I-1))/(PTHET(I)-PTHET(I-1))
00202	24.	11	CONTINUE
00204	23.		0015 J=2, EGR
00207	24.		DELLOP(J)= ( EANDS(J)- EUNDS(J-1)) / (POFE(J)-POFE(J-1))
00210	25.	15	CONTINUE
00210	20.	C S	ELECT THETA FROM CUM THETA
00212	21.	16	HEFLINNF (BUMP T)
00215	ANY YND	TTNA	THEY NAME HORACT Y BOLY FROM THE COURSE AND THE TOP OF
BOCH K	XXX		K-XX
00217	30.		IF (PTHET(1)-R) 20:30,40
00222	31.	20	CUNTINUE
00224	34.	30	ToT=ToET(1)
00225	33.		WATE="THET(I)
00226	34.		COSTHI = COSTHIL
00227	35.		GO TO AD
00227	30.	C 1	TERPOLATE DETALEN THETA HOUNDS
00230	37.	40	COSTHIT = (R - P (HET(I - 1)) + UCE MOP(1) + COSTH(1 - 1)
00231	30.		
00232	59.	c	SELECT PHI
002.13	41.		CALL AZIAN(SINPHI, COSPHI)
00233	44.	C	UTRECTION COSINES
00234	43.		U=COSTHT
00235	44.		V=COSPHI*SINTHT
00236	43.		*=SINPHI+SINTHT
00237	40.		X=X1+U
00240	41.		Y=Y1+V
00241	40.	~	CARCIEV HUE, FOR LUC UNISPHERE
00241	49.	Ľ	IL (14 98081) B1.81.82
00245	5		NPFG=1.0
00246	5.		GO TO 85
00247	53.	8	IF (U+.93969) 83.83,84
00252	54.	6	NREG=2
00253	50.		GO TO 65
00254	50.		NREGES
00254	57.	C	SELECT E FROM COM F
00255	50.	0	DOLOG JED FEP
00250	SY.	XXXXXX	THE REAL SOLVER AS A REAL REAL REAL REAL REAL REAL REAL RE
DENXX	XXX		XXX
00262	61.		IF (POFE(J)-R) 100,120,130
00265	64.	10	CONTINUE
00217	6	12	D ELERE EDNUS(J)
60270	64.		WATE=MATE EAT (J-1)
00271	63.		GO TO 140
00271	60.	С	INTERPOLATE BETWEEN ENERGY BOUNUS
00272	61.	13	ENER = (R-POFE(J-1))+DELLUP(J) +EBNDS(J-1)
00273	60.		WATE WATE + LWT (J-1)
00274	67.	14	5 SPUSGE ENER • 1.913220092212
00275	10.	14	5 FURMAL (7E10-27
00270	7	15	RETIDA
19300	7.		ELD
	1		

C-1

### APPENDIX D

# ACTIFK USER ROUTINES

00101	1.	SUBROUTINE STATCH
00103	2.	COMMON/NEW/MAXE, MAXX, MCSLB, NEUB., NDXX , ELB, SANOR, XLO, XU
00104	3.	COMMON/NEW/CSLB(10),DELE(30),ELn(25),ENOR(25),EUB(30),
00104	4.	1FL2X1(1,10,3n),FL2X2(1,1n,30),FLUX(1,10,30),FLUX2(1,10,3n,10),
00104	5.	25IGMAP(30)+SIGMAU(30)+SIGMAS(15.25)+SIGMA2(1+10+30)+SRS(15+25)+
00104	ь.	35RS2(15,25),STER(1n),TCSLB(10),UNFLUX(30),UN1FL(30,10),UN2FL(30),
00104	7.	4xPFLUX(30),XP1FL(30,10),XP2FL(3,),XX(15),SR51(15,25,10)
00105	8.	COMMON/UNCOL/UNDOSE(10)
00106	9.	COMMON/DET/NODET,XD(10),YD(10),7D(10)
00107	10.	COMMON/CPLIST/NCOLL(1),NAME(1),S12(1),U(1),V(1),W(1),
00107	11.	<pre>A X(1),Y(1),Z(1),WATE(1),SPOLD(1),UOLD(1),VOLD(1),WOLD(1).</pre>
00107	12.	B OLDWT(1),NGRP(1),LELEM(1),NMED(1),DUM(1A)
00110	13.	COMMON/ASINGL /NSTRT+NITS+NBIN.NETAPE,ETOP+EBOT+ECUT+NXTAPE+NYTAP
00110	14.	1E,NFTAP1, NFTAP2, NFTAP ,MEDIA.NHISTR,NHISMX,NWPCOL,NSGP,NCOLPR.N
00110	15.	2ANISO ,NDSGP,NLAST,KTH,NGRUUP,LBATCH,NVAR,NF,NL,IB,NCP5R2,NCPNGP,
00110	16.	3NCPELM ,NCPMED,NTYPE,DOSE,NRSUM,NZRO,NBSUM,NYTABL ,NGEOM,NM,MGZ,NO
00110	17.	4NEUT+NYSUM+NZR+IVAR
00111	18.	COMMON STORAG(7000)
00112	19.	DATA IFIRST/0/
00114	20.	IF(IFIRST) 1,1,2
00117	21.	1 IFIRST=1
00120	22.	CALL OSRSET(NHISTR, NHISMY, NWPCO, , NCOLPR)
00121	23.	READ(5,100) NODET,(XD(I),YD(I),70(I),I=1,NODET)
00132	24.	NDXX=NODET
00133	25.	100 FORMAT(15/(6E10.5))
00134	26.	READ(5,10) NEUBM, MCSLB, ELB, SANOD
00142	27.	10 FORMAT(216,2E12.A)
00143	28.	READ(5,20) (EUB(MDE),MDE=1,NEUBu)
00151	29.	READ(5,20) (DELE(MDE), MDE=1, NEUBM)
00157	30.	READ(5,20) (TCSLB(NEWMU).NEWMU=1.MCSLB)
00165	51.	READ(5,20) (CSLR(NEWWO))NEWMO=1.MCSLB)
00173	32.	20 FORMAT(BE12,R)
001/4	33.	READ(5)30) MARE, WARE, ALU, AU
00202	34.	30 FORMAT(216)2E6.4J
00203	35.	
00211	30.	READ(5:20) (XX(1),1=1,MAXA)
00217	30	READ(SIZU) (ENUR(J)/J-I/MAAE)
00225	30.	WRITE (0)101/ NUDETFLAUTT/JULIT/JULIT/JIETFNUDET/
00236	39.	TOT FORMATIZATIONE NU OF DEFENDERS IS 12757HUTHE ATTE COURTINATES AR
00235	40.	
00257	45	
00242	42.	
00245	44	
00250	45	
00251	45	DU EUU ALIMMAA
00255	47.	
00261	46.	
00264	49.	
00267	50.	
00273	51.	
00276	52.	
00277	53.	
00301	54.	DETURN
00302	55.	END

D-1

Identification of Input Read by STBTCH for use in ACTIFK uses routines. This input follows the normal ACTIFK input.

Card 1	16	NEUBM	Max number of energy upper bounds	
	16	MCSLB	Max number of lower bounds of $\cos \theta$ (polar angle at exit face)	
	E12.8	ELB	Lower bound of energy group structure	
	E12.8	SANOR	Solid angle normalization factor to account for source generating angles	
Card 2	6E12.8	EUB	Upper energy bounds	
Card 3	6E12.8	DELE	Energy group normalization constants. In this calculation DELE = $\Delta E/E$ for each group	
Card 4	6E12.8	TCSLB	Average angle for polar angle increments	
Card 5	6E12.8	CSLB	Cosine of upper angle increment bound (Lower value of $\cos \theta$ for the increment)	
Cards 6 through 9 pertain only to the 1 ev source calculation.				
Card 6	16	MAXE	Number of energy groups	
	16	MAXX	Number of thickness increments	
	E6.4	XLO	Lowest value of x increment bound	
	E6.4	XU	Highest value of x increment bound	
Card 7	6E12.8	ELO	Lower bounds of energy groups in ev high to low	
Card 8	6E12.8	xx	Values of upper bounding x planes low to high in cm	
Card 9	6E12.8	ENOR	Normalization factors to be applied to energy groups.	
00101	1.	SUBROUTINE SDATA		
-------	-----	--		
00103	2.	COMMON/NEW/MAXE, MAXX, MCSI B, NE, IB., , NDXX , ELB, SANOR, XLO, XU		
00104	3.	COMMON/NEW/C5L3(10), JELE(30), ELO(25), ENOR(25), EUB(30),		
00104	4.	1FL2X1(1,10,30),FL2X2(1,10,30),FLUX(1,10,30),FLUX2(1,10,30,10),		
00104	5.	2513MAP(30),SIGMAU(30),SIGMAS(15.25),SIGMA2(1,10,30),SRS(15,25),		
00104	6.	35R52(15,25), STER(10), TCSL8(10), UNFLUX(30), UN1FL(30,10), UN2FL(30),		
00104	7.	4xPFLUX(3c),XP1FL(3n,10),XP2FL(3n),XX(15),SR51(15,25,10)		
00105	ь.	COMMON/CPLIST/NCOLL(1),NAME(1),512(1),J(1),V(1),W(1),		
00105	9.	A X(1),Y(1),Z(1),WATE(1),SPOLD(1),UOLD(1),VOLD(1),WOLD(1).		
00105	10.	B OLDWT(1),NGRP(1),LELEM(1),NMED(1),DUM(1A)		
00106	11.	COMMON/UNCOL/UNDOSE(10)		
00107	12.	COMMON/DET/NODET, KDET(10), YDET(10), ZDET(10)		
00110	13.	COMMON/ASINGL /NSTRT, NITS, NB17. NETAPE, ETOP, EBOT, ECUT, NXTAPE, NYTAP		
00110	14.	1E,NFTAP1, NFTAP2, NFTAP ,MEDIA,NHISTR,NHISMX,N#PCOL,NSGP,NCOLPR,N		
00110	15.	ZANISO , NOSGP, NLAST, KTH, NGROUP, BATCH, NVAR, NF, NL, IB, NCPSR2, NCPNGP,		
00110	16.	SUCPELM INCPMEDINTYPE, DOSFINRSUM. NZROINBSUM, NYTABL INGEOMINMIMGZINO		
00110	17.	4NEUT, NYSJM, NZR, IVAR		
00111	10.	SPD2=512(KTH)		
00112	19.	IF(SPD2-ECUT) 2,1,1		
00115	20.	1 XA=X(KTH)		
00116	21.	YA=Y(KTH)		
00117	22.	2A=2(KTH)		
00120	23.	EJC= 5P02/1.913220092E12		
00121	24.	DO 3 MDE=1.NEJBM		
00124	25.	IF(EUB(MOE)-EUC) 3,4,4		
00127	20.	3 CONTINUE		
00131	27.	4 XD= XDET(1)		
00132	28.	YD= YDET(1)		
00133	29.	7D= 7DET(1)		
00134	30.	A=XD-XA		
00135	31.	BEYD-YA		
00136	32.	C=ZD-ZA		
00137	33.	SD2=A+A+B+B+C+C		
00140	34.	SD=SQRT(SD2)		
00141	35.	CALL EUCLID (61.0.0000050.5P-2.ARG,0)		
00142	36.	JNFLJX(MDE) =JNFLUX(MDE) + WATE(KTH)+(6.55E-3)+EXP(ARG)		
00143	37.	2 RETURN		
00144	38	END		

00101	1.	SUBROUTINE RELCOL
00103	2.	COMMON/CPLIST/NCOLL(1),NAME(1),-12(1),U(1),V(1),W(1),
00103	3.	A X(1),Y(1),Z(1),ATE(1),SPOLD(1),UOLD(1),VOLD(1),WOLD(1).
00103	4.	B OLDAT(1), NGRP(1), LELEM(1), NMED, 1), DUM(1A)
00104	5.	COMMON/DET/NODET, XDET(10), YDET(10), ZDET(10)
00105	ь.	COMMON/AMEDEL /ALFA(32, A), BETA(32, B), ALPBET (32, B),
00105	7.	1LF1(32,8),ASSES(32,8)
00106	ы.	COMMON/ASINGL /NSTRT, NITS, NAIN, NETAPE, ETOP, EBOT, ECUT, NXTAPE, NYTAP
00106	9.	1E,NFTAP1, NFTAP2, NFTAP ,MEDIA,NHISTR,NHISMX,NWPCOL,NSGP,NCOL
00106	10.	2ANISO INDSGPINLASTIKTHINGROUPILRATCHINVARINFINLIIBINCPSAZINCPNSP,
00106	11.	SNCPELM .NCPMED, NTYPE, DOSE, NRSUM, NZRO, NASUM, NYTABL , NGEOM, NM, MGZ, 10
00107	12.	COMMON/NEW/MAXE, MAXX, MCSI, B, NEUBM, NDXX , ELB, SANOR, XLO, XU
00110	13.	COMMON/NEW/C5LB(10); DELE(30); ELn(25); ENOR(25); EUB(30);
00110	14.	1FL2x1(1,10,30),FL2x2(1,10,30),FLUX(1,10,30),FLUX2(1,10,30,10),
00110	15.	25IGMAP(30),5IGMAU(30),SIGMAS(15,25),SIGMA2(1,10,30),SRS(15,25),
00110	16.	35R52(15,25),STER(10),TCSLB(10),UNFLUX(30),UN1FL(30,10),UN2FL(30).
00110	17.	4xPFLUX(30), XP1FL(30,10), xP2FL(3,), XX(15), SRS1(15,25,10)
00111	18.	NN=NAME(KTH)
00112	19.	xA=x(KT→)
00113	20.	YA=Y(KTH)
00114	21.	7A=2(KTH)
00115	22.	WT1=WATE(KTH)
00116	23.	IF( 512(KTH)=1.91322E12) 1,10,40
00121	24.	1 IF(SPOLD(KTH)-1.91322E12) 10,2,2
00124	25.	2 IF(XA-XLO) 10,3,3
00127	26.	3 IF(XA-XU) 4.4.10
00132	27.	4 RAIYA+YA+ZA+ZA
00133	28.	IF(RA-100.) 5,5,10
00136	29.	5 D0 6 I=1,MAXX
00141	30.	IF(XX(I)-XA) 6.7.7
00144	31.	6 CONTINUE
00146	32.	7 EVA= 512(KTH)/1.913220092E12
00147	33.	D0 8 J=1,MAXE
00152	34.	IF(ELO(J)-EVA) 9,9,8
00155	35.	A CONTINUE
00157	36.	9 SRS(I,J)=SRS(I,J)+WT1
00160	37.	10 IF(XA) 400,400,11
00163	38.	11 SB=SQRT(SPOLD(KTH))
00164	39.	JELEMELELEM(KTH)
00165	40.	MED=NMED(KTH)
00166	41.	AT1=#ATE(KTH)

			RELCOL/ S
00167	42.		XMASS=ASSES(JELEM.MED)
00170	43.		DU 1HO I=1.NODET
00173	44.		XD=XDET(I)
00174	45.		YD=YDET(I)
00175	46.		ZD=ZDET(I)
00176	47.		A=XD=XA
00177	48.		B=YD-YA
00200	49.		C=ZD=ZA
00201	50.		502=A+A+B+B+C+C
00202	51.		SD=SQRT(SD2)
00203	52.	16	COSLB=(A+UOLD(KTH)+B+VOLD(KTH)++++OLD(KTH))/(SD+SB)
00204	53.	-	IF(XMASS) 35,35,30
00207	54.	30	CALL ELAS(COSLB, VA2, FMU)
00210	55.		GO TO 40
00211	56.	35	CALL NONLAS (COSLR. VA2. FMII)
00212	57.	40	IF(FMU) 180,180,45
00215	58.	45	EVA=VA2/1.913220092E12
00216	59.		1F(EVA-ELB) 180,50,50
00221	60.	50	CALL EUCLID (XA, YA, 7A, XD, YD, ZD, Sn, VA2, ARG. 0)
00222	61.		COSE=A/SD
00223	62.		DO 60 NEWMU=1+MCSLR
00225	63.		IF(CSLB(NEWMU)-COS E) 65,65,60
00231	64.	60	CONTINUE
00233	65.	65	NO 70 MDE=1.NEUBM
00236	66.		IF(EUB(MDE)-EVA) 70,75,75
00241	67.	70	CONTINUE
00243	68.		IF(1SD2) 75,75,73
00246	69.	73	502=0.4
00247	70.	75	FLUX(I,NEWMU,MDE)=FLUX(I,NEWMU,nE)+WT1+EXP(ARG)+(FMU/SD>)
00250	71.	180	CONTINUE
00252	72.		IF((YA**2+ZA**2)=25.) 200,200,4.0
00255	73.	500	CUSLB=-UOLD(KTH)/SR
00256	74.		xD=0.
00257	75.		YD=Y(KTH)
00260	76.		7D=2(KTH)
00261	77.		SD=X(KTH)
00262	78.		IF(XMASS) 220,220,210
00265	79.	210	CALL ELAS(COSLB,VA2,FMU)
00266	80.		GO TO 240.
00267	81.	220	CALL NONLAS(COSLB.VA2.FMII)
00270	82.	240	IF(FMU) 400,400,245
00273	83.	245	IF(VA2-ECUT) 400,250,250
00276	84.	250	CALL EUCLID (XA, YA, ZA, XD, YD, 70, PD, VAP, ARG, 0)
00277	85.		EVA=VA2/1.913220099E12
00300	86.		NO 270 MDE=1, NEUBM
00303	87.	_	IF(EUB(MDE)-EVA) 270,275,275
00306	88.	270	CONTINUE
00310	89.	275	XPFLUX(MDE)=XPFLUX(MDE)+WT1*EXP/ARG)*FMU/7A.53975
00311	90.	400	RETURN
00512	91.		END

(010)	14	SUBROUTINE ENDBAT
0103	2.	COMMON/ASINGL /NSTRT, NITS, NAIN NETAPE, ETOP, EBOT, ECUT, NX+APE, NYTAP
0103	3.	1E.NFTAP1, NFTAP2, NFTAP MEDIA.NHISTR.NHISMX,NWPCOL,NSGP,NCOLPR,N
0103	4.	2ANISO , NDSGP, NLAST, KTH, NGROUP, LRATCH, NVAR, NF, NL, IB, NCPSR2, NCPNGP,
00103	5.	3NCPELM INCPMED. NTYPE. DOSE NRSHM. NZRO, NBSHM. NYTABL INGEOM, NM, MGZ. NO
00103	b,	UNEUT INYSUMINZRI IVAR
00104	7.	COMMON STORAG(1)
10105	1.	COMMON/NEW/MAXE, MAXX, MCSI B. NEUB., NODET. EL B. SANOR, XLO. XII
00100	9.	(DVMDN/NE#/CSLB(10), DELE(30), FL0(28), END2(25), FUB(30),
00106	10.	IFL2X1(1,10,30),FL2X2(1,10,30),FLUX(1,10,30),FLUX2(1,10,30,10),
00105	11.	251344P(30), STGVAU(30), STGMAS(15, 05), STGVA2(1,10,30), SPC(15,25),
00100	12.	35652(15.25) STF2(10) . TCSI B(10) . UNFLUX(30) . UN1FL (30.10) . UN2FL (30)
00100	13.	4 2 PEL (12 (30) + 201 EL (30, 10) + 202 EL (3, 1, 22 (15) - 5051 (15, 25, 10)
00107	14	
00111	15.	
10111	16	
00115	17	STER (REMOVE (SED(REMOVE/-GALA(REMOV/)EEF 3.14134
00113	14	
00120	10	STERITIA "USEST
00120	29.	
00123	20,	DO GO NEAMOET MCSER
00120	210	
1013)	de.	30 FLUX2 (I.NEWWU, MDE.LAATCH) = FLUC(I, NEWMU, MDE) = (SANOR/NSTRT)/
00131	2.5.	1 (STER (NEAMU) EDELE (MDE/)
00133	24.	SG CONTINUE
00135	25.	SO CONTINUE
00137	26.	NO 55 MDE=1,NEJBM
00142	27.	XPIFL(MDE+LBATCH) = XPFLUX(MDE) + SANDR/(NSTRT +DELE(MDE))
00143	28.	55 JNIFL(MDE, LBATCH) = UNFLUX(MDE)/(DELE(MDE)+NSTRT)
00145	27.	DO BO IEI,MAXX
0150	30.	DO 60 JEI, MAXE
00153	31.	50 SRS1(I,J,LBATCH)= SRS(I,J)* SAUGR / (NSTRT * ENOR(J))
00155	.32,	BO CONTINUE
00157	.33.	KRITE(6,100) LBATCH
00162	34 .	100 FORMAT(141,50X,7HLRATCH= 14/50X.
001.62	35.	A35HNEJTRON SOURCE FALLING THROUGH 1 EV
10162	36.	1/50X, 31HNEUTRONS/CM3 -SOURCE NEUTRON
00105	37.	2/1H0,9X,12HLOWER ENERGY,30X,17HDEL X UPPER LIMIT)
00163	30.	18=1
00164	39,	IESMINO( aAXX,7)
00165	40.	105 #RITE(6,110) (XX(I),I=IB,IE)
00175	41.	DO120 J=1.MAXE
00176	«2.	120 WRITE(6+130) ELO(J)+(SRS1(I+J+L=ATCH)+I=IB+IE)
00206	43.	110 FORMAT(1H0+20X,1P7E15.5)
00207	144 .	30 FORMAT(1H0+5X+1PAE15.5)
0226	45.	TF(MAXX.EQ.IE) GO TO 140
00212	40.	18=8
00213	47.	IE=MAXX
00214	48.	60 TO 105
11500	49.	140 CONTINUE
00216	50.	DU 240 I=1.NODEY
15500	51.	WRITE(6,200) I
00220	52.	200 FORMAT(1H1+40X,17HENERGY ANGLE -LUX
00556	53,	2/20X 52H (NEUTS /CH2-STERADIAN-MEV-SOURCE NEUTRON) X EAVE

#### ENDBAT/

5

00224	54.	3/20X, 34H(AS VIEWED BY COLLIMATER DETECTOR)
00224	55.	2/40X,11HDETECTOR NO,I3
00224	56.	3/1H0,5X,12HUPPER ENERGY,15X,15H ANGLE(AVERAGE))
00225	57.	JB=1
00226	58.	JE=MINO(MCSLB;7)
U0227	59.	205 WRITE(6,210) (TCSLR(J), J=JB, JE)
00235	60.	210 FORMAT(1H0,13X,1P7E14.5)
00236	61.	DO 220 MDE=1.NEUBM
00241	62.	220 WRITE(6,230) EUB(MOE), (FLUX2(I, ,,MDE,LBATCH), J=JB, JE)
00251	63.	230 FORMAT(1X, 1PAE14.5)
00252	64.	IF(JE.EQ.MCSLB) GO TO 250
00254	65.	JB=8
00255	66.	JE=MCSLB
00256	67.	GO TO 205
00257	68.	240 CONTINUE
00261	69.	250 CONTINUE
00262	70.	wRITE(6.300)
00264	71.	JUD FORMAT(1H1, 30X, 32HNORMALLY EXIT, NG UNCOLLIDED FLUX
00264	72.	2/20X,52H(NEUTS /CM2-STERADIAN-JEV-SOURCE NEUTRON) X EAVE
00264	73.	3/20X, 34H(AS VIEWED BY COLLIMATED DETECTOR)
00264	74.	2/1H0.5X,12HUPPER ENERGY, 5X,4HFLUX)
00265	75.	DO 310 MDE=1.NEJBM
00270	76.	310 #RITE(6,320)EUB(MDE), UNIFL(MDE, BATCH)
00275	77.	320 FORMAT(1X, 1P2E15.5)
00276	78.	wRITE(6,400)
00300	79.	400 FORMAT(1H1+30X.30HNORMALLY EXIT+NG COLLIDED FLUX
00300	80.	1/20X+52H(NEUTS /CM2-STERADIAN-MEV-SOURCE NEUTROND & EANT
00300	81.	2/1H0.5X.12HUPPER ENERGY. 5X.4HFL UX)
00301	82.	WRITE(6,410) (EUB(K), XP1FL(K, LBATCH), K=1. NEUBAN)
00310	83.	410 FORMAT(1X,1P2E15.5)
00311	84.	11 RETURN
00312	85.	END

00101	1.	SUBROUTINE ENDRUN
00103	2.	DIMENSION NSTOR(1)
00104	3.	COMMON STORAG(1)
00105	4.	EQUIVALENCE(NSTOR(1),STORAG(1))
00105	5.	C **** TEWARE - THIS ASINGL IS NOT THE SAME AS OTHERS
00106	b.	COMMON/ASINGL /NSTRT, NITS, NBIN NETAPE, ETOP, EBOT, ECUT, NXTAPE, NYTAP
00100	7.	15.NFTAP1, NFTAP2, NFTAP , MEDIA, WHISTR, WHISMX, NMPCOL, NSGP, NCOLPR, N
00105		24NISO ,NDSGP, NLAST, KTH, NGROUP, BATCH, NVAR, NF, NL, IB, NCPSH2, NCPNGP,
00105	ÿ.	SNCPELM INCPMEDINTYPE DOSFINES M. NZROINAS MINYTABL INGEOMINMINGZINO
U010n	10.	4 JEUT NYSUM NZR IVAR
00107	11.	COMMON/NEW/MAXE, MAXA, MCSLB, NEIBU, NODET, ELB, SANOR, XLO, XU
00110	12.	CUMMON/NE#/CSLB(10), DELE(30), ELn(25), ENOR(25), EUB(30),
00110	13.	1FL2X1(1,10,30).FL2X2(1,10,30),FL1X(1,10,30),FLUX2(1,10,30,10).
00110	14.	251GMAP(30), SIGMAU(30), SIGMAS(15, 25), SIGMA2(1, 10, 30), SRS(15, 25),
00110	15.	35R52(15,25),STER(10),TCSLB(10),UNFLUX(30),UN1FL(30,10),UNPFL(30).
00110	16.	4xPFLJX(30), xP1FL(30,10), xP2FL(30), xX(15), SRS1(15,25,10)
00111	17.	COMMON/NEw2/XP2FLS(3n), UN2FLS(3n), FL2X2S(1,10,30), SRS2S(15,25)
00112	18.	DO 30 MDE=1.NEUBM
00115	19.	DJ 20 NEUMJE1, MCSLA
00120	20.	DO 10 I=1,NODET
00123	21.	FL2X1 (I.NEWMU, MDE)=0
00124	22.	10 FL2X2 (I.NEWMU.MDE)=0
00126	23.	20 CONTINUE
00130	24.	30 CONTINUE
0013e	25.	DO 50 MDE=1,NEJBM
U0135	26.	xP2FL(MDE)=0
00130	27.	xP2FLS(MDE)=0.0
00137	28.	UN2FLS(MDE)=0.0
00140	29.	40 JN2FL(MDE)=0
00141	30.	50 CONTINUE
00143	31.	DO 100 ADEE1, NEURAM
00146	32.	DU 60 LBATCH =1.NITS
00151	30.	xP2FL(MDE)=XP2FL(MDE)+ XP1FL(MDE,LBATCH)/NITS
00152	34.	xP2FLS(MDE)=xP2FLS(MDE)+xP1FL(MDE,LBATCH) #xP1FL(MDE,LBATCH)
00153	35.	UN2FLS(MDE)=UN2FLS(MDE)+UN1FL(MDE+LBATCH)+UN1FL(MDE+LBATCH)
00154	36.	50 JN2FL(MDE)=UN2FL(MDE)+ UNIFL(MDE,LHATCH)/VITS
00156	37.	90 IIII.NODET
00161	38.	DO BO NEWMUEL, MCSLR
00164	39.	DO 70 LBATCH=1.VITS
00167	40.	FL2X2(1+NEWMU+MDE) =FL2X2(1+NEW-U),MDE++ FLUX2(1+NEWMU+MDE+LBATC4)/
00167	41.	
00170	42.	FLEXES(I)NEWNO,MDEJEFLEXPS(I)NEWNO,MDEJFFLUX2(I)NEWNO,MDEJLS41(H)
00170	43.	1472 TO THE MELLINE A STORY (T. MELLINES) A SULVA (TANEMULINE LI BATCH) -
00171	44.	A FLEAT(1) NERMUMBE/ SFLEAT(1) NEWMIMME/T FLUAZ(1) NEWMUMBE/EBATCH/T
001/1	40.	1SIEK(NEWWO/FDELE(MDE//NITS

00175	46.	BO CONTINUE	
00175	47.	JA CONTINUE	
00177	48.	100 CONTINUE	
00201	44.	10 130 AJE=1 NE.HW	
0020.	50.	TO 120 VEWMUEL WCSLA	
00207	51 -	00 110 I= 1. NODET	
U0212	52.	SIGMA2(1,1)EXMU, ADE) = SORT ((FL212-11.N)	WWU.MDE) /NITS) -
00212	53.	1 (FL2X2(I, NEW), MD	1
00213	54 -	110 CONTINUE	
40215	55.	120 CONTINUE	
44217	56.	SIGNAP(MOF) = SOUTH APPELS(MOF)	(NTTS) - (XP25: (HDE) ++21)
40220	57.	SIGNALI(NOF) = SORT( 112FI S(NOF)	(NITS) -(11N2E) (UDE) ++211
00221	56.	130 CONTINUE	FREISF - TOMEFET HOLF + 2/1
00221	59	100 COTTINCE	
00225	60	00150 1-1-MAXE	
00228	61	CHESCIT, DEC O	
00231	62	SH32511137-0.0	
00232	62.		
00233	63.		4
00236			
00240	05.	SIGMAS(1,J/=SURT((SRS2(1,J//NITC)=(SP	(25(1)))**2//
00241	00.	15h CONTINUE	
00243			
00245	68.	ARTIELLUT NEUDA, MUSLA, MAYA, MAXE	
00253	69.	NO 240 1=1.NODET	
00256	70.	ARITE(5+201) 1	
00261	1.	201 FURMAT (1H1 . 40X . 17HENERGY ANGLE PLUX	
00261	12.	2/20X, 52H (NEUTS /CW2-STERADIAN- FV-50	DURCE NEUTRON) X EAVF
00261	73.	1/45X,13HDETECTOR NO, I2,	and the second
00261	74.	2/2HO .12HUPPER ENERGY . SOX . 13HAVERAGE	ANGLE)
00262	75.	JB=1	
00263	76.	JEEMINO (MCSLR,7)	
00264	77.	205 *RITE(6.210) (TCSLR(J).J=JB.JE)	
00272	78.	210 FORMAT(15X,197E14.5)	
00273	79.	DO 225 MDE=1.NEURM	
0027ь	80.	ATTE(6,220) EUB(MDE), (FI 2X2(1, ., MDE)	(J=JB, JE)
00305	81.	WRITE(10) (FL2x2(I,J,MDE),J=JR,JE)	
00313	82.	220 FORMAT(1X, 1PAL14.5)	
00314	83.	225 WHITE(6,230) (SIGMA2(1, J. MOE). J-JA, JE	.)
00323	84.	231 FURMAT(15X+1P7E14.5)	
00324	85.	IF(JE.EG.MCSLB) GO TO 240	
00326	86.	Jd=8	
00327	87.	JEEMCSLO	
00330	88.	GO TO 205	
00331	89.	240 CONTINUE	
00333	90.	00 640 1=1.NODET	
00336	91.	*RITE(6.601) 1	
00341	92.	601 FORMAT(1H1+40X, 17HENERGY ANGLE -1 UK	
00341	93.	2/20X. 37HNEUTRONS/CM2-SOURCE NEUTRON-1	NCREMENT
00341	94.	3/45X,11HOLTECTOR NO .13	
00341	95.	4/1HO.12HUPPER EVERGY. 50X. 13HAVERAGE	NGLF)
00342	96.	J#=1	
00343	97.	JEEMIND (ACSLA,7)	
00344	96.	605 #RITE(6.210) (TCSLA(1).J=JA. (F)	
40352	99.	an has worst NE IAM	

00355	100.	625 #RITE(6,220) EUB(MDE), (FL2X1(I, ., MDE), J=JB, JE)
00365	101.	1F(JE.EQ.MCSLB) GO TO 640
00367	102.	JB=8
U0370	103.	JE=MCSLB
00371	104.	60 10 605
00372	105.	640 CONTINUE
00374	106.	#RITE(6:300)
00376	107.	300 FORMAT(1H1,40X, 32HNORMALLY EXIT.NG UNCOLLIDED FLUX
00376	108.	1/20X 52H (NEUTS /CH2-STERADIAN-UFV-SOURCE NEUTRON) X FAVE
00375	109.	2/140.124UPPER ENERGY.10X.4HELUX.10X.5HST (MA)
00.377	110.	RITE(6.310) (FUR (MOE) + UNOEL (MOE) + SIGNAU(MDE) + MOE=1 + NEURY)
00407	111.	310 FORMAT(1x, 1935)4.5)
00410	112.	-BITE(6.400)
00412	113.	HAN EORMAT (141 HOX, SOHNOPMALLY EVIT-NG COLLIDED ELLY.
00412	114.	2/20X 52H (NEUTS / Cu2-STEAD) AN-UEV-SOUDCE NEUTODN) X FAVE
00412	115.	2/140.124 (DEF) FYEREY 10Y (WELLY 10Y, SHETAYA)
00413	115	PITCHARD (CHRANTS) YOZE (UST) CRADINGS) NOE-1, NEURAL
00413	117	ATTECTAL (VOLUME), MOETANE, METANA (MOETANE INCLASS
00423	110	WITCHUS (APEPL(MICH) MICHINELAM /
00431	110	011 PURMAI(18/170113.5/
00432	120	
00434	121	SUB FURMATINISTAN SANNESTRUN SUBRE FALLING THROUGH I SV
00434	122.	1/30///SINNEURONS/CMS -SURRCE NEITRON
00434	122.	2/100/20LOWER ENERGY, 40A. IBH DEL X OPPER LIVIT
00435	123.	
00436	124.	
00437	125.	505 WRITE(6,510) (XX(J), (=38, 32)
00445	120.	510 FORMAT(15X)17/E14.5/
00446	127.	DO 525 JEI MAXE
00451	128.	wRITE(6,520) ELO(J),(SR52(I,J), = JB, JE)
00460	129.	wRITE(10) (SRS2(1,J),I=JR,JE)
00466	130.	520 FORMAT(1X,8E14.5)
00467	131.	525 WRITE(6,530)(SIGMAS(1,J),I=JB,Je)
00476	132.	530 FORMAT(15X+1P7E14-5)
00477	133.	IF(JE.EQ.MAXX) GO TU 540
00501	134.	J8=8
00502	135.	JEEMAX
00503	136.	GO TO 505
00504	137.	540 CONTINUE
00505	138.	200 RETURN
00506	139.	END

ENDRUN/

## APPENDIX E

# TABULATED DATA OF THE SPECTRAL MEASUREMENT

TABLE E1 COMPUTED 3-25-69

HUN 4 5-23-04

MEMISPHERE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH AIR COOLED DEPLETED URANIJM SPHERE BIAS = .0189 STANDARD LIGHT UNITS ENENGT RESULUTION FOR GROUPING = .05 LARGEST FRACTIONAL EPROR = .20

NEUT ON LILKOY	ACUTRON FLUX	FL X*ENERGY	RELATIVE
(-, )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.5470+07	3,5121-10	5.4353-03	1,4979-01
1.40:2.407	5.3253-10	7.4725-03	1.7607-01
1.31 0+07	0.2987-10	8.4373-03	1.7087-01
1.24,4+07	9.3325-10	1,1585-02	1,4111-01
1.17 10+07	1.1353-09	1,3292-02	1,2782-01
1.10.0+07	1.0906-09	1,2062-02	1.3337-01
1.0455+07	1.7490-09	1.8303-02	1.0898-01
9.9153+00	2.4972-09	2.8730-02	8.7644-02
9.17.5+06	4.2591-09	3,9079-02	5.3658-02
8.3049+05	7.6233-09	6.3311-02	4.3181-02
7.5026+00	1,2519-08	9.4551-02	3.6778-02
b.89 1+00	1.7231-08	1,1921-01	3.3859-02
0.3251+00	2.7341-08	1.7293-01	2.8615-02
5.86.1+05	3.2223-08	1.8756-01	2.8157-02
5.3740+06	4.1299-08	2.2195-01	2,6907-02
4.9774+06	4283-08	2.4032-01	2,6926-02
4.0220106	6.0213-08	2.7835-01	2.5740-02
4.3049+06	7.0628-08	3,0405-01	2.5116-02
4.01:6+05	8.0392-08	3,2306-01	2.4572-02
3.70 0+06	8.7102-08	3,2750-01	2.4366-02
3.5250+00	1.0408-07	3.6694-01	2.2958-02
3.31:4+00	1.0306-07	3.4138-01	2.3098-02
3.11.0+06	1.0299-07	3,2112-01	2.3336-02
2.9402+00	1.1519-07	3.3868-01	2.2552-02
2.7172+06	1.1743-07	3,2613-01	2.1353-02
2.0274+06	1.3029-07	3,5021-01	2.0566-02
2.4094+00	1.5043-07	3.8942-01	2.0536-02
2.30, 0+00	1.4541-07	3.4346-01	1.9845-02
2.2100+00	1.7843-07	3.9549-01	1.5894-02
2.00 :5+00	2.2415-07	4.6141-01	1.5058-02
1.9100+00	2.4236-07	4.6456-01	1,4611-02
1.7392+00	3.4344-07	6.1448-01	1.3466-02
1.6/ 0+06	2.9652-07	4.9637-01	1.4485-02
1.5095+00	3.7457-07	5.8789-01	1.3496-02
1.4745+00	4.4461-07	6.5558-01	1.2944-02
1.3079+36	4.5915-07	6.3725-01	1.3506-02
1.30.7+00	5.1326-07	6.7170-01	1.3171-02
1.2061+06	6.1906-07	7.6522-01	1.2983-02
1.1054+00	6.5014-07	7.6027-01	1.3296-02
1.1079+00	6.4085-07	7.1000-01	1.3387-02
1.0512+06	7.1610-07	7.5276-01	1.2983-02
9.90.00+05	8.5590-07	8.4773-01	1.1617-02
9.6/15+05	9.0597-07	8.3997-01	1.1689-02

UN-4 ...-₂3−€4 UN1110€D

LMISPHERE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH AIP UDED DEPLETED UNANIUM SPHERE BIAS = .0189 STANDARD LIGHT UNITS HENGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .00

REUTRON FLUX	FLUX*ENERGY (RELATIVE UNITS)	RELATIVE
1.0859-06	9.4443-01	1.1161-02
1.1404-06	9.3223-01	1.2721-02
1.0585-06	8.1481-01	1.3600-02
1.3233-06	9.6091-01	1.2615-02
1.0826-06	7.4280-01	1.3765-02
1.4262-06	9.2607-01	1.3187-02
1.1683-06	7.1898-01	1.4629-02
1.3313-06	7.7760-01	1.4683-02
1.2911-06	7.1226-01	1.4037-02
1.4149-06	7.3385-01	1,4931-02
	HEUTRON FLUX (RELATIVE UNITS) 1.0859-06 1.1404-06 1.0585-06 1.3233-06 1.0826-06 1.4262-06 1.1683-06 1.3513-06 1.2911-06 1.4149-06	HEUTRON FLUX FLUX*ENERGY   (RELATIVE UNITS) (RELATIVE UNITS)   1.0859-06 9.4443-01   1.1404-06 9.3223-01   1.0585-06 8.1481-01   1.3233-06 9.6091-01   1.0826-06 7.4280-01   1.4262-06 7.1898-01   1.3313-06 7.7760-01   1.2911-06 7.1226-01   1.4149-06 7.3385-01

RUN T4 1-29-06

COMPUTED 3-25-69

HEMISPHERE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH WATER COULED DEPLETED URANIUM SPHERE BIAS = .0335 STANDARD LIGHT UNITS 1.238 INCH URANIUM FILTER AT 16 METER POSITION OF 50 METER FLIGHT PATH

HEUTKON ENERGY	NLUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.53.0+07	3.7643-09	5.7906-02	1.9457-01
1.4429107	6.0731-09	8 7629-02	1.5736-01
1. 3561+117	8 3848-09	1,1371-01	1 3689-01
1. 27/ 9107	1 0626-08	1 3566-01	1 2370-01
1.201 5407	1 5431-08	1 8587-01	1 0521-01
1.13.0+07	1 7761-08	2 0212-01	1 0108-01
1.07.9107	3.2358-08	3 4845-01	7 8248-02
1.0206107	3 1 376-09	3 5090-01	7 0976-02
9 445 4446	5 9629-08	5 6324-01	4 5055-02
0.5415446	9 7046-08	B 2990-01	3 7998-02
7 77 +00	1 ( ( ( 0 7	8.2330-01	3,1150,02
7.1720+06	1.6666-07	1.2964+00	3.1150-02
7.1059+00	2.2004-07	1.0204+00	2.8414-02
6.5169+06	3.0551-07	1,9910+00	2.0100-02
5.9962+06	3.9820-07	2.3765+00	2,4350-02
5.5390+06	4.5490-07	2.5197+00	2.3969-02
5.1500+00	5.7416-07	2.9458+00	2.2504-02
4.7058+06	6.8444-07	3.2619+00	2.1670-02
4.4385+06	7.8044-07	3.4640+00	2.1346-02
4.1439+06	9.0116-07	3.7343+00	2.001/-02
3.8/70+06	1.0492-06	4.0684+00	1,9946-02
3.6362+06	1.1893-06	4.3245+00	1.9479-02
3.4167+06	1.3026-06	4.4506+00	1.9236-02
3.2164+06	1.5307-06	4.9233+00	1.8259-02
3.0333+06	1.7048-06	5.1712+00	1.7827-02
2.8054+06	1.8428-06	5.2804+00	1.7665-02
2.7110+06	2.1019-06	5,6983+00	1.7011-02
2.5050+06	2.2193-06	5,7009+00	1.7071-02
2.4375+06	2.5023-06	6,0994+00	1.6521-02
2.2875+00	2.7192-06	6.2202+00	1.3410-02
2.1240+06	3.0115-06	6.3982+00	1.3302-02
1.9725+06	3.4047-06	6.7362+00	1.3004-02
1.8469+06	3.8125-06	7.0413+00	1.2847-02
1.7281+06	4,2670-06	7.3738+00	1.2620-02
1.6204+06	4.7873-06	7.7573+00	1.2441-02
1.5224+06	5.1970-06	7.9119+00	1.2356-02
1.4331+06	5,5368-06	7,9348+00	1.2428-02
1.3514+06	6.1577-06	8,3215+00	1,2293-02
1.2755+06	6.6327-06	8,4666+00	1,2202-02
1.2070+00	7.4355-06	8,9791+00	1,2027-02
1.1442+00	7.9788-06	9,1293+00	1.2158-02
1.0057+06	8.4477-06	9,1717+00	1.2462-02
1.0230+06	9.1876-06	9,3989+00	1.0992-02
9.5765+45	1.0499-05	1.0054+01	1.0805-02

COMPLITED 3-25-69

CON 14 1-29-06 TABLE E2

HEMISPHENE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH LATER COULD JEPLETED URANIUM SPHERE 51AS = .0335 STANDARD LIGHT UNITS 1.238 14CH URANIUM FILTER AT 16 METER POSITION OF 50 METER FLIGHT PATH

NEUTION ENLROY	EUTRON FLUX	FLJX*ELERGY	RELATIVE
()	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
3.9838+05	1.1794-05	1,0595+01	1.0779-02
3.44.4405	1.3030-05	1,1003+01	1.0913-02
7.95,2+05	1,4021-05	1,1150+01	1.1220-02
7.50:3+05	1.5090-05	1,1320+01	1.1538-02
7.03.6+05	1.0026-05	1.1.360+01	1.1951-02
6.70 7+05	1.7126-05	1.1489+01	1.2354-02
0.55 5+05	1.3137-05	1.1532+01	1.2825-02
0.0050+05	1.9085-05	1.1518+01	1.3374-02
5.7002+05	2.1034-05	1.1990+01	1.2592-02
5.3004+05	2.2491-05	1.2054+01	1.3702-02
5.04_3+05	2,4479-05	1.2356+01	1.4838-02

### TABLE E3

COMPUTED 4-24-69

. . . . . . . . . . . . .

(a) A set of the se

#### KUN T5 1-29-06

HEMISPHERE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH VATER COOLED DEPLETED URANIUM SPHERE BIAS = .0335 STANDARD LIGHT UNITS IN URANIUM FILTER AT 16 LETER POSITION OF 50 METER FLIGHT PATH

NEUTRUN ENERGY	AEUTRON FLUX	FLUX*EI.ERGY	RELATIVE
(正言)	(RELATIVE JUITS)	(RELATIVE UNITS)	UNCERTAINTY
1.5415+07	1.6951-09	2,6130-02	1.5481-01
1.3165+07	3.2161-09	4.2340-02	1.1924-01
1.2045+07	6.4432-09	7.7600-02	1,1455-01
1.15 0+07	7.5310-09	8,5703-02	1.0859-01
1.0759+07	9.2234-09	9,9327-02	1,0068-01
1.0206+07	1.5680-08	1.6003-01	7.8833-02
9.4458+06	1.9042-08	1.8270-01	5,2392-02
8.5510+06	3.0750-08	2,6296-01	4.3637-02
7.77.0+05	5.0817-08	3.9529-01	3,5600-02
7.10.39+06	6.7200-08	4.7752-01	3.2133-02
6.51.9+05	9.2072-08	6.0002-01	2.8517-02
5.9462+46	1.1908-07	7.1427-01	2.6022-02
5.5090+06	1.3879-07	7.6876-01	2.5009-02
5.1000+85	1.6534-07	8.4829-01	2.3865-02
4.7058+06	1.9962-07	9.5135-01	2.2641-02
4.45.5+06	2.3134-07	1.0268+00	2.2023-02
4.1439446	2.597-07	1,1022+00	2.1258-02
3.8770+16	3.0967-07	1.2000+00	2.0555-02
J. bubetib	3.5471-07	1.2898+00	1.9973-02
5-6167400	4.2085-07	1.4379+00	1.4973-02
	4. 5984-117	1.4147+00	1.9139-02
3-10-0+46	4.9437-07	1.4996+00	1.8654-02
5 BUSHANA	5 4661-07	1.5720+00	1.8291-02
2.7110+00	6.0987-07	1.6534+00	1.7886-02
2.501.0406	6.5903-07	1-6929+00	1.7809-02
2.00.0700	7 1307-07	1.7381+00	1.7681-02
244070400	a odlom07	1 8302+00	1.4192-02
2 1.1 (54.16	9 0938-07	1 9310+00	1.3994-02
1 97:5406	9 7840-07	1.9358400	1.4119-02
1 607 GANG	1 1017-06	2.0347+00	1,4006=02
1 72 1236	1 2056-06	2.0834+00	1.4002-02
1 AMERICANE	1 2749-06	2 2270+00	1.3777-02
1 5 2 4705	1 6251-06	2 321 5400	1 3607-02
1.0332.00	1.6565-06	2 3807+00	1.3659-02
1.1001400	1.0161-06	2.0007400	1 3768-02
1 37/8400	1 0010-04	2.4343400	1 3502-02
	1.9919-06		1 3720-02
	5. 15.40-00 5. 15.63-44	5 690HADO	1.3500-02
エキエヤ特ピサリロ	2.330 <b>3-00</b> 9.8251-06	2.7415+00	1.3799-02
	5.075-75-00 2 0.510-05	2 7198+00	1.2339-02
エキリとうりキリピ	2 1145 14 5 0012-00	C. ( 1277UV	1 2001-02
9.5/65+05	3.0100-00	2.00000000	1 1900-03 1°CAAT-05
0.9038405	3,4002-00	3.0043400	1.077402
3.4444405	3.//04-00	3.1034400	1°%^10=0%

TABLE E3

COMPUTED 4-24-69

PON 15 1-29-06 CUNTINULL

ILMISPHERE INTEGRATED FAST NEUTRON SOURCE SPECTRUM FOR A 3 INCH MATER COOLED DEPLETED URANIUM SPHERE BIAS = .0335 STANDARD LIGHT UNITS NO URANIUM FILTER AT 16 METER PUSITION OF 50 METER FLIGHT PATH

HLUTION LULKUY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
()	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.95,2+05	4.1057-06	3,3126+00	1,2121-02
7.5010+05	4.5367-06	3,4033+00	1.2319-02
7.03 0+05	4.7919-06	3.3968+00	1.2718-02
6.70 7+05	5.1075-06	3.4667+00	1.3015-02
5.55 5+05	5.5286-06	3.5154+00	1.3367-02
n.U350+05	5.9237-06	3.5780+00	1.3757-02
5.70.2+03	6.5002-06	3.7052+00	1.2919-02
5.3394+05	7.1240-06	3.8180+00	1.3781-02
5.0405+05	7.7954-06	3.9354+00	1.4823-02

#### REPRODUCIBILITY OF THE ORIGINAL PAGE IS FOOR.

NUN 0 7-31-04

TABLE E4

COMPUTED 3-25-69

LEGRELS--- PEUTRUM HAS BEEN CORRECTED FOR TRANSMISSION OF DEWAR AT D LEGRELS--- PEUTRUM HAS BEEN CORRECTED FOR TRANSMISSION OF DEWAR JETECTOR 31A5 = .0276 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR

NEUTION ENERSY	LUT YON FLUX	FLUX*EHERGY	RELATIVE
(Ľ'.)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1 5.10 - 10.7	6 #851-09	9 7568-02	1 6045-01
1.3045+07	1 335-09	2 4259-01	1 1325-01
1.32.51+07	2 5001-09	3 1141-01	9 2327-02
1.2450407	2.5001-00	3 7945 01	7 0771-02
1.1/45+07	5.2214-08	5.7845-01	6 5730 02
1.1049+07	4.5053-08	5.2002-01	6.1000.02
1.0301+07	5.4950-08	5.7709-01	5.1989-02
9.9313+00	8.2039-08	8.2436-01	5.1380-02
9.20 2+05	1.2139-07	1.1178+00	3,1303-02
0.0040+05	1.9889-07	1.65//+00	2.75/0-02
7.58/0+00	3.1702-07	2.4030+00	2.1699-02
6.94-34+30	4.0205-07	2.7836+00	2.0516-02
0.3+1,5+00	5.3926-07	3.4235+00	1.8535-02
5.0425+30	6.5265-07	3.8130+00	1.7500-02
5.394+00	7.9741-07	4.3015+00	1.6508-02
4.9961+05	9.2342-07	4.6385+00	1,5996-02
4.0403+00	1. ,940-06	5,0765+00	1.5360-02
4.3212+00	1.2010-06	5.4058+00	1.4983-02
4.03.0+05	1.4770-06	5.9582+00	1.4315-02
3.7744+36	1.6985-06	6.4108+00	1.3847-02
3.5372+00	1.0328-06	6,6636+00	1.3653-02
3.3252+00	2.1140-06	7.0315+00	1.3271-02
3.1301+06	2.2997-06	7.1983+00	1.3087-02
2.9011+05	2.5478-06	7.5203+00	1.2753-02
2.73. 1+05	2.0470-06	7.9377+00	1.2626-02
2.0377+00	3,3192-06	8,7551+00	1.2457-02
2.4992+06	3.3045-06	8.4086+00	1.2789-02
2.3713+05	3.3517-06	9,1335+00	1.2174-02
2.2253+06	4.3050-06	9,5799+00	9.7654-03
2.0010+05	4.9811-06	1.0294+01	9.7418-03
1.9244+00	5.3992-06	1.0390+01	9.6664-03
1.7900+00	6.0323-06	1.0836+01	9.4059-03
1.od otuo	6.3445-06	1.0663+01	9.5186-03
1.5750+00	7.1585-06	1.1280+01	9.6113-03
1.43.5+05	8.0858-06	1.1971+01	9.6502-03
1.3935+06	8.6055-06	1,1992+01	9.5437-03
1.3140+06	9.3501-06	1.2286+01	9,6913-03
1.2411+06	1.0081-05	1.2512+01	9,9165-03
1.17.1+00	1.0963-05	1.2872+01	9,5537-03
1.1124+06	1.1975-05	1.3321+01	9,6592-03
1.0055+06	1.3209-05	1.3942+01	9,9096-03
9.9449+05	1.3275-05	1.3202+01	8.5635-03
9.3093+05	1.4406-05	1.3411+01	8.2774-03
8.73:7+05	1.5336-05	1,3567+01	8.7935-03

COMPUTED 3-25-69

CUNTINULL

HAST NEUTRON SOURCE SPECTRUM MEASURED THROUGH THE EMPTY LH2 DEWAR AT OULGKEES---SPECTRUM HAS BEEN CORRECTED FOR TRANSMISSION OF DEWAR DETECTOR BIAS = .0276 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR

NEUTRON ENERGY	(KELATIVE UNITS)	FLUX*ENERGY (RELATIVE UNITS)	RELATIVE
8.20 1+05	1.9495-05	1,5181+01	9.0575-03
7.7294+05	1.7481-05	1.3512+01	1,0316-02
7.2914+05	1.7816-05	1.2990+01	9.9119-03
0.8890+05	2.0232-05	1,3939+01	9.2634-03
6.5201+05	2.0792-05	1.3557+01	9.8141-03
6.1795+05	2.2428-05	1.3859+01	9.7495-03
5.3050+05	2.3307-05	1.6602+01	9.4103-03
5.5395+05	2.4255-05	1.3436+01	9,9231-03
5.2011+05	2.6044-05	1,3564+01	1.0486-02

COMPUTED 5-15-69

KUN TH 2-20-06

FAST NEUTRON SOURCE SPECTRUM MEASURED THROUGH THE EMPTY LH2 DEWAR AT O DEGREES---SPECTRUM HAS BEEN CORRECTED FOR TRANSMISSION OF DEWAR DETECTOR BIAS = .04233 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR

HEUTINGN LINERSY	NLUTRON FLUX	FLUX*ENERGY	RELATIVE
(E7)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.3927+07	1.0052-06	1.3999+01	1.0344-01
1.2330+117	1.7534-16	2,1630+01	7.4800-02
1.1002+07	3.0335-06	3,3375+01	5.8704-02
9.87.2+06	5.3390-06	5.7661+01	4-5244-02
8.9111+00	9.1743-06	8,1753+01	3.9143-02
3.0024+06	1.6270-05	1.3150+02	3.1889-02
7.3001+00	2.3002-05	1.6939+02	2.8971-02
6.7375+06	3.1114-05	2.0963+02	2.6865-02
6.1070+00	3.9179-05	2.4242+02	2.5681-02
5.7023+06	4.6582-05	2.6562+02	2.5033-02
5.2720+06	5.4737-05	2.8857+02	2.4306-02
4.80.7+00	6.6221-05	3.2373+02	2.3357-02
4.4073+06	7.8953-05	3.5271+02	1.8563-02
4.0279+06	9.3502-05	3.7662+02	1.8100-02
3.65 4+00	1.1100-04	4.0541+02	1. 7587-02
3.3235+06	1.2719-04	4.2272+02	1.7138-02
3.0337+06	1.5336-04	4.6602+02	1.6098-02
2.70:9+05	1.7587-04	4.9048+02	1.5863-02
2.5028+06	2.0970-04	5,3868+02	1.5577-02
2.3737+06	2.3619-04	5.6064+02	1.5094-02
2.2000+06	2.7518-04	6.0540+02	1.4697-02
2.0210+06	3.2415-04	6.5511+02	1.2412-02
1.0413+06	3.6734-04	6.7638+02	1.2051-02
1.00.0+0+00	4.1305-04	7.0425+02	1.1764-02
1.5+71+00	4.6169-04	7.1428+02	1.2066-02
1.4253+00	5,2055-04	7.5075+02	1.1979-02
1.31.2+06	5.7778-04	7.6163+02	1.2019-02
1.2220+00	6,2031-04	7,6554+02	1.2250-02
1.1267+00	7.4457-04	8.3891+02	1.0639-02
1.0023+06	8.0135-04	8.2763+02	1.1432-02
9.5010+05	8.0421-04	7.6413+02	1.1325-02
8.7707+05	9.0968-04	7.9785+02	1.1777-02
8.1211+65	1.0524-03	8.5466+02	1.2580-02
7.4070+05	1.0212-03	7.6462+02	1.3029-02
6.87.5+05	1.1498-03	7.9064+02	1.2520-02
6.3370+05	1.2668-03	8.0277+02	1.3366-02
5.8530+05	1.4/97-03	8.6693+02	1.4185-02
5.4327+05	1.3989-03	7.5998+02	1.7377-02
5.0221+05	1.6156-03	8,1137+02	1.8/19-02

COMPUTED 5-5-69

FAST ALUTICAL SPLETRUT FOR EMPTY DEWAR AT ULLEGREES USING 2-IN. ME-211 DETECTOR TAS = .2.746 STANDARD LIGHT UNITS 0.87 INCH PRECOLLINATOR CHENGT RESILUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

W. UTION LICKUY	LUT ON FLUX	FL:JX*ENERGY	RELATIVE
(亡、)	(REGATIVE UNITS)	(RELATIVE UNITS)	UCERTAINTY
1.00.0+07	4.3562-09	6.5539-02	1.6045-01
1.23.4+67	1.4214-08	1.8256-01	7.1367-02
1.14, 3+17	2. 219-08	2.3803-01	5.0799-02
1.0220+07	4.2720-08	4.3692-01	3.9583-02
9.20 2+00	7.3448-08	6.7632-01	3.1305-02
1.33.3+05	1.1750-07	9.3001-01	2.5878-02
1.53 0+03	1. 350-07	1.3914+00	2.1699-02
0.9:34+30	2.2385-07	1.5846+00	2.0316-02
0.34. 3+05	3.0250-07	1,9204+00	1.8336-02
5.84 3+00	3.0320-07	2.1223+00	1.7500-02
5.3744+05	4.1134-07	2.3335+00	1.6508-02
4.99,1+05	5.1211-07	2.5580+00	1.5996-02
4.5092+06	6.2240-07	2.3379+00	1.2494-02
4.10.0+05	7.3561-07	3,2245+00	1.1813-02
3.71.,1+00	9.5669-07	3.5612+00	1.1340-02
3.0179+00	1.1570-06	3.9+20+00	1.0859-02
3.0349+00	1.3+38-06	4.1455+00	1.0650-02
2.02 4+05	1.0050-06	4.5396+00	1.0291-02
2. WJebtus	1. 195-06	4.7354+00	1.0311-02
2.40. 3+00	2.0350-06	5.0115+00	1.0051-02
2.22.0+06	2.4030-06	5,4600+00	9.7654-03
2.04, ++00	2.1290-06	5.7792+00	8.4166-03
1.0391+06	3.3465-06	6.2215+00	8,2299-03
1.0345+00	3.7963-06	6.4527+00	8.2486-03
1.5390+06	4.2117-06	6.5635+00	8.4346-03
1.43.3+06	4.9916-06	7.1694+00	8.2893-03
1.3.270+00	5.0433-06	7.4887+00	8.2822-03
1,2290+00	6.0202-06	7.4036+00	8.5808-03
1.1020+00	7.3322-06	8.3059+00	7.4002-03
1.0378+00	7.9832-06	8.2850+00	7.6213-03
7.5413+35	8.8463-36	8.4410+00	7.5459-03
3.8J31+05	9.4998-06	8.3628+00	7.7694-03
0.14/0+05	1.0074-05	8.2073+00	9.2329-03
1.5010+05	8.6284-06	6.5245+00	9.3321-03
5.90 0+05	1.1760-05	8.2186+00	7.6293-05
5.+3.3+05	1.2730-05	8,1902+00	7.9555-05
0.9+0+05	1.3519-05	9.1935+00	7.8029-03
5.50.00+0.5	1.3//9-05	7.5862+00	9.0750-05
3. 0.3 . 0 + 0 3	1.11/2-11	1.2070+00	4.104.3=0.3

COMPUTED 5-5-69

NUN 5 7-31-04

FAST NEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 0 DEGREES USING A NE-211 LETECTOR STAS = .U2746 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

14	EUTION ENERUY	NEUTPON FLUX	FLUX*ENERGY	RELATIVE
	(19)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	1.5045+07	4.5957-09	6,9142-02	1.4347-01
	1.2044+07	1.5898-08	2.0419-01	5,9501-02
	1.1425+07	2.2206-08	2.5366-01	4.8403-02
	1.0220+07	3.7598-08	3.8448-01	3,7121-02
	9.20/2+06	6.2209-08	5.7283-01	2,9914-02
	8.3340+06	1.0099-07	8.4173-01	2.4502-02
	7.3000+06	1.5259-07	1.1566+00	2.0920-02
	6.92 4+06	1.9471-07	1.3481+00	1.9350-02
	6.3485+06	2.2895-07	1.4535+00	1.9562-02
	5.8423+06	2.6972-07	1,5758+00	1.7887-02
	5.3944+06	3.1813-07	1.7161+00	1.7144-02
	4.9961+06	3.6047-07	1.8309+00	1.6679-02
	4.5092+06	4.3123-07	1,9661+00	1.3257-02
	4.1045+06	5.2105-07	2.1386+00	1.2813-02
	3.7147+06	6.0687-07	2.2543+00	1.2607-02
	3.3779+00	7.4316-07	2.5103+00	1.2046-02
	3.0049+05	8.3057-07	2,5807+00	1.1965-02
	2.8224+06	9.6401-07	2.7266+00	1.1778-02
	2.0020+06	1.0454-06	2.7208+00	1.2078-02
	2.4029+06	1.2098-06	2.9070+00	1.1722-02
	2.2253+00	1.3566-06	3,0188+00	1.1671-02
	2.0424+06	1,4952-06	3,0538+00	1.0290-02
	1.5591+06	1.0765-06	3.1168+00	1.0342-02
	1.0995+00	1.8345-06	3,1177+00	1.0553-02
	1.5590+06	1,9329-06	3.0146+00	1,1068-02
	1.4303+00	2.1821-06	3.1342+00	1.1139-02
	1.3270+06	2.2804-06	3.0261+00	1.1564-02
	1.2298+00	2.2471-06	2.7635+00	1.2447-02
	1.1328+00	2.5693-06	2.9105+00	1.1058-02
	1.0378+06	2.6233-06	2.7225+00	1.1726-02
	9.5418+05	2.8168-06	2.6877+00	1.1767-02
	8.6031+05	2,9505-06	2.5974+00	1.2229-02
	8.1+70+05	2.9517-06	2.4047+00	1.3312-02
	7.5016+05	2.3675-06	1.7902+00	1.5552-02
	6.93A0+U5	3.2155-06	2.2472+00	1.2692-02
	6.4338+05	3.2994-06	2.1228+00	1.3549-02
	5.9420+05	3.8523-06	2.2893+00	1.3421-02
	5.5050+05	3.1062-06	1.7101+00	1.6432-02
	5.0350+05	3.1473-06	1.6004+00	1.6698-02
	4.6821+05	3.0925-06	1.4482+00	1.9547-02
	4.3271+05	3.1469-06	1.3617+00	2.2755-02
	4.11102405	2.5000=05	1.0266+00	2.89/2-02

101 4 7-31-64

COMPUTED 5-5-69

FAST NEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 0 DEGREES USING A NE-211 ULTECTOR TAS = .02746 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERFOR = .20

÷.	EUTRUN ENERGY	NEUIRON FLUX	FLUX*ENERGY	RELATIVE
	(2)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	1.50+5+07	5.0759-09	7.6367-02	1.3828-01
	1.2044+47	1.2700-08	1.6312-01	7.6884-02
	1.14/0+07	2.0139-08	2.3005-01	5.7243-02
	1.0225+07	3.1293-08	3,2000-01	4.6296-02
	9.00 .+06	5.0098-08	4.6131-01	3.7855-02
	3.3340+03	7.2002-08	6.0429-01	3.2869-02
	7.58 0+05	1.1297-07	8.5631-01	2.7575-02
	0.923++00	1.4002-07	9,6941-01	2.5873-02
	0.34.5+00	1.7713-07	1.1245+00	2.3901-02
	3.04. 3+06	1.9956-07	1.1659+00	2.3552-02
	5.3344+05	2.3927-07	1.2907+00	2.2383-02
	4.9951+06	2.4052-07	1.3016+00	2.2395-02
	4.5592+06	2.0099-07	1.3084+00	1.8363-02
	4.10.0+06	3.4980-07	1.4358+00	1.7675-02
	3.7147+06	3,9310-07	1.4602+00	1.7691-02
	3.3174+00	4.7038-07	1.6109+00	1.6961-02
	3.0349+06	5.1598-07	1.5917+00	1.7157-02
	2.8204+06	5.9409+07	1.6803+00	1.6871-02
	2.6020+00	6.3955-07	1.6645+00	1.7342-02
	2.40,9+06	7.0951-07	1.7049+00	1.7170-02
	2.2253+06	8.0266-07	1.7862+00	1.6995-02
	2.0+24+00	8.5410-07	1.7445+00	1.5213-02
	1.3391+06	9.2058-07	1.7228+00	1.5507-02
	1.0995+06	1.0234-06	1.7495+00	1,5660-02
	1.5590+06	1.0360-06	1.6157+00	1.6775-02
	1.4363+05	1.0316-06	1.5535+00	1,7523-02
	1.5270+05	1.1138-05	1.4780+00	1.8279-02
	1.2290+06	1.0899-06	1.3404+00	1.9707-02
	1.1328+05	1.1473-06	1.2997+00	1.8208-02
	1.0.376+06	1.1136-06	1,1557+00	1.9759-02
	9.5410+05	1.0338-06	1.0341+00	2.0768-02
	8.6031+05	1.0170-06	8,9528-01	2.2754-02
	3.1470+05	1.1002-06	8.9633-01	2.3719-02
	7.5010+35	8.2925-07	6.2705-01	2.8645-02
	6.90.0+05	1.0186-06	7.1186-01	2.4513-02
	6.4330+05	9.6515-07	6,2096-01	2.7193-02
	5.9420+05	1.0061-06	6.3354-01	2.765/-02
	5.5000+05	8.5107-07	4.7407-01	3.3850-02
	5.0350+05	8,0006-07	4.1090-01	3.5836-02
	4.6031+05	7.9098-07	3.7042-01	4.1801-02
	4.3271+05	6.7381-07	2.9156-01	5.4570-02
	4.0122+15	5. 3400-07	2.3447-01	6.8430-02

KUN 8 -1-6+

COMPUTED 5-13-69

FAST NEUTRON SPECTRUM FOR EMPTY DEWAR AT 15 DEGREES USING 5-IN. NE-211 DETECTOR DIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERUY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(=,)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.1151+07	1,2578-10	1.4026-03	1.9638-01
9.20 32+06	3.5202-10	3,2415-03	1.3140-01
8.3348+06	4.7760-10	3,9807-03	1.1220-01
7.5300+06	7.3135-10	5,9226-03	8.9178-02
6.9234+06	1.0720-09	7.4219-03	7.9343-02
6.3485+06	1.4673-09	9.3152-03	7.0627-02
5.8423+06	1.7747-09	1,0368-02	6.7366-02
5.3944+06	2.1839-09	1.1781-02	6.3272-02
4.9961+06	2.1665-09	1.0824-02	6.6805-02
4.5592+06	2.8048-09	1.2788-02	5.0724-02
4.1045+06	3.6649-09	1.5043-02	4.7207-02
3.7147+06	4.7814-09	1.7761-02	4.3992-02
3.3779+00	5.6693-09	1,9150-02	4.2729-02
3.0849+06	7.0295-09	2.1685-02	4.0334-02
2.5264+00	8.7144-09	2.4648-02	3.8301-02
2.6020+06	1.0513-08	2.7361-02	3.7459-02
2.4029+06	1.2576-08	3.0219-02	3.5745-02
2.2253+06	1,4179-08	3.1553-02	3.5745-02
2.0424+06	1.7243-08	3.5217-02	3.0150-02
1.8591+06	1.9994-08	3.7171-02	2.9761-02
1.0995+00	2.4895-08	4.2309-02	2,8513-02
1.5590+06	2.7744-08	4.3270-02	2.9086-02
1.4363+00	3.1453-08	4.5176-02	2,9297-02
1.3270+06	3.7409-08	4.9642-02	2.8606-02
1.2290+06	4.4743-08	5.5025-02	2.8094-02
1.1328+06	4.0780-08	5.2992-02	2.6186-02
1.0378+06	5.4029-08	5,6071-02	2,5087-02
9.5418+05	6.0087-08	5.7334-02	2.5886-02
8.8031+05	7.2784-08	6.4072-02	2.5195-02
8.1470+05	8.1355-08	6.6280-02	2.6079-02
7.5010+05	8.8191-08	6.6687-02	2.5195-02
6.9820+05	8.8637-08	6.1945-02	2.5002-02
6.4330+05	8.2177-08	5.2871-02	2.8107-02
5.9426+05	8.8334-08	5.2493-02	2,9199-02
5.5050+05	8.7609-08	4.8234-02	3,2951-02
5-0450+05	8.3047-08	4-4772-02	3.4492-02

COMPUTED 5-13-69

KUN 7 -1-64

TAST NEUTRON SPECIRUM FOR 2.5 INCHES LH2 AT 15 DEGREES USING A NE-211 UEILCTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

141	UT WHI LILLEY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(E.)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	1	1 0012-10	1 2407-03	1 031/1-01
	1.24/0+07	1.0012-10	1.2493-03	1.8514-01
	1.0000+07	2.2555-10	2.4336-03	1.1328-01
	9.6972+06	2.2761-10	2.2072-03	1,1583-01
	3.7052+06	3.3187-10	2,9056-03	9,6688-02
	7.94.0+06	4.8758-10	3.8733-03	8.2728-02
	7.24.00+06	6.0355-10	4.3701-03	7.7732-02
	0.0200+06	7.5302-10	4,9900-03	7,2934-02
	0.0075+06	1.1350-09	6,9093-03	6.1904-02
	5.0117+06	1.2274-09	6.8878-03	6.2370-02
	5.1395+06	1.6129-09	8.3701-03	5,6890-02
	4.8133+06	1,9870-09	9.5640-03	5,3383-02
	4.3990+06	2.5062-09	1.1290-02	4.0653-02
	3.90 (0+06	3.1505-09	1.2541-02	3,8983-02
	3.5970+06	3.0855-09	1.3976-02	3.7399-02
	3.2757+06	4.7175-09	1.5453-02	3.5694-02
	2.9950+06	6.1754-09	1.8499-02	3.2830-02
	2.7510+06	6.7054-09	1.8440-02	3,3430-02
	2.5334+35	8.0202-09	2.0318-02	3.2655-02
	2.34:4+06	9.0018-09	2.1779-02	3.1642-02
	2.1704+06	1.0704-08	2.3232-02	3.1595-02
	1.99.1+06	1.2219-08	2.4366-02	2.7259-02
	1.0172+00	1.4367-08	2.6108-02	2.6776-02
	1.6028+00	1.7064-08	2.9374-02	2.6246-02
	1.5273+06	1.5403-08	2.8107-02	2.7209-02
	1.4077+06	2.1056-08	2.9641-02	2.7219-02
	1.3010+06	2.3024-08	2.9968-02	2.7866-02
	1.2171+00	2.6033-08	3.1424-02	2.8019-02
	1.11.8+06	2.9783-08	3.3143-02	2.4870-02
	1.0202+06	3.2001-08	3.3260-02	2.5418-02
	9.345 4+45	3.4575-08	3.2455-02	2.5779-02
	3.00.7+05	3 9500-08	3.4230-02	2.5995-02
	1.0001+00	4 4962-03	3.6080-02	2.6585-02
	0-0240+05	4,4702-00	0.0000-02	2.0303-02

JN 6 8-1-64

COMPUTED 5-13-69

AST NEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 15 DEGREES USING A NE-211 TECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR NERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.1423+07	1.2048-10	1.3762-03	1.9678-01
1.0220+17	1.7264-10	1.7654-03	1.3976-01
9.2052+06	2.4940-10	2.2965-03	1.0760-01
8.3348+06	3.6375-10	3.0318-03	8.7870-02
7.5dnu+06	5.6085-10	4.2512-03	7.2044-02
6.9234+06	8.2538-10	5.7144-03	6.1325-02
6.3485+06	1.0641-09	6.7554-03	5.6426-02
5.8423+06	1.1184-09	6.5340-03	5.7899-02
5.3944+06	1.3123-09	7.0791-03	5.5833-02
4.9961+06	1.6643-09	8.3150-03	5.1783-02
4.5592+06	2.1175-09	9.6541-03	3.9629-02
4.1040+06	2.5302-09	1.0385-02	3,8639-02
3.7147+06	3.1210-09	1.1594-02	3.7043-02
3.3779+06	4.0295-09	1.3611-02	3.4388-02
3.0349+06	4.5283-09	1.3969-02	3.4108-02
2.8254+06	5.6506-09	1.5982-02	3.2266-02
2.6026+06	6.5482-09	1.7042-02	3.2149-02
2.4029+06	7.7278-09	1.8569-02	3.0913-02
2.2253+06	7.9948-09	1.7791-02	3.2317-02
2.0424+06	9.8821-09	2.0183-02	2.6988-02
1.8591+06	1.0988-08	2.0428-02	2.7267-02
1.0995+06	1.2368-08	2.1019-02	2.7441-02
1.5590+06	1.3626-08	2.1251-02	2.8160-02
1.4363+06	1.5680-08	2.2521-02	2.8104-02
1.3270+06	1.5970-08	2.1192-02	2.9772-02
1.2298+06	1.9280-08	2.3711-02	2,9035-02
1.1328+06	1.9759-08	2.2383-02	2.7330-02
1.0378+00	2.1619-08	2.2436-02	2.8007-02
9.5418+05	2.2870-08	2.1822-02	2.8480-02
3.8031+05	2.6363-08	2.3208-02	2.8399-02
8.1470+05	2.8162-08	2.2944-02	3,0108-02

RUN 5 8-1-64

COMPUTED 5-13-69

FAST NEUTRON SPECTRUM FOR 7.0 INCHES LH2 AT 15 DEGREES USING A NE-211 DITECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.2844+07	9.5154-11	1,2222-03	1.5405-01
1.1423+07	1.4858-10	1.6972-03	1.0145-01
1.0226+07	1.9178-10	1.9611-03	8.5949-02
9.2082+06	2.2827-10	2.1020-03	8,1290-02
8.3340+06	2.6733-10	2.2281-03	7.8287-02
7.5800+06	4.6162-10	3.4991-03	6.1069-02
6.9234+06	5.6931-10	3,9416-03	5.7558-02
0.3485+06	7.8342-10	4.9735-03	5,1120-02
5.8423+06	9.5675-10	5.5896-03	4.8530-02
5.3944+06	1.0654-09	5.7472-03	4.8079-02
4.9961+06	1,2752-09	6.3710-03	4.6006-02
4.5592+06	1.5582-09	7.1041-03	3,5953-02
4.1045+00	1.8685-09	7,6693-03	3.4890-02
3.7147+06	2.1300-09	7,9123-03	3.4932-02
3.3779+06	2.7500-09	9.2892-03	3.2374-02
3.0849+06	3.1189-09	9,6215-03	3,1926-02
2.82:4+06	3.5187-09	9,9523-03	3.1813-02
2.6020+06	3.8322-09	9,9737-03	3.2753-02
2.4029+06	4.4740-09	1.0751-02	3,1622-02
2.2253+00	4.9497-09	1,1015-02	3.1959-02
2.0424+06	5.6523-09	1.1544-02	2.7778-02
1.8591+06	6.1191-09	1.1376-02	2.8465-02
1.6995+06	6.3470-09	1.0787-02	2.9915-02
1.5596+06	6.8601-09	1.0699-02	3,1058-02
1.4363+06	7.6964-09	1.1054-02	3.1391-02
1.3270+06	7.5635-09	1.0037-02	3.3876-02
1.2298+06	8,7883-09	1.0808-02	3.3723-02
1.1320+06	8.8597-09	1.0036-02	3,2107-02
1.0378+06	9.5270-09	9.8871-03	3.3233-02
9.5418+05	9.8000-09	9.3510-03	3.4323-02
8.8031+05	9.3556-09	8.2358-03	3.8009-02
8.1470+05	1.0889-08	8.8713-03	3,8452-02

RUN 4 0-1-64

COMPUTED 5-13-69

FAST NEUTRON SPECTRUM FOR 10.5 INCHES LH2 AT 15 DEGREES USING A NE-211 DETLCTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	F . 057		
1.3256+07	5.1957-11	6.8874-04	1.7482-01
1.1423+07	1.2067-10	1.3784-03	9.7811-02
1.0226+07	1.4920-10	1.5257-03	8.5994-02
9.2082+06	2,1072-10	1.9404-03	7.2657-02
8.3340+06	2.8718-10	2.3936-03	6.4003-02
7.5800+06	3,9380-10	2.9850-03	5.7017-02
6.9234+06	4.8751-10	3.3752-03	5.3522-02
6.3435+06	6.0979-10	3.8713-03	5.0094-02
5.8423+06	8,0097-10	4.6795-03	4.5626-02
5.3944+06	9,2480-10	4.9887-03	4.4352-02
4.9961+06	1.1399-09	5.6951-03	4.1806-02
4.5592+05	1,3307-09	6.0669-03	3.3375-02
4.1045+06	1.5714-09	6.4498-03	3.2744-02
3.7147+06	1.7391-09	6.4602-03	3.3282-02
3.3779+06	2.2656-09	7.6530-03	3,0681-02
3.0849+06	2.4734-09	7.6302-03	3.0900-02
2.8284+06	2.6981-09	7.6313-03	3.1351-02
2.6020+06	2.9323-09	7.6316-03	3,2315-02
2.4029+06	3.2871-09	7.8986-03	3.1870-02
2.2253+06	3.6962-09	8.2252-03	3,1935-02
2.0424+06	4.0343-09	8.2397-03	2.8420-02
1.8591+06	4.2355-09	7.8742-03	2.9599-02
1.6995+06	4.6349-09	7.8770-03	3.0281-02
1.5596+06	5.3454-09	8.3367-03	3.0308-02
1.4363+06	5,1975-09	7.4652-03	3.3123-02
1.3270+06	5.0944-09	6.7603-03	3.5844-02
1.2296+06	5.6674-09	6,9698-03	3.6550-02
1.1320+06	5.5573-09	6.2953-03	3.5299-02
1.0370+06	6.0983-09	6.3288-03	3.6179-02
9.5418+05	6.0845-09	5.8057-03	3.8073-02
8.8031+05	5.9347-09	5.2244-03	4.1574-02
8.1470+05	6.5247-09	5.3157-03	4.3701-02

TABLE E14

COMPUTED 5-13-69

RUN 3 5-1-64

FAST NEUTRON SPECTRUM FOR 13.0 INCHES LH2 AT 15 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERSY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9,2082+06	1,9585-10	1.8034-03	9.0085-02
8.3348+06	2.5027-10	2.0860-03	8.2665-02
7.5300+06	4.3113-10	3.2680-03	6.4463-02
6.9234+06	4.6454-10	3.2152-03	6.5477-02
6.3485+06	6.8656-10	4.3586-03	5.5869-02
5.8423+06	7.8478-10	4.5849-03	5.4855-02
5.3944+06	9.9545-10	5.3699-03	5.0579-02
4.9961+06	1.1073-09	5.5322-03	5.0323-02
4.5592+06	1.2414-09	5.6598-03	4.1135-02
4.1045+06	1.5329-09	6.2918-03	3,9416-02
3.7147+06	1.7066-09	6.3395-03	3,9818-02
3.3779+06	2.2641-09	7.6479-03	3.6367-02
3.0349+06	2.2212-09	6.8522-03	3.8792-02
2.8294+06	2.5765-09	7,2874-03	3.8003-02
2.6026+06	2.7939-09	7.2714-03	3.9355-02
2.4029+06	3.1938-09	7.6744-03	3.8307-02
2.2255+06	2.9779-09	6,6267-03	4.2553-02
2.0424+06	3.4182-09	6,9813-03	3.6781-02
1.8591+06	3.7583-09	6.9871-03	3.7388-02
1.6995+06	3.9759-09	6.7570-03	3.8989-02
1.5596+06	4.0609-09	6,3334-03	4.1786-02
1.4363+06	4.3318-09	6.2218-03	4.3550-02
1.3270+06	4.5939-09	6.0961-03	4.5141-02
1.2298+06	4.9417-09	6.0773-03	4.6916-02
1.1323+06	4.5902-09	5.1998-03	4.7079-02
1.0378+06	4.6897-09	4.8670-03	5.0153-02
9.5418+05	4.7537-09	4.5359-03	5.2513-02
8.8031+05	4.7131-09	4.1490-03	5.7191-02
8.1470+05	4.5733-09	3.7259-03	6.4941-02

RUN 8 -3-64

COMPUTED 5-5-69

FAST NEUTRON SPLCTRUM FOR EMPTY DEWAR AT 37 DEGREES USING 5-IN. NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENLROY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.2082+06	6.7007-11	6,1701-04	1.4411-01
8.3348+06	9.8832-11	8.2374-04	1.2175-01
7.5000+06	1.4484-10	1.0979-03	1.0465-01
6.9234+06	1.6581-10	1.1480-03	1.0296-01
6.3425+06	1.9936-10	1.2656-03	9.8412-02
5.8423+06	2.7090-10	1.5827-03	8.8030-02
5.3944+06	3.3546-10	1.8096-03	8.2218-02
4.9961+06	3.8307-10	1.9139-03	8.0870-02
4.5092+06	4.6127-10	2.1030-03	6.3418-02
4.1045+06	5.5588-10	2.2816-03	6.1711-02
3.7147+06	7.2791-10	2.7040-03	5.7313-02
3.3779+06	9.5852-10	3.2378-03	5.2466-02
3.0849+06	1.1331-09	3.4955-03	5.0752-02
2.8284+06	1.3035-09	3.6868-03	4.9987-02
2.6026+06	1.7509-09	4.5569-03	4.6171-02
2.4029+06	2.1957-09	5.2760-03	4.2993-02
2.2253+06	2.4269-09	5.4006-03	4.3437-02
2.0424+06	2.9650-09	6.0557-03	3,6528-02
1.8591+06	3.3426-09	6.2142-03	3.6577-02
1.6995+06	4.0077-09	6.8111-03	3,5682-02
1.5590+06	4.8831-09	7.6157-03	3.4765-02
1.4363+06	5.7444-09	8.2507-03	3,4332-02
1.3270+06	6.2679-09	8.3175-03	3,5041-02
1.2290+06	7.6311-09	9.3847-03	3.4033-02
1.1328+06	8.2679-09	9.3659-03	3.1177-02
1.0378+06	1.0066-08	1.0446-02	3.0169-02
9.5418+05	9.1920-09	8.7708-03	3.3061-02
8.8031+05	1.1099-08	9,7706-03	3.2177-02
8.1470+05	1.2944-08	1.0545-02	3.2654-02

KUN 7 8-3-64

COMPUTED 5 5- 9

FAST NEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 37 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(=;)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
7.5860+06	6,2476-11	4.7357-04	1,4306-01
6.9234+06	7.5667-11	5.2387-04	1.3284-01
6.3465+06	1.0424-10	6.6177-04	1.1520-01
5.8423+06	1.5410-10	9.0030-04	9.6195-02
5.3944+06	1.8868-10	1.0178-03	8.9728-02
4.9961+06	2.1780-10	1,0882-03	8.7579-02
4.5592+06	3.2167-10	1.4666-03	6.1335-02
4.1045+06	4.2158-10	1.7304-03	5,6670-02
3.7147+00	5.3677-10	2.1797-03	5.0592-02
3.3779+06	7.2505-10	2.4491-03	4.7979-02
3.0849+06	9.1976-10	2.8374-03	4.4543-02
2.82 34+06	1.1089-09	3.1364-03	4.2881-02
2.6026+06	1,5008-09	3,9060-03	3,9301-02
2.4029+06	1.7048-09	4.0965-03	3.8509-02
2.2253+06	2.1539-09	4.7931-03	3,6295-02
2.0424+06	2.5469-09	5.2018-03	3.0993-02
1.8591+06	2.9200-09	5.4286-03	3.0773-02
1.6995+06	3.4518-09	5.8663-03	3.0221-02
1.5590+06	4.2297-09	6.5966-03	2.9326-02
1.4300+06	4.7312-09	6,7954-03	2.9711-02
1.3270+00	5.3350-09	7.0795-03	2.9829-02
1.2293+06	6.2325-09	7.6647-03	2.9632-02
1.1028+06	6.4923-09	7.3545-03	2.7652-02
1.0378+06	7.5225-09	7.8069-03	2.7474-02
9.5418+05	8.3293-09	7.9477-03	2.7270-02
8.8031+05	9.5896-09	8.4418-03	2.7199-02
8.1470+05	1.0370-08	8.4484-03	2.8700-02

TABLE E17

COMPUTED 5-5-60

RUN 6 -3-64

FAST NEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 37 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTION ENERGY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.0025+07	2,2234-11	2.4068-04	1,9393-01
9.20-2+06	3.6321-11	3.3445-04	1.7303-01
4.3348+06	4.3785-11	3.6494-04	1.5520-01
7.5800+06	7.2088-11	5.4643-04	1.1492-01
6.9234+06	7.8349-11	5.4244-04	1.1433-01
6.3485+00	8.3160-11	5,2794-04	1.1697-01
5.8423+06	1.5433-10	9.0164-04	8.4420-02
5.3944+06	1.8990-10	1.024-03	7.8616-02
4.9961+06	2.2692-10	1.1337-03	7.4946-02
4.5592+00	2.7928-10	1.2733-03	5.8048-02
4.1045+06	3.7499-10	1.5391-03	5,2918-02
3.7147+06	5.5499-10	2.0616-03	4.5698-02
3.3779+06	6.4336-10	2.1732-03	4.4822-02
3.0849+06	8,3951-10	2.5898-03	4.0994-02
2.82.4+06	9.7284-10	2.7516-03	4.0195-02
2.6020+06	1.3689-09	3.5627-03	3.6103-02
2.4029+06	1.5219-09	3.6570-03	3.5733-02
2.2253+06	1.8495-09	4.1157-03	3.4382-02
2.0424+06	2.1393-09	4.3693-03	2.9681-02
1.8591+06	2.5013-09	4.6502-03	2,9198-02
1.6995+06	2.8905-09	4.9124-03	2.9001-02
1.5390+06	3.5952-09	5.6071-03	2.7943-02
1.4363+06	3.8049-09	5.4650-03	2.9173-02
1.3270+06	4.3613-09	5.7874-03	2,9026-02
1.2290+06	4.5597-09	5.6075-03	3.0580-02
1.1328+06	5.1208-09	5.8008-03	2.7395-02
1.0370+06	5.6862-09	5.9011-03	2.7825-02
9.5418+05	6.1588-09	5.8766-03	2.7955-02
3.8031+05	6.5809-09	5.7932-03	2.8988-02
3.1470+05	7.4036-09	6.0317-03	3,0023-02

TABLE E18

COMPUTED 5-5-69

RUN 5 8-3-6+

FAST NEUTRON SPECTRUM FOR 7.0 INCHES LH2 AT 37 DEGREES USING A HE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.2082+06	3,8209-11	3,5184-04	1,5875-01
8.3348+06	4.7705-11	3.9761-04	1.4157-01
7.5000+06	5,7683-11	4.3724-04	1.3205-01
6.9234+06	6.1950-11	4.2890-04	1.3415-01
6.345+06	9.2215-11	5.8543-04	1.1081-01
5.8423+06	8.7227-11	5.0961-04	1.2226-01
5.3944+06	1.2340-10	6.6567-04	1.0399-01
4.9961+06	1.3819-10	6.9041-04	1.0289-01
4.5592+06	2.0121-10	9.1736-04	7.2262-02
4.1045+06	3.1498-10	1.2928-03	5.9936-02
3.7147+06	4.0990-10	1.5227-03	5,5673-02
3.3779+06	4.8944-10	1.6533-03	5,3585-02
3.0849+06	6,1314-10	1.8915-03	4,9959-02
2.8234+06	1.7556-10	2.1936-03	4.6721-02
2.6026+06	9.1258-10	2.3751-03	4.6077-02
2.4029+06	1.0978-09	2.6379-03	4.3826-02
2.2253+06	1.1456-09	2.5493-03	4.5694-02
2.0424+06	1.4670-09	2.9962-03	3.7347-02
1.8591+06	1,7055-09	3.1707-03	3.6840-02
1.6995+06	1.9323-09	3.2839-03	3.6965-02
1.5596+06	2.1150-09	3.2995-03	3.8148-02
1.4363+06	2.3886-09	3.4307-03	3.8427-02
1.3270+06	2.7106-09	3.5970-03	3.8494-02
1.2298+06	2.8723-09	3.5324-03	4.0236-02
1.1328+06	3.1444-09	3.5620-03	3.6630-02
1.0378+06	3.1771-09	3.2972-03	3,9185-02
9.5418+05	3.1573-09	3.0126-03	4.1307-02
8.8031+05	3.9269-09	3.4569-03	3,9456-02
8.1470+05	4.1107-09	3.3490-03	4.2580-02

NUN 4 -3-64

COMPUTED 5-5-69

FAST HEUTRON SPECTRUM FOR 10.5 INCHES LH2 AT 37 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTHON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
7.5000+00	4.4233-11	3.3529-04	1.4820-01
6.9234+06	4.5085-11	3.1214-04	1.5722-01
6.3485+06	5.3230-11	3.3793-04	1.5062-01
5.8423+46	6.4250-11	4.9225-04	1.1778-01
5.3944+46	1,1542-10	6.2262-04	1.0164-01
4.9961+46	1.3237-10	6.6133-04	9.9277-02
4.5592+46	1.5389-10	7.0162-04	7.8870-02
4.10.5+06	2.0476-10	8.4044-04	7.2029-02
3.7147+46	2.8649-10	1.0642-03	6.3483-02
3.3779+66	3.5363-10	1.1945-03	5.9869-02
3.0049+46	4.0226-10	1.2409-03	5.8856-02
2.82:4+46	4.4997-10	1.2727-03	5.8856-02
2.6020+00	6.1706-10	1.6060-03	5.3306-02
2.4029+06	6.1643-10	1.4812-03	5.5923-02
2.2253+06	7.1449-10	1.5900-03	5.5091-02
2.0424+06	8.4178-10	1.7193-03	4.6979-02
1.8591+06	9.3165-10	1.7320-03	4.7756-02
1.6495+06	9,7697-10	1.6638-03	5.0086-02
1.5596+06	1.2105-09	1.8879-03	4.8186-02
1.4363+06	1.2587-09	1.8079-03	5.0961-02
1.3270+06	1.3379-09	1.7754-03	5.3068-02
1.2290+06	1.5191-09	1.8682-03	5.3729-02
1.1328+06	1.6184-09	1.8333-03	4.9690-02
1.0378+06	1.4611-09	1.5163-03	5.7343-02
9.5418+05	1.6917-09	1.6142-03	5, 5318-02
		· ·	and the second

COMPUTED 5-5-69

RUN 3 3-3-64

FAST NEUTRON SPECTRUM FOR 13.0 INCHES LH2 AT 37 DEGREES USING A NE-211 ULTECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
7.9440+06	3.8649-11	3,0703-04	1.5271-01
7.2406+06	5.6688-11	4.1046-04	1.2452-01
0.6260+06	6.2461-11	4.1390-04	1.2446-01
6.0875+06	7.2629-11	4.4213-04	1.2017-01
5.61:7+06	1.0195-10	5.7211-04	1.0163-01
5.1095+06	1.1970-10	6.2118-04	9.7994-02
4.8133+06	1.5607-10	7.5121-04	8.7115-02
4.3996+06	1.7673-10	7.7754-04	7.0943-02
3.9080+06	2.5048-10	9,9390-04	6.1814-02
3.5970+06	2.7665-10	9.9511-04	6.2847-02
3.2757+06	3.9239-10	1.2854-03	5.4137-02
2.9956+06	4.8924-10	1.4656-03	5.0601-02
2.7500+06	5.8695-10	1.6141-03	4.8755-02
2.5334+06	7.1838-10	1.8199-03	4.6890-02
2.3+14+06	6.9761-10	1.6334-03	5.0166-02
2.1704+06	9.2801-10	2.0142-03	4.5867-02
1.9941+06	1.0035-09	2.0011-03	4.0896-02
1.8172+06	1.0734-09	1.9506-03	4,2224-02
1.6626+06	1.2476-09	2.0745-03	4.1755-02
1.5273+06	1.2511-09	1.9108-03	4.5443-02
1.4077+06	1.5382-09	2.1653-03	4.3566-02
1.3010+06	1.5406-09	2.0052-03	4.7117-02
1.2071+06	1.6906-09	2.0407-03	4.8265-02
1.1120+06	1.6687-09	1.8569-03	4.6782-02
1.0202+06	1.9911-09	2.0313-03	4.5562-02
9.3068+05	1.9364-09	1.8177-03	4.8910-02
8.6057+05	1.9588-09	1.6974-03	5.3072-02
8.0240+05	2.0683-09	1.6597-03	5.7354-02

COMPUTED 5-13-69

RUN 9 8-5-64

FAST NEUTRON SPECTRUM FOR EMPTY DEWAR AT 53 DEGREES USING 5-IN. NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
8.4092+06	6.6338-11	5.5785-04	1.8414-01
6.6260+06	1.6946-10	1.1229-03	1.7759-01
6.0075+06	2.1950-10	1.3362-03	1.5617-01
5.6117+06	2.1082-10	1.1831-03	1.6697-01
5.1095+00	2.8170-10	1.4619-03	1.4704-01
4.8133+06	3.5508-10	1.7091-03	1.3455-01
4.3990+06	4.5627-10	2.0074-03	1.0156-01
3.9680+06	5.6459-10	2.2403-03	9.6772-02
3.5970+06	6.8649-10	2,4693-03	9.3170-02
3.2757+06	9.3969-10	3.0781-03	8.2970-02
2.9950+06	1.1878-09	3.5582-03	7.7313-02
2.7500+06	1.5238-09	4.1904-03	7.2303-02
2.5334+06	1.7144-09	4.3433-03	7.2868-02
2.3414+06	2.0778-09	4.8650-03	6.8959-02
2.1704+06	2.9753-09	6,4576-03	6.1293-02
1.9941+06	2.9939-09	5.9701-03	5.6473-02
1.8172+06	3.7375-09	6.7918-03	5.3955-02
1.6028+06	4.2687-09	7,0980-03	5.3799-02
1.5273+06	6.1302-09	9.3627-03	4.8036-02
1.4077+06	6.3750-09	8.9749-03	5.0416-02
1.3010+06	7.1811-09	9.3469-03	5.0936-02
1.2071+06	8.7508-09	1.0563-02	4,9243-02
1.1120+00	1.0186-08	1.1335-02	4.3210-02
1.0202+06	1.0311-08	1.0519-02	4.6044-02
9.3868+05	1.1379-08	1.0681-02	4.5707-02
8.6657+05	1.2148-08	1.0527-02	4.7688-02
8.0246+05	1.6073-08	1.2898-02	4,5330-02

COMPUTED 5-13-69

RUN 8 8-5-64

FAST NEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 53 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

iv	EUTRON LILERUY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	8.6949+06	3.2943-11	2.8644-04	1.7962-01
	6.3465+06	8.1840-11	5.1956-04	1.7353-01
	5.8423+06	8.9490-11	5.2283-04	1,6906-01
	5.3944+06	1.2888-10	6.9523-04	1.3498-01
	4.9961+06	1.7230-10	8.6083-04	1.1710-01
	4.5592+06	2.4523-10	1.1181-03	8.1320-02
	4.1045+06	2.5978-10	1.0663-03	8.4599-02
	3.7147+06	3.1869-10	1.1838-03	8.0345-02
	3.3779+06	4.5276-10	1.5294-03	6.9528-02
	3.0849+06	4.8255-10	1.4886-03	7.0930-02
	2.82.4+06	6.8795-10	1.9458-03	6.1520-02
	2.6026+06	9.1978-10	2.3938-03	5.6575-02
	2.4029+06	1.1955-09	2.8727-03	5.1200-02
	2.2253+06	1.4440-09	3.2133-03	4,9257-02
	2.0424+06	1.7845-09	3.6447-03	4.1030-02
	1.8591+00	2.0708-09	3.8498-03	4.0493-02
	1.6995+06	2.6262-09	4.4632-03	3.8175-02
	1.5596+06	3.1597-09	4.9279-03	3.7476-02
	1.4363+06	3.5866-09	5.1514-03	3,7687-02
	1.3270+06	4.0887-09	5.4257-03	3,7529-02
	1.2298+06	4.7585-09	5.8520-03	3.7321-02
	1.1320+06	5.5270-09	6.2610-03	3.2907-02
	1.0378+06	6.6517-09	6,9031-03	3.2073-02
	9.5418+05	6.7069-09	6.3996-03	3.3488-02
	8.8031+05	7.5580-09	6,6534-03	3.3791-02
	8.1470+05	8.5353-09	6.9537-03	3.4790-02

RUN 7 8-5-64

COMPUTED 5-13-69

FAST HEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 53 DEGREES USING A NE-211 UETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	HEL RON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.7508+06	3.6479-11	3.5570-04	1.9428-01
7.5923+06	4.5425-11	3.4488-04	1.8717-01
0.6260+06	8,1369-11	5.3920-04	1.6967-01
6.0875+06	1.1830-10	7.2015-04	1.3545-01
5.6117+06	1.0800-10	6.0606-04	1.4984-01
5.1895+06	1.6224-10	8.4194-04	1.1980-01
4.8133+06	1.8434-10	8.8728-04	1.1654-01
4.3996+06	1.7861-10	7.8581-04	1.0425-01
3.9680+06	2.7015-10	1.0720-03	8.6605-02
3.5970+06	3.3792-10	1.2155-03	8.1081-02
3.2757+06	4.1191-10	1.3493-03	7.6549-02
2.9956+06	4.3770-10	1.3112-03	7.8503-02
2.7500+06	5.6002-10	1.5401-03	7.2697-02
2.5334+06	8.3386-10	2.1125-03	6.2356-02
2.3414+06	9,5562-10	2.2375-03	6.0649-02
2.1704+06	1.2245-09	2.6577-03	5.6698-02
1.9941+06	1.3591-09	2.7102-03	4.9838-02
1.8172+06	1.7541-09	3.1876-03	4.6349-02
1.6628+06	2.2099-09	3.6746-03	4.3822-02
1.5273+06	2.3993-09	3,6645-03	4.5398-02
1.4077+06	3.1500-09	4.4343-03	4.1990-02
1.3016+06	3.4215-09	4.4534-03	4.3172-02
1.2071+06	3.8837-09	4.6880-03	4.3413-02
1.1128+06	4.4787-09	4.9839-03	3.8293-02
1.0202+06	4.5446-09	4.6364-03	4.0923-02
9.3868+05	5.2428-09	4.9213-03	3,9635-02
8.6657+05	5.8875-09	5.1019-03	4.0393-02
8.0246+05	6,3174-09	5.0695-03	4.2760-02
RUN 5 8-5-6+

TABLE E24 COMPUTED 5-13-69

FAST NEUTRON SPECTRUM FOR 7.0 INCHES LH2 AT 53 DEGREES USING A NE-211 DETLCTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.9935+06	2.7746-11	2.7728-04	1.5654-01
8.3348+06	3.7028-11	3.0862-04	1.7688-01
7.5300+06	4.7476-11	3.5987-04	1.4478-01
6.9234+06	6.4846-11	4.4895-04	1.1556-01
6.3455+06	7.6421-11	4.8516-04	1.0584-01
5.8423+06	7.8505-11	4.5865-04	1.0894-01
5.3944+06	8.9330-11	4.8188-04	1.0446-01
4.9961+06	1.1466-10	5.7285-04	9.2653-02
4.5592+06	1.0509-10	4.7913-04	8.7009-02
4.1045+06	1.7343-10	7.1184-04	6.6713-02
3.7147+06	1.8139-10	6.7381-04	7.0535-02
3.3779+06	2.4631-10	8.3201-04	6.1918-02
3.0849+06	3.0347-10	9.3617-04	5.7641-02
2.8294+06	3,9266-10	1,1106-03	5.2435-02
2.6020+06	4.0612-10	1.0370-03	5.6429-02
2.4029+06	5.0494-10	1.2133-03	5.1703-02
2.2253+06	5.7838-10	1.2871-03	5.1257-02
2.0424+06	7.4339-10	1.5183-03	4.1349-02
1.8591+06	9.0380-10	1.6803-03	3,9626-02
1.6995+06	1.1118-09	1.8895-03	3.7885-02
1.5596+06	1.2022-09	1.8750-03	3.9508-02
1.4363+06	1.4269-09	2.0495-03	3.8658-02
1.3270+06	1.5990-09	2.1219-03	3.8908-02
1.2298+06	1.6507-09	2.0300-03	4.1685-02
1.1328+06	1.8363-09	2.0802-03	3.7683-02
1.0378+06	2.1031-09	2.1826-03	3.7683-02
9.5418+05	2.1666-09	2,0673-03	3,9084-02
8.8031+05	2.1603-09	1,9017-03	4.2676-02
8.1470+05	2.5082-09	2.0434-03	4.3406-02

COMPUTED 5-13-69

FAST NEUTRON SPECTRUM FOR 10.5 INCHES LH2 AT 53 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E\)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
4.3996+06	9,6552-11	4,2479-04	1.6514-01
3.9040+00	8.6996-11	3.4520-04	1,9503-01
3.5970+00	1.4163-10	5.0944-04	1.5060-01
3.2757+06	1.7854-10	5.8484-04	1.3711-01
2.9956+06	2.0107-10	6.0233-04	1.3602-01
2.7500+06	2.6887-10	7.3939-04	1.1998-01
2.5334+06	2,9800-10	7.5495-04	1.2309-01
2.3414+06	3,0032-10	7.0317-04	1.3003-01
2.1704+06	4.0639-10	1.0123-03	1.0549-01
1.9941+06	6.3173-10	1.2597-03	8.1433-02
1.8172+06	6.3073-10	1.1462-03	8.8719-02
1.6028+06	6.6733-10	1.1096-03	9.2751-02
1.5273+00	9.1070-10	1.3909-03	8.3493-02
1.4077+06	1.0116-09	1.4240-03	8.5138-02
1.301 06	1.0341-09	1.4111-03	8.8718-02
1.2071+06	1.3472-09	1.6262-03	8.4032-02
1.1120+06	1.3302-09	1.4802-03	8.1496-02
1.0202+06	1.5236-09	1.5544-03	8.1740-02
9.3868+05	1,4581-09	1.3687-03	8.9205-02
8.0057+05	1,3927-09	1.2069-03	1.0101-01
8.0240+05	1.6564-09	1.3292-03	1.0189-01

COMPUTED 5-13-69

RUN 3 8-4-64

FAST NEUTRON SPECTRUM FOR 13.0 INCHES LH2 AT 53 DEGREES USING A NE-211 DETLOTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
6.2194+06	3,7430-11	2,3279-04	1.6371-01
5.6117+06	5.3525-11	3.0037-04	1.6674-01
5.1895+06	6.5930-11	3.4214-04	1.5360-01
4.8133+06	6.0746-11	2.9239-04	1.7307-01
4.3990+06	6.7976-11	2.9907-04	1.4174-01
3.9650+06	8.8119-11	3.4966-04	1.2909-01
3.5970+06	1.0528-10	3.7869-04	1.2465-01
3.2757+06	1.2387-10	4.0576-04	1.1987-01
2.9950+06	1.4095-10	4.2223-04	1.1766-01
2.7500+06	2.0154-10	5.5423-04	1.0047-01
2.5334+06	2.0093-10	5.0904-04	1.0987-01
2.34:4+05	2.9211-10	6.8395-04	9.0861-02
2.170++06	2.8463-10	6.1776-04	1.0094-01
1.9941+06	3.6728-10	7.3239-04	8.0398-02
1.8172+06	4.5711-10	8.3066-04	7.6173-02
1.6028+06	4.8258-10	8.0243-04	7.9687-02
1.5273+06	5.4238-10	8.2838-04	8.1374-02
1.4077+06	6.8619-10	9.6595-04	7.6173-02
1.3010+06	7.8576-10	1.0227-03	7,5968-02
1.2071+06	8.6888-10	1.0488-03	7,7655-02
1.1120+06	9.2656-10	1.0311-03	7.2025-02
1.0202+06	9.6123-10	9,8065-04	7,6528-02
9.3868+05	8.9663-10	8.4165-04	8.4761-02
8.6657+05	9.1196-10	7.9028-04	9.2386-02
8.0246+05	1.0739-09	8.6176-04	9.3432-02

RUN 8 8-7-64

COMPUTED 5-7-69

FAST NEUTRON SPECTRUM FOR EMPTY DEWAR AT 78 DEGREES USING 5-IN. NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.0886+06	5,2629-11	4.7832-04	1.8953-01
6.7766+06	1.1804-10	7,9991-04	1.5415-01
6.0875+06	1,7609-10	1.0719-03	1.5081-01
5.6117+06	2.2654-10	1.2713-03	1.3473-01
5.1895+06	2.5589-10	1.3279-03	1.3092-01
4.8133+06	3.2014-10	1.5409-03	1.2064-01
4.3996+06	3.7026-10	1.6290-03	9.6601-02
3.9640+06	4.6669-10	1.8518-03	9.0837-02
3.5970+06	5.7781-10	2.0784-03	8.6807-02
3.2757+06	5.7754-10	1.8918-03	9.1641-02
2.9956+06	7.6439-10	2.2898-03	8.3307-02
2.7500+06	9.6135-10	2.6437-03	7.8545-02
2.5334+06	1.2163-09	3.0814-03	7.4520-02
2.3414+06	1.3653-09	3,1967-03	7.3506-02
2.1704+06	1.8691-09	4.0567-03	6.6816-02
1.9941+06	2.1711-09	4.3294-03	5,7083-02
1.8172+06	2.3936-09	4.3496-03	5.8035-02
1.6628+06	2.9001-09	4.8223-03	5,6088-02
1.5273+06	3.0649-09	4.6810-03	5.8729-02
1.4077+06	3.7828-09	5.3250-03	5.6624-02
1.3016+06	4.7220-09	6.1462-03	5.4008-02
1.2071+05	5.2544-09	6.3426-03	5.4895-02
1.1128+06	6.3693-09	7.0878-03	4,7160-02
1.0202+06	6.3790-09	6.5079-03	5.0654-02
9.3868+05	6.8827-09	6.4607-03	5.0850-02
8.6657+05	7.6223-09	6.6053-03	5,2074-02
8.0246+05	9.8943-09	7.9398-03	4.9831-02

UN 7 :-7-64

AST HEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 78 DEGREES USING A NE-211 ETECTON BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECULLIMATOR NERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON LNERUY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E1)	(RELATIVE UNITS)	(RELATIVE UNITS)	LINCERTAINTY
5.3944+06	1.1670-10	6,2953-04	1.1777-01
4.9961+06	1.3523-10	6.7562-04	1,1297-01
4.5552+06	1.5716-10	7.1652-04	8.9952-02
4.1045+06	2.1946-10	9.0077-04	7.9382-02
3.7147+06	2.2950-10	8.5252-04	8.3391-02
3.3779+06	2.9522-10	9.9722-04	7.6618-02
3.001.9+06	3.3548-10	1.0349-03	7.5155-02
2.04.4+46	3.8816-10	1.0979-03	7.4160-02
2.6020+06	4.5071-10	1.1730-03	7.3391-02
2.40.9+06	5.7831-10	1.3896-03	6,6656-02
2.2253+06	5.8012-10	1.3043-03	7.1047-02
2.04:4+06	7.7965-10	1.5924-03	5.6620-02
1.8591+06	9.3371-10	1.7359-03	5.4699-02
1.0995+06	1.0959-09	1.8625-03	5.3846-02
1.5590+06	1.1185-09	1.7444-03	5.7796-02
1.4363+06	1.3068-09	1.9631-03	5.5845-02
1.3270+06	1.4934-09	1.9817-03	5.7154-02
1.2290+00	1.8549-09	2.2812-03	5.4620-02
1.1308+06	1.9168-09	2.1714-03	5.1612-02
1.0378+06	2.3482-09	2.4370-03	4.9490-02

UN 6 -7-64

AST NEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 78 DEGREES USING A NE-211 DETECTOR DIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR INERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON LINERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
5.8423+06	9.6543-11	5,6403-04	1.3199-01
5.3944+06	1.2567-10	6.7791-04	1.1612-01
4.9961+00	1.2853-10	6.4215-04	1.2064-01
4.5592+06	1.5620-10	7.1215-04	9.3240-02
4.1045+06	1.3593-10	7.6315-04	9.0310-02
3.7147+06	2.0130-10	7.4777-04	9.3630-02
3.3779+06	2.5749-10	8.6978-04	8.5630-02
3.0849+06	2,9035-10	8.9570-04	8.4728-02
2.0244+06	3,1521-10	8.9154-04	8.6152-02
2.60.0+06	4.0546-10	1.0553-03	8.0747-02
2.4029+06	5.1588-10	1.2396-03	7.3062-02
2.2253+00	4.9977-10	1,1121-03	8.0496-02
2.0424+06	6.6290-10	1.3539-03	6.3729-02
1.8591+06	7.2782-10	1.3531-03	6.4990-02
1.6995+06	8.3732-10	1.5080-03	6.2303-02
1.5590+06	9.3892-10	1.4643-03	6.5917-02
1.4363+06	1.1291-09	1.6217-03	6.3975-02
1.3270+06	1.2603-09	1.6724-03	6.4477-02
1.2298+06	1.5011-09	1.8461-03	6.3484-02
1.1323+06	1.5391-09	1.7435-03	6.0129-02
1.0373+06	1.9421-09	1.9117-03	5.8649-02
9.5410+05	1.3962-09	1.8093-03	6.0856-02
8.8031+05	1.9100-09	1.6814-03	6.5950-02
8.1470+05	2.3470-09	1.9121-03	6.4659-02

KUN 5 -7-64

COMPUTED 5-7-69

FAST NEUTRON SPECTRUM FOR 7.0 INCHES LH2 AT 78 DEGREES USING A NE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

ŀ	LUTRON LHERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(cv)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	5.3944+06	1,2436-10	6,7085-04	9,3993-02
	4.9961+06	1.1995-10	5,9928-04	1.0218-01
	4.5592+06	1.6450-10	7.4999-04	7.3221-02
	4.1045+06	1.7913-10	7.3524-04	7.5096-02
	3.7147+06	2.2769-10	8.4580-04	7.0222-02
	3.3779+06	2.2688-10	7.6638-04	7.5168-02
	3.0849+06	2.8507-10	8.7941-04	6.9384-02
	2.02.4+06	2.7903-10	7.8921-04	7.5540-02
	2.60:0+06	3,9828-10	1.0366-03	6.6164-02
	2.4029+06	4.2075-10	1.0110-03	6.7570-02
	2.2253+06	4.9793-10	1.1030-03	6.5623-02
	2.0424+06	5.5216-10	1.1277-03	5,7980-02
	1.8591+36	5.6322-10	1.0471-03	6.1901-02
	1.6995+06	6.2935-10	1.06%6-03	6.2687-02
	1.5590+06	7.9618-10	1,2417-03	5,9208-02
	1.4363+06	8.4141-10	1.2085-03	6.2573-02
	1.3270+06	1.0629-09	1.4105-03	5.8350-02
	1.2298+06	1.1065-09	1.3608-03	6.2234-02
	1.1320+06	1.2989-09	1.4714-03	5.4454-02
	1.0378+06	1.3414-09	1.3921-03	5.8171-02
	9.5418+05	1.5781-09	1,5058-03	5,5380-02

COMPUTED 5-7-69

.: UN +

-7-04

PAST NEUTRON SPECTRUM FOR 10.5 INCHES LH2 AT 78 DEGREES USING A HE-211 DETECTOR BIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR LNENGY RESILUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTION E	NERGY N	EUTRUN FLUX	FLUX*ENERGY	RELATIVE
(ヒ.)	(RE	LATIVE UNIT	S) (RELATIVE UNITS)	UNCERTAINTY
0.9234	+00	6.9825-11	4.8343-04	1.4249-01
5.34 3	+96	8.7279-11	5.5409-04	1,2554-01
5.8423	+00	9.2050-11	5.4134-04	1.2632-01
5.3944	+uo	1.2742-10	6.8735-04	1.0715-01
4.99.1	+00	1.2912-10	6.4510-04	1.1208-01
4.5002	+06	1.4534-10	6.6263-04	9.0617-02
4.1045	+00	1546-10	6.7913-04	9.0354-02
3.7117	+06	1.9865-10	7.3793-04	8.7504-02
3.5174	+00	2.2098-10	7.4645-04	8.7385-02
3.00+9	+00	2.2058-10	6.8047-04	9.3405-02
2.82.4	+ 10	2.9212-10	8.2623-04	8.3826-02
2.0000	+uó	3.4592-10	9.0810-04	8.1959-02
2.40.9	+10	3.0517-10	9.2072-04	8,1698-02
2.2200	+00	3.7121-10	8.2005-04	9.0176-02
2.0424	+16	4.7253-10	9.6510-04	7.3250-02
1.8591	+06	4.7512-10	8.8330-04	7.8915-02
1.0995	+46	6.5057-10	1.1328-03	6.9058-02
1.5596	+00	6.5373-10	1.0196-03	7.6962-02
1.4303	+05	8. 1482-10	1.1560-03	7.3432-02
1.3270	+06	8212-10	1.1308-03	7.6546-02
1.2298	+00	1.0071-09	1.2385-03	7.5733-02
1.13.20	+00	1.0293-09	1.1660-03	7.1897-02
1.0370	+46	1.1754-09	1.2198-03	7.2237-02
9.5410	+15	1.2389-09	1.1821-03	7.4010-02
8.80.1	+05	1.3036-09	1.2180-03	7.5133-02
8.1470	+05	1.4774-09	1,2036-03	8.0841-02

RUN 3 -7-64

COMPUTED 5-7-69

FAST NEUTRON SPECTRUM FOR 13.0 INCHES LH2 AT 78 DEGREES USING A NE-211 DETECTOR GIAS = .032 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR LNERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

HEUTRON LNERGY	LUTRON FLUX	FLUX*ENERGY	RELATIVE
( = . / )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
5.61.4+06	1.1814-10	6.6376-04	1.9938-01
4.7339+06	1.2535-10	5.9339-04	1.9545-01
3.9725+06	1.7036-10	6.7676-04	1.8161-01
3.3011+06	2.0531-10	6.9417-04	1.8402-01
2.8728+00	2.1500-10	5.8892-04	1.9022-01
2.5005+06	3.1870-10	7.9691-04	1.9928-01
2.2407+00	3.6265-10	8.0751-04	1.8388-01
1.97:2+05	3,9658-10	7.6203-04	1.9636-01
1.73.0+06	4.3304-10	8.3977-04	1.7640-01
1.5590+06	6.3207-10	1.0638-03	1.9233-01
1.40.4+06	5.9466-10	8.3752-04	1,9019-01
1.10.3+06	4.6923-10	5.5759-04	1.9937-01
9.8732+05	7.5067-10	7.4115-04	1.9570-01
3.6033+05	1.1227-09	9.6589-04	1.8445-01

RUN T4 -20-06

FAST NEUTRON, SPECTRUM FOR EMPTY DEWAR AT 0 DEGREES USING 2-IN. HE-211 DEFECTOR BIAS = .04233 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E))	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.3927+07	6.7078-07	9,3420+00	1.0344-01
1.23 30+47	1.1407-06	1,4072+01	7.4800-02
1.1002+07	1.9153-06	2.1072+01	5.8704-02
9.87.2+06	3.5870-06	3,5419+01	4.5244-02
8.9111+06	5.5044-06	4,9050+01	3.9143-02
8.08-4+06	9.5435-06	7.7134+01	3.1889-02
7.30-1+06	1.3242-05	9.7515+01	2.8971-02
6.7.37.+46	1.7024-05	1,1874+02	2.6865-02
0.1070+06	2.1912-05	1.3558+02	2.5681-02
5.7023+06	2.5928-05	1.4785+02	2.5033-02
5.2/20+06	3.0309-05	1.5979+02	2.4306-02
4.8057+06	3.6635-05	1.7910+02	2.3357-02
4.4073+06	4.3759-05	1.9548+02	1.8563-02
4.0279+06	5.2053-05	2.0966+02	1.8100-02
3.03.4+06	6.2260-05	2.2727+02	1.7587-02
3.3235+95	7.2175-05	2.3987+02	1.7138-02
3.03.7+06	8.9145-05	2.7088+02	1.6098-02
2.78.9+06	1. 1136-04	2.8268+02	1.5863-02
2.50 0+00	1.1659-04	2,9950+02	1.5577-02
2.3737+06	1.5474-04	3,1983+02	1.5094-02
2.2000+06	1.5617-04	3,4357+02	1.4697-02
2.0210+06	1.8237-04	3.6857+02	1.2412-02
1.8413+06	2.1656-04	3.9875+02	1.2051-02
1.0040+06	2.5255-04	4.2545+02	1.1764-02
1.5471+06	2.7225-04	4,2120+02	1,2066-02
1.4250+06	3.0687-04	4.3754+02	1.1979-02
1.3182+36	3.4117-04	4.4973+02	1.2019-02
1.2223+00	3.6841-04	4,5031+02	1.2250-02
1.1267+06	4.5123-04	5.0840+02	1.0639-02
1.0328+06	4.6736-04	4.8269+02	1.1432-02
9.50 0+05	5.2763-04	5.0133+02	1.1325-02
8.77 7+05	5.6767-04	4.9789+02	1.1777-02
5.1<11+05	5.9509-04	4.8328+02	1.2580-02

FUN 13 -20-06

FAST NEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 0 DEGREES USING A NE-211 UETLOTON BIAS = .04233 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR UNERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON LUEROY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(LV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
		<	6 HOH7 00
1.39.2+07	4.4194-07	6,1527+00	6.4243-02
1.23:2+07	7.5498-07	9.3104+00	5.0765-02
1.0999+07	1.4916-06	1.6406+01	3.8693-02
9.6/09+06	2.7133-06	2.6783+01	3.1272-02
8.90 1+06	4.6089-06	4.1057+01	2.6069-02
8.0790+00	6.3692-06	5.5500+01	2.3122-02
7.3610+06	9.2457-06	6.8063+01	2.1428-02
6.73:1+06	1.2355-05	8.3212+01	1.9920-02
6.10-4+00	1.5409-05	9.5311+01	1.9051-02
5.7004+06	1.0210-05	1.0380+02	1.8636-02
5.2702+06	1.9816-05	1.0443+02	1.8773-02
4.8870+06	2.3057-05	1.1268+02	1.8416-02
4.4057+06	2.6839-05	1.1985+02	1.4878-02
4.0205+06	2.9965-05	1.2065+02	1.4996-02
3.6491+06	3.5538-05	1.2968+02	1.4676-02
3.3223+46	4.1758-05	1.3873+02	1.4254-02
3.0370+00	4.6384-05	1.4241+02	1.4072-02
2.7079+06	5.2767-05	1.4711+02	1.3977-02
2.5078+06	5.314-05	1.4974+02	1.4029-02
2.37.0+00	6.5828-05	1.5620+02	1,3803-02
2.1992+06	7.5689-05	1.6646+02	1,3531-02
2.0202+06	8.0724-05	1.6308+02	1.1994-02
1.84.7+05	9.2049-05	1.6943+02	1.1925-02
1.00.0+00	1.0137-04	1,7071+02	1,2007-02
1.5+60+06	1.0155-04	1.5706+02	1.2823-02
1.4253+06	1.1293-04	1.6096+02	1.2846-02
1.3177+06	1.1575-04	1.5252+02	1.3455-02
1.22:9+06	1.1920-04	1.4565+02	1.4076-02
1.1252+06	1.4229-04	1.6025+02	1.2401-02
1.03:4+06	1.4130-04	1.4588+02	1.3635-02
9.49-1+15	1.5046-04	1.4291+02	1.3936-02
8.7075+05	1.6308-04	1.4298+02	1.4468-02
8.11.1+45	1.0296-04	1.3229+02	1.5857-02

RUN T2 2-20-06

CO 1PUTED 4-30-69

FAST NEUTRON SPLCTRUM FOR 7.0 INCHES LH2 AT 0 DEGREES USING A NE-211 DETECTOR BIAS = .04233 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR LUERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTPON FLUX	FLUX*ENERGY	RELATIVE
(E,)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	1 000 07	0.0047.00	6 0677 00
1.4036+07	1.3905-07	2.8047+00	6.2033-02
1.3091+07	3.9275-07	5.1415+00	4,2367-02
1.1037+07	7.1945-07	8.3722+00	3.2481-02
1.04:2+07	1,2856-06	1.3386+01	2.6196-02
9.3710+06	2,1206-06	1.9872+01	2.2223-02
8.47-7+00	3.4302-06	2,9084+01	1.8960-02
7.70-1+06	5.0944-06	3.9268+01	1.6804-02
7.0379+06	6.2589-06	4.4050+01	1.6319-02
6.4515+06	7.9300-06	5.1164+01	1.5551-02
5.9355+00	9.1368-06	5.4528+01	1.5372-02
5.4790+06	9.9163-06	5.4331+01	1.5604-02
5.0732+06	1.0643-05	5.3994+01	1.6047-02
4.62 2+06	1.1096-05	5,4131+01	1.3313-02
4.1054+06	1.3077-05	5.4 171+01	1.3496-02
3.70-3+05	1,4381-05	5.4199+01	1.3728-02
3.4202+00	1.5052-05	5.1571+01	1.4174-02
3.12.3+00	1.6509-05	5.1645+01	1.4173-02
2.8576+06	1.7609-05	5.0496+01	1.4426-02
2.03 2+06	1.8038-05	4.7588+01	1.5068-02
2.43:3+05	1.9293-05	4.6984+01	1.5256-02
2.2349+06	2.1042-05	4.7448+01	1.5263-02
2.0002+00	2.1215-05	4.3898+01	1.3938-02
1.8332+06	2.2561-05	4.2487+01	1.4323-02
1.7212+06	2.2340-05	3,9312+01	1.5022-02
1.5790+00	2.2618-05	3,5721+01	1.6078-02
1.4342+06	2.2855-05	3.3236+01	1.6919-02
1.3434+06	2.2777-05	3,0599+01	1.7848-02
1.24+2+06	2.0349-05	2.5330+01	2.0055-02
1.1460+06	2.2789-05	2.6130+01	1.8116-02
1.05:2+00	2.1449-05	2.2526+01	2.0421-02
9.6551+05	1.9964-05	1.9275+01	2.2360-02
8.9057+05	2.0403-05	1.8172+01	2.3695-02
8.2420+05	1.8733-05	1.5440+01	2.7037-02

KUN TI , -20-06 TABLE E36 COMPUTED 4-30-49

HAST NEUTRUM SPLCTRUM FOR 13.0 INCHES LH2 AT 0 DEGREES USING A ME-211 DETECTOR BIAS = .04233 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR LNERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(ビリ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1 685 407	1 4090-07	2 1802+00	4 8429-02
1.4030407	2 7953-07	3 6500+00	3 5237-02
1.3091+07	2.7952-07	5.0392+00	3.9700-02
1.103/+07	4.5009-07	3.3308+00	2.0700-02
1.0412+07	7.5169-07	7.8266+00	2.4369-02
9.3710+06	1.2728-06	1.1927+01	2.0419-02
8.41. 7+06	1.7893-06	1.51/1+01	1.8656-02
7.70 11+06	2.5701-06	1.9811+01	1.6809-02
7.0579+06	3.0141-06	2.1213+01	1.6683-02
0.4515+06	3.4777-06	2.2436+01	1.6638-02
5.93:5+06	3.7804-06	2.2439+01	1.6962-02
5.4790+06	4.0114-06	2.1978+01	1.7330-02
5.0732+06	4.1862-06	2.1237+01	1.8040-02
4.62:2+06	4.3619-06	2.0188+01	1.5334-02
4.1054+06	4.5627-06	1.9005+01	1.6034-02
3.7048+06	4.4574-06	1.6799+01	1.7240-02
3.4202+06	4.6230-06	1.5839+01	1.7838-02
3.12.3+06	4.0073-06	1.4601+01	1.8542-02
2.8076+00	4.5749-06	1.3119+01	1.9628-02
2.63,2+06	4.4990-06	1.1869+01	2.0853-02
2.4353+46	4.3645-06	1.0629+01	2.2131-02
2.2549+06	4.0862-06	9.2140+00	2.3845-02
2.0052+06	4.1350-06	8.5561+00	2.1654-02
1.0032+00	4.0795-06	7.6825+00	2.3029-02
1.7212+06	3.8023-06	6.5445+00	2.5090-02
1.5793+06	3. 3813-06	5.3401+00	2.8266-02
1.450.2400	3.0226-06	4.3955+00	3.1582-02
1.3434440	2 9078-06	3,9063+00	3.3833-02
1.24/14/00	2 4590-06	3.0610+00	3.8966-02
1 100 100	2 4238-06	2 7791+00	3 7490-02
1 (50)+00	2 2790-06	2 3934400	4 2164-02
0.6552406	1 7317-06	1 6720+00	5 1130-02
9.0351+05	1 5090-06	1 3440+00	5 0768-02
8.9067+05	1.5090-06	1.0170.00	5.8/60-02
8.2420+05	1.4/1/-06	1,2130+00	6.4880-02

TABLE E37 COMPUTED 4-14-69

RUN 15 2-24-66

FAST NEUTRON SPECTRUM FOR EMPTY DEWAR AT 37 DEGREES USING 5-IN. NE-211 DETECTOR BIAS = .0447 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.0240+07	2.8976-09	2.9671-02	1.8187-01
8.9850+06	5.0621-09	4.5483-02	1.7756-01
8.1454+06	8.214/-09	6.6912-02	1.4683-01
7.4182+06	1.5971-08	1.1848-01	1.1179-01
6.784.0+06	1,9160-08	1.2999-01	1.0976-01
6.2282+06	3.0418-08	1.8945-01	9.2909-02
5.7378+06	3.0808-08	1.7677-01	9.9062-02
5.3032+06	4.8923-08	2.5945-01	8.2829-02
4.9161+06	5.5742-08	2.7403-01	8.1718-02
4.4909+06	6.9955-08	3.1416-01	6.3888-02
4.0477+06	7.598/-08	3.0757-01	6.5791-02
3.6671+06	1.0989-07	4.0298-01	5.8132-02
3.3377+06	1.4669-07	4.8961-01	5.3086-02
3.0508+06	1.6445-07	5.0170-01	5.2498-02
2.7994+06	2.1378-07	5.9846-01	4.8405-02
2.5778+06	2.848/-07	7.3434-01	4.4206-02
2.3815+06	2.7034-07	6.4381-01	4.7523-02
2.2068+06	3.2248-07	7.1165-01	4.5751-02
2.0268+06	3.9686-07	8.0436-01	3.7746-02
1.8462+06	4.5655-07	8.4288-01	3.7454-02
1.6888+06	5.5223-07	9.3261-01	3.6018-02
1.5506+06	6.3830-07	9.8975-01	3.5856-02
1.4288+06	7.5938-07	1.0850+00	3.4798-02
1.3208+06	9.3333-07	1.2327+00	3.3227-02
1.2245+06	9.2272-07	1.1299+00	3.5426-02
1.1285+06	1.0442-06	1.1784+00	3.2110-02
1.0343+06	1.2430-06	1.2856+00	3.2176-02
9.5145+05	1.2923-06	1.2296+00	3.3372-02
8.7815+05	1.4630-06	1.2847+00	3.4041-02
8.1300+05	1.8958-06	1.5413+00	3.2758-02

RUN T3 2-24-66

FAST NEUTRON SPECTRUM FOR 2.5 INCHES LH2 AT 37 DEGREES USING A NE-211 DETECTOR BIAS = .0447 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
9.0021+06	2,9506-09	2.6562-02	1.4588-01
7.7691+06	6.7115-09	5.2142-02	1.4220-01
7.0906+06	8.3162-09	5.8967-02	1.3690-01
6,4973+06	1.5832-08	1.0287-01	1.0383-01
5.9755+06	1.9308-08	1.1537-01	1.0065-01
5.5141+06	2.4171-08	1.3328-01	9.5514-02
5.1042+06	3,4425-08	1.7570-01 .	8.4089-02
4.6549+06	5.0760-08	2,3628-01	6.0162-02
4.1879+06	6.3792-08	2.6715-01	5.7714-02
3.7878+96	8,3231-08	3,1526-01	5.3907-02
3.4424+06	1.0977-07	3.7787-01	4.9568-02
3.1422+06	1.4528-07	4.5650-01	4.4997-02
2.8796+06	1.7942-07	5,1666-01	4.2530-02
2.6480+06	2.3045-07	6.1037-01	3.9659-02
2.4444+06	2,4870-07	6.0792-01	4.0071-02
2.2629+06	2,9870-07	6,7593-01	3.8215-02
2.0761+06	3.330/-07	6,9149-01	3.3305-02
1.8891+06	4.0926-07	7.7513-01	3.1920-02
1.7262+06	5.2121-07	8.9971-01	2.9902-02
1.5835+06	5.4134-07	8,5721-01	3.1443-02
1.4579+06	6.0765-07	8,8589-01	3.1537-02
1.3466+06	7.4088-07	9.9767-01	3.0132-02
1.2476+06	7.7210-07	9.6327-01	3.1366-02
1.1489+06	9,2576-07	1.0636+00	2.7482-02
1.0522+06	1.0053-06	1.0578+00	2.8938-02
9.6721+05	1.1094-06	1.0730+00	2.9146-02
8.9211+05	1.2719-06	1.1347+00	2.9384-02
8.2544+05	1.3387-06	1,1050+00	3.1431-02
7.6597+05	1,5982-06	1.2242+00	3.1794-02

COMPUTED 4-14-69

RUN 12 2-24-06

FAST NEUTRON SPECTRUM FOR 4.5 INCHES LH2 AT 37 DEGREES USING A HE-211 DETECTOR BLAS = .0447 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERUY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(ヒッ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.0846+07	8.8068-10	9.5519-03	1,9996-01
8.9850+06	2.9685-09	2.6672-02	1.6618-01
8.1454+06	3,9092-09	3,1842-02	1.5448-01
7.41 12+46	5.5735-09	4.1345-02	1.3681-01
6.7343+46	9.7064-09	6.6258-02	1.0806-01
6.2202+06	1.3182-08	8.2100-02	9,9310-02
5.7378+06	1.7970-08	9.7944-02	9.3157-02
5.3052+06	2.5511-08	1.3529-01	7.9838-02
4.9161+06	3.3446-08	1.6442-01	7.3182-02
4.4904+16	4.5333-08	2,0359-01	5.4781-02
4.04771.06	5 5694-118	2,2543-01	5.2841-02
3.6671+46	7.5853-08	2.7816-01	4.8171-02
3.3377+116	1.0262-07	3,4251-01	4.3570-02
3.0500+06	1,1944-07	3.6439-01	4.2306-02
2.7904+00	1.5141-07	4,2386-01	3.9478-02
2.5/78+06	1.6893-07	4.3547-01	3.9509-02
2.3415+06	2,1095-07	5.0238-01	3.6897-02
2.20.8+06	2.4026-07	5.3021-01	3.6357-02
2.02.4+06	2.8776-07	5.8323-01	3.0418-02
1.8462+06	3,1768-07	5.8650-01	3.0834-02
1.68.8+06	3.5295-07	5,9606-01	3.0937-02
1.5500+06	4.1045-07	6.4575-01	3.0446-02
1.42.04.16	4.4757-07	6.3949-01	3.1162-02
1.3200+06	5,2403-07	6.9214-01	3.0503-02
1.2405+06	5.0456-07	6.9130-01	3.1147-02
1.1205+06	6.2335-07	7.0345-01	2.8583-02
1.03.3+00	7.0066-07	7.2469-01	2.9487-02
9.510.405	7.6208-07	7.2508-01	2.9839-02
8.7415+115	8-5584-07	7.5156-01	3.0597-02
8.1001+05	9.0541-07	7.3610-01	3.2697-02

RUN T1 2-24-06

COMPUTED 4-14-69

FAST MEUTRON SPECTRUM FOR 10.5 INCHES LH2 AT 37 DEGREES USING A ME-211 DETECTOR BIAS = .0447 STANDARD LIGHT UNITS 0.87 INCH PRECOLLIMATOR ENERGY RESOLUTION FOR GROUPING = .07 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENLAGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E <sub>V</sub> )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
8.77:1+06	1,5037-09	1,3188-02	1.8171-01
7.7001+00	3.2099-09	2.4938-02	1.5201-01
7.0900+00	5.1451-09	3.6482-02	1.2434-01
6.4973+06	7.5588-09	4.9112-02	1.0865-01
5.9755+06	1.0213-08	6.1028-02	9.8501-02
5.5141+06	1.3928-08	7.6800-02	8.8927-02
5.1042+06	1.7761-08	9.0656-02	8.2744-02
4.6549+06	2.2142-08	1.0307-01	6.4482-02
4.1079+06	3.2594-08	1.3650-01	5.6561-02
3.7878+06	3.9004-08	1,4774-01	5.5014-02
3.44,4+06	5.4055-08	1.8608-01	4.9162-02
3.1422+00	6.0734-08	1.9084-01	4.8525-02
2.8790+06	7.1260-08	2.0520-01	4.7112-02
2.0480+06	8.7349-08	2.3135-01	4.4798-02
2.4444+06	1.0230-07	2,5006-01	4.3390-02
2.2029+06	1.0521-07	2.3808-01	4.4888-02
2.0761+06	1.2497-07	2.5945-01	3,7799-02
1.8891+06	1.4174-07	2.6776-01	3.7719-02
1.7262+06	1.6961-07	2.9278-01	3.6402-02
1.5835+06	1.8613-07	2.9474-01	3.7247-02
1.4579+66	1.8461-07	2.6914-01	3,9960-02
1.3460+06	1.9656-07	2.6469-01	4.0913-02
1.2476+06	2.2236-07	2.7742-01	4.0845-02
1.14.9+06	2.3801-07	2.7345-01	3.7921-02
1.0522+06	2.5068-07	2,6377-01	4.0821-02
9.0721+05	2.6983-97	2,6098-01	4.1691-02

RUN T2 0-10-06

COMPUTED 0-23-69

INTERMEDIATE NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 78 DEGREES USING A DETECTOR COMPOSED OF A 64C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESILUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTROL ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2 210 404	7 0376-07	1 7394+00	5 0038-02
2.2195+06	1.650.06	1.0354+00	5.9030-02
1.0015+06	1.1000-00	1.9356+00	5.5910-02
1.2903+06	1.3750-06	1.7742+00	6.0423-02
9.3076+05	2.4979-06	2.3399+00	3.7910-02
6.4729+05	4.0026-06	2.5908+00	3.7377-02
4.7399+05	5.8988-06	2.7960+00	4.0200-02
3.62.0+05	7.2113-06	2.6109+00	4.7151-02
2.7125+05	9.5700-06	2.5959+00	4.1569-02
2.0029+05	1,3610-05	2.7259+00	4.7853-02
1.5304+05	1.9808-05	3.0492+00	5.2948-02
1.1794+05	2.2934-05	2.7048+00	5.4265-02

INTLEMEJIATE HEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 78 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

HEUTRON ENERGY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E.)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	9.3969-07	2,0855+00	5.3202-02
1.0015+06	1.3237-06	2.1993+00	5.1757-02
1.29,3+06	1.7317-06	2.2344+00	5.3127-02
9.3070+05	2.7989-06	2.6219+00	3.5610-02
6.4729+05	4.3555-06	2.8193+00	3.5232-02
4.7399+05	7.2567-06	3.4396+00	3,5930-02
3.6206+45	8.4649-06	3.0648+00	4,2930-02
2.7125+05	1.3116-05	3.5577+00	3.5313-02
2.0029+05	1.7120-05	3.4290+00	4.2111-02
1.5394+05	1.9074-05	2.9363+00	5.3082-02
1.1794+05	1.7564-05	2.0715+00	5,9954-02
9.0133+04	1.8250-05	1.6449+00	7.9255-02
6.9235+04	1.2668-05	8,7783-01	9.8295-02
5.3520+04	1.3868-05	7.4222-01	1.1215-01

KUN TO 3-17-06

COMPUTED 6-23-69

INTERMEDIATE NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 78 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESULUTION FOR GROUPING = .20 LARGEST FRACTIONAL EPROR = .20

i.	EUTRON ENERGY	NEUTRON FLUX	FLIJX*ENERGY	RELATIVE
	(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	2.2193+06	8.6982-07	1,9304+00	7.0817-02
	1.0015+06	1.0840-06	1.8011+00	7.3245-02
	1.29:3+06	2.1274-06	2.7450+00	6.1383-02
	9.3070+05	3.5000-06	3.2787+00	4.0629-02
	0.47:9+05	4.8208-06	3,1205+00	4.3282-02
	4.7399+05	7.2387-06	3.4311+00	4.6186-02
	3.04 00+05	1.0343-05	3.7448+00	4.9703-02
	2.7125+05	1.2015-05	3.2591+00	4.7235-02
	2.0029+05	1.1169-05	2.2370+00	6.8168-02
	1.5394+05	1.6439-05	2.5306+00	7.1211-02
			-	

TABLE E44

COMPUTED 6-23-69

INTERMEDIATE NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 78 DEGREES USING A DETECTOR COMPUSED OF A 04C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

HUTRON ENERGY	NEUTRON FLUX	FL JX*ENERGY	RELATIVE
(Lv)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.5914-06	3,5318+00	6.0802-02
1.00,5+06	2.6315-06	4.3722+00	5.4595-02
1.29-3+06	3.1735-06	4.0948+00	5.8371-02
9.3070+05	6.0092-06	5.6292+00	3,6037-02
6.47,9+05	8.5019-06	5.5550+00	3.7477-02
4.7399+05	1.3980-05	6.6264+00	3.8125-02
3.6200+05	1.9088-05	7.1282+00	4.1812-02
2.71. 0+05	2.8705-05	7.7862+00	3,5567-02
2.00:9+05	3.9501-05	7.9117+00	4.1867-02
1.5394+35	5.3880-05	9.0640+00	4.4879-02
1.1794+05	6.9386-05	8.1834+00	4.5175-02
9.0130+04	9.5521-05	8,6096+00	5.2058-02
6.9295+04	1.0851-04	7.5192+00	5.0998-02
5.3520+04	1.4357-04	7.6839+00	5.6033-02
4.17:0+04	9.2440-05	3.8573+00	7.5378-02
3.2775+04	2.2923-04	7.5130+00	6.8843-02
2.6014+04	4.0517-04	1.0536+01	6.3693-02
2.00:4+04	2.3258-04	5.1789+00	6.7574-02
1.61.3+04	3.4120-04	5,5080+00	7.1338-02

HUN T10 3-17-06

COMPUTED 6-23-69

INTERMEDIATE NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 78 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESELUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROP = .20

NEUTRON ELLERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	2.0287-06	4,5023+00	5.2559-02
1.0015+06	2.5927-06	4.3078+00	5.3683-02
1.29.3+06	3.9860-06	5.1431+00	5.0833-02
9.3076+05	6.7189-06	6.2940+00	3.3095-02
6.4729+05	1.0560-05	6.8393+00	3.3077-02
4.7399+05	1.7953-05	8.5095+00	3.3260-02
3.6200+05	2.3844-05	8,6330+00	3.7113-02
2.71:5+05	3.4792-05	9.4373+00	3.1560-02
2.00:9+05	5.2284-05	1.0472+01	3.5245-02
1.5344+05	6.5740-05	1.0120+01	4.1451-02
1.1794+05	8.2961-05	9.7844+00	3.9873-02
9.0133+04	1.2451-04	1.1222+01	4.4281-02
0.9295+04	1.1371-04	7.8795+00	4.8679-02
5.35.0+04	1.6546-04	8.8554+00	4.9938-02
4.1723+04	1.9643-04	8.1966+00	4.9938-02
3.2775+04	2.8321-04	9.2822+00	5.6433-02
2.60:4+04	4.5747-04	1.1896+01	6.1780-02

NUN TI1 3-17-00

COMPUTED 6-26-69

INTERMEDIATE NEUTRON SPECTRUM FOR EMPTY LH2 DEWAR AT 78 DEGREES USING A DETLETOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR CHERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	PELATIVE
(ビッ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	2.8859-06	6.4047+00	1.5659-01
1.0015+00	5.8645-06	9.7439+00	1.2157-01
1.2903+06	7.9884-06	1.0307+01	1.2245-01
9.3070+05	1.3182-05	1.2348+01	8.0380-02
6.4729+05	2.3055-05	1,4923+01	7.5104-02
4.7399+05	4.1197-05	1.9527+01	7.2351-02
3.02.0+05	4.9007-05	1.7961+01	8.8214-02
2.7125+05	6.7753-05	1.8378+01	7.8724-02
2.0029+05	7.2941-05	1.4609+01	1.0901-01
1.539++05	7.1045-05	1.0937+01	1.6253-01
1.1794+05	1.2732-04	1.5016+01	1.2838-01
8.5115+04	1.1037-04	9.3941+00	1.9376-01
4.6370+04	1.6648-04	7.7207+00	1.9943-01

TABLE 447

COMPUTED 6-23-69

INTERMEDIATE NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 53 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR CHERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ľ/)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	5,0523-07	1,1213+00	9.0167-02
1.6015+06	1.3032-06	2,1653+00	6,4820-02
1.29:3+06	1.5530-06	2,1329+00	6.7574-02
9.3070+05	2.6546-06	2.4867+00	4.4901-02
0.4729+05	3.5846-06	2.3203+00	4.8737-02
4.7399+05	5.9418-06	2,7690+00	5.0315-02
3.62.0+05	5.9623-06	2.1587+00	6.3372-02
2.7125+05	8.2502-06	2.2379+00	5.4965-02
2.0029+05	1.3776-05	2.7592+00	5.8222-02
1.5394+05	1.1457-05	1.7637+00	8.5436-02
1.1794+35	1.3850-05	2,2232+00	7,1796-02

COMPUTED 0-23-69

RUN 115 3-1/-66

INTERMEDIATE NEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 53 DEGREES USING A DETECTOR COMPOSED OF A 34C-PARAFFIN DISK AND A NE-226 SCINTILLATOR CHERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

WEUTRON ENERGY	HELLTRON FLUX	FLUX*FUERGY	RELATIVE
(E,)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2195+06	8.0943-07	1,7964+00	5.5806-02
1.0015+06	1.3043-06	2,1671+00	5.0761-02
1.29:3+06	2.0066-06	2.6665+00	4.7346-02
9.3070+05	3.0128-06	2.8223+00	3.3201-02
6.47.9+05	4.1904-06	2.7124+00	3,5285-02
4.7399+05	6.7843-06	3,2157+00	3.5940-02
3.6216+05	9.2968-06	3,3660+00	3.9772-02
2.7125+05	8.9633-06	2,4313+00	4.1089-02
2.0029+05	1.1090-05	2,2212+00	5.0233-02
1.5094+05	1,5236-05	2.3454+00	5.6769-02
1.1794+05	9.0766-06	1.0705+00	8.1541-02
8.5115+04	1.7018-06	1.4996-01	1.8078-01

INTERMEDIATE DEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 53 DEGREES USING A DETECTOR COMPOSED OF A 34C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
( = , )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2195+06	1.1152-06	2,4750+00	4.8570-02
1.6015+06	1.6868-06	2.8026+00	4.5601-02
1.2700+06	2.9732-06	3,8363+00	4.0327-02
9.3070+05	4.5147-06	4.2292+00	2.7844-02
6.47:9+05	7.7350-06	5,0068+00	2.6530-02
4.7399+05	1.1319-05	5.3651+00	2.8550-02
3.0200+05	1.4975-05	5.4218+00	3.2062-02
2.7125+05	1.7753-05	4.8155+00	3.0073-02
2.0029+05	2.2311-05	4.4687+00	3.6768-02
1.5394+05	2.9050-05	4.4720+00	4.2808-02
1.1794+05	3.7628-05	4.4376+00	4.1079-02
9.0133+04	4.2764-05	3.8544+00	5.2016-02
6.9295+04	4.5003-05	3.1185+00	5,2668-02
5.35, 0+04	4.9431-05	2.6455+00	6.5724-02
4.17:8+04	4.2422-05	1.7702+00	7.3841-02
3.2775+04	4.9112-05	1.6096+00	1.0758-01
2.6004+04	9.9692-06	2.5924-01	1.0827-01
2.0504+04	7.4593-05	1.5295+00	8.2423-02

TABLE ESO COMPUTED 6-25-69

NUN T17 3-17-00

INTERMEDIATE NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 53 DEGREES USING A DETECTOR COMPUSED OF A BAC-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

INE	UTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	2.2195+06	1.5728-06	3,4905+00	6.2932-02
	1.6015+06	2.4709-06	4,1054+00	5.7977-02
	1.29:3+06	4.0822-06	5.2673+00	5.2963-02
	9.3076+05	6.8812-06	6.4460+00	3.4690-02
	6.4729+05	1.1514-05	7.4529+00	3.3352-02
	4.7399+05	1.9868-05	9.4172+00	3.3150-02
	3.6210+35	2.6787-05	9.6985+00	3,6911-02
	2.71:5+05	3.3897-05	9,1946+00	3.3567-02
	2.0029+05	3.8654-05	7.7420+00	4.3093-02
	1.5394+05	5.2002-05	8.0976+00	4.8832-02
	1.1794+05	6.3271-05	7.4622+00	4.8912-02
	9.0133+04	8.6425-05	7,7897+00	5.6705-02
	6.9295+04	8.5820-05	5.9469+00	5.9080-02
	5.33,0+04	9.9784-05	5.3404+00	7.0273-02
	4.1720+04	1.6165-04	6.7453+00	5.8926-02
	3.2775+04	1.8016-04	5,9047+00	7.6923-02
	2.0004+04	2.5041-04	6.6677+00	7.8934-02
	2.0504+04	2.6490-04	5.4315+00	6.6667-02
	1.6145+04	3.0067-04	4.8537+00	7.9682-02
	1.2732+04	3.5760-04	4.5530+00	7,7152-02
	1.0033+04	5.4921-04	5,5377+00	7.6584-02

RUN 120 3-18-06

INTERMEDIATE NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 37 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESULUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEOTRON ENERGY NEOTRON FLOX FLOXFEIERGY RELATIV	/ C.
(EV) (RELATIVE UNITS) (RELATIVE UNITS) UNCERTAI	INTY
2.2193+06 1.3988-06 3.1044+00 5.8026-0	202
1.0015+06 1.9028-06 3.1615+00 5.7448-0	02
1.2903+00 2.4501-06 3.1614+00 5.9444-0	20
9.3076+05 3.1504-96 2.9512+00 4.4108-0	20
6.4729+05 3.7396-06 2.4206+00 5.0637-0	20
4.7399+05 4.6842-06 2.2203+00 5.9028-0	20
3.62(0+05 5.1583-06 1.8676+00 7.2932-0	20
2.7125+05 5.2125-06 1.4139+00 7.4744-0	20
2.0029+05 6.0379-06 1.2093+00 9.2450-0	20
1.5394+05 5.7820-06 8.9008-01 1.2804-0	1
1.1794+05 6.6435-06 7.8353-01 1.3245-0	1
9.0133+04 1.1554-05 1.0414+00 1.3131-0	01

RUN T21 3-10-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 37 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR LHERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL EPROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1,4952-06	3.3183+00	5.3683-02
1.60:5+06	2.1660-06	3.5988+00	5.1503-02
1.29:13+06	2.8912-06	3.7305+00	5.2342-02
9.3076+05	4.2230-06	3,9559+00	3.6637-02
6.47:9+05	5.3629-06	3.4714+00	4.0656-02
4.7399+05	6.9532-06	3,2957+00	4.6274-02
3.6260+05	7.3386-06	2.6570+00	5.8722-02
2.7125+05	9,1592-06	2.4844+00	5.3916-02
2.0029+05	1.2215-05	2.4465+00	6.4820-02
1.5394+05	1.1179-05	1,7209+00	8.5749-02
1.1794+05	8.7418-06	1.0310+00	1.1111-01

RUN 122 3-10-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 37 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	2.6225-06	5,8201+00	4.0791-02
1.0015+06	3.6953-06	6.1397+00	3,9684-02
1.29-3+06	5.0556-06	6.5232+00	3.9841-02
9.3070+05	6.0513-06	7.5421+00	2.6688-02
6.47.9+05	1.1046-05	7.1500+00	2.8421-02
4.7399+05	1.6097-05	7.6298+00	3.0817-02
3.6276+05	1.9165-05	6,9389+00	3.6564-02
2.71:5+05	2.0490-05	5.5579+00	3.6084-02
2.0029+05	2.6767-05	5,3612+00	4.3193-02
1.5394+05	3.3323-05	5,1297+00	5.1164-02
1.1794+05	3.6616-05	4.3185+00	5.4074-02
7.9714+04	4.0494-05	3.2279+00	7.2212-02
4.2074+04	2.5263-05	1.0781+00	1.4942-01
2.06.4+04	7.2430-05	1.5126+00	1.2632-01
1.1408+04	1.2376-04	1.4119+00	1.3079-01
6.4860+03	3.2284-04	2.0939+00	1.2403-01
3.2449+03	4.1225-04	1.3394+00	1.4592-01
1.40,0+03	1.0360-04	1.5424-01	1.0311-01

## RUN T23 3-18-66

## TABLE E54

## COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 37 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2,2193+06	3,3158-06	7.3588+00	3.4155-02
1.6615+06	4.8715-06	8.0940+00	3.2544-02
1.2903+06	7.7933-06	1.0056+01	3.0217-02
9.3676+05	1.2462-05	1.1674+01	2.0289-02
6.4729+05	1.9706-05	1.2755+01	2.0111-02
4.7399+05	2.9217-05	1.3849+01	2.1595-02
3.6200+05	3.6859-05	1.3345+01	2.4842-02
2.7125+05	3.9148-05	1.0619+01	2.4689-02
2.0029+05	4.7760-05	9.5659+00	3.0350-02
1.5394+05	5.3260-05	8,1988+00	3.8473-02
1.1794+05	5.6141-05	6,6213+00	4.0696-02
9.0133+04	7.2778-05	6.5597+00	4.8633-02
6.9295+04	7.3637-05	5,1027+00	5.0125-02
5.3520+04	7.8903-05	4.2229+00	6.1898-02
4.1728+04	7,6689-05	3,2001+00	6.6637-02
3.2775+04	1.1953-04	3.9176+00	7.3681-02
2.6004+04	7.9942-05	2.0788+00	1.1918-01
2.0504+04	1.164/-04	2.3881+00	8.1759-02
1.3592+04	2.2542-05	3,0639-01	1.8634-01

RUN T24 3-19-66

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 37 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	3.7510-06	8.3246+00	3.0236-02
1.6615+06	5.8636-06	9.7424+00	2.7931-02
1.2903+06	8.6057-06	1.1104+01	2.7079-02
9.3676+05	1.5113-05	1.4157+01	1.7351-02
6.4729+05	2.6723-05	1.7298+01	1.6294-02
4.7399+05	4.3041-05	2.0401+01	1.6804-02
3.6206+05	5.3733-05	1.9455+01	1.9424-02
2.7125+05	6.3274-05	1.7163+01	1.8317-02
2.0029+05	7.5816-05	1.5185+01	2.2843-02
1.5394+05	9.2041-05	1.4169+01	2.7437-02
1.1794+05	9.5349-05	1.1245+01	2.9460-02
9.0133+04	1.1785-04	1.0622+01	3.5847-02
6.9295+04	1.0842-04	7.5130+00	3.9193-02
5.3520+04	1.4971-04	8.0125+00	4.2524-02
4.1728+04	1.3448-04	5,6116+00	4.7522-02
3.2775+04	2.1051-04	6.8995+00	5.3164-02
2.6004+04	2.3146-04	6.0189+00	6.6285-02
2.0504+04	2.7273-04	5.5921+00	4.9975-02
1.6143+04	3.215/-04	5.1911+00	5.7620-02
1.2732+04	2.7559-04	3.5088+00	6.6082-02
1.0083+04	4.1143-04	4.1484+00	6.2403-02
8.0292+03	8.9448-04	7.1820+00	6.1476-02
6.3844+03	6.6970-04	4,2756+00	7.8039-02
5.0443+03	9.2974-04	4.6899+00	6.4712-02
4.0034+03	6,5301-04	2.6143+00	8.7171-02
3.1979+03	6.7609-04	2.1621+00	8.9016-02
2.5453+03	1,2145-03	3.0913+00	7.3403-02
2.0248+03	1.5519-03	3.1423+00	7.4702-02
1.6149+03	3,3079-03	5.3419+00	5.7774-02
1.2884+03	3.7758-03	4.8647+00	6.3500-02
1.0274+03	5.0866-03	5.2260+00	6.1176-02

5 AT INTERNAL SEQUENCE NUMBER 0103, LOCATION 014000

KUN 125 3-19-06

COMPUTED 6-26-69

INTERMEDIATE NEUTRON SPECTRUM FOR EMPTY LH2 DEWAR AT 37 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E <sub>1</sub> )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	4,1621-06	9,2369+00	4.8031-02
1.0015+06	5.6333-06	9.3597+00	4.7652-02
1.2903+00	9.8531-06	1,2713+01	4.1844-02
9.3070+05	1.6989-05	1.5915+01	2.6923-02
6.4729+05	3.0533-05	1,9764+01	2.4919-02
4.7399+05	4.7619-05	2.2571+01	2.6161-02
3.6200+05	5.4326-05	1.9669+01	3.2146-02
2.7125+05	6.2574-05	1.6973+01	3.1217-02
2.00,9+05	7.6451-05	1.5312+01	3.9984-02
1.5394+05	9.2589-05	1.4253+01	5.0364-02
1.1794+05	9.0770-05	1.0705+01	6.0165-02
9.0133+04	9.2417-05	8.3298+00	8.9214-02
6.9295+04	8.8992-05	6,1667+00	1.0779-01
5.3520+04	9.3607-05	5.0098+00	1.6213-01

TABLE E57

RUN 127 3-19-66

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 13.0 INCHES LH2 AT 15 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

N	LUTKON LNERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	2.2193+06	3.3517-06	7,4364+00	4.1024-02
	1.6015+06	3.6272-06	6.0266+00	4.5539-02
	1.2903+06	4.0365-06	5,2083+00	5.0689-02
	9.3676+05	4.9664-06	4.6523+00	3.8738-02
	6.4729+05	4.9986-06	3,2355+00	4.7815-02
	4.7399+05	5.3978-06	2.5585+00	5.9613-02
	3.6206+05	6.4674-06	2.3416+00	7.1175-02
	2.7125+05	6.2803-06	1,7035+00	7.4003-02
	2.0029+05	6,4152-06	1,2849+00	9.7312-02
	1.5394+05	3.4941-06	5.3788-01	1.7514-01
	1.1794+05	5.1171-06	6.0351-01	1.6485-01
	8.0027+04	1.0975-05	8.8488-01	1.3063-01
	5.2539+04	7.9295-06	4,1661-01	1.9389-01

TABLE E58

COMPUTED 6-25-69

PUN 128 3-19-06

INTERMEDIATE NEUTRON SPECTRUM FOR 10,5-INCHES LH2 AT 15 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

.VE	UTRON LNERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
	2.2193+06	5.2632-06	1.1681+01	3,1114-02
	1.6015+06	6.0893-06	1.0117+01	3.3408-02
	1.2903+06	7.6946-06	9,9283+00	3.4900-02
	9.3670+05	9.2142-06	8.6315+00	2.6958-02
	6.4729+05	1.0079-05	6.5240+00	3.2075-02
	4.7399+05	1.2944-05	6.1353+00	3.7242-02
	3.6206+05	1.2655-05	4.5819+00	4.8622-02
	2.7125+05	1.2750-05	3.4584+00	4.9447-02
	2.0029+05	1.3436-05	3.6925+00	5,5815-02
	1.5394+05	2,1255-05	3.2720+00	6.9505-02
	1.1794+05	1.3737-05	2.2098+00	8.0064-02
	9.0133+04	2.7645-05	2.4917+00	9.2848-02
	6.9295+04	1.5795-05	1.0945+00	1.2309-01
	5.3520+04	2.5811-05	1.3814+00	1.2039-01

RUN T29 3-19-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 15 DEGREES USING A DETECTOR COMPOSED OF A 84C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERUY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	7.6968-06	1,7082+01	2.4508-02
1.6615+06	9.5726-06	1.5905+01	2.5384-02
1.2903+00	1.3299-05	1.7160+01	2.5295-02
9.3076+05	1.7061-05	1.5982+01	1.8869-02
6.4729+05	2.2325-05	1.4451+01	2.0616-02
4.7399+05	2.7470-05	1.3021+01	2.4163-02
3.6206+05	2.7507-05	9,9592+00	3.1469-02
2.7125+05	3.0076-05	8.1581+00	3.0836-02
2.0029+05	3.4075-05	6.8249+00	3.9445-02
1.5394+05	4.2837-05	6.5943+00	4.6845-02
1.1794+05	5.2758-05	6.2223+00	4.6245-02
9.0133+04	5.5025-05	4.9596+00	6.0237-02
6.9295+04	6.1278-05	4.2463+00	6.0467-02
5.3520+04	9.0832-05	4.8613+00	6.3823-02
4.1720+04	1.0734-04	4.4791+00	6.1039-02
3.2775+04	1.3298-04	4.3584+00	7.4247-02
2.6004+04	2.7400-04	7.1251+00	7.2112-02
2.05:4+04	2.0008-04	4.1024+00	6.6785-02
1.6143+04	2.9084-04	4.6950+00	6.9656-02
1.2732+04	3.2258-04	4.1071+00	6,9505-02
1.0023+04	4.5461-04	4.5838+00	7.0552-02
TABLE E60

COMPUTED 6-25-69

KUN T30 3-19-66

INTERMEDIATE NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 15 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.0042-05	2.2286+01	2.0879-02
1.6015+06	1.3651-05	2.2681+01	2.0690-02
1.2903+06	1.3198-05	2.3481+01	2.1054-02
9.3576+05	2.6722-05	2.5032+01	1.4728-02
6.4729+05	3.9390-05	2,5497+01	1.5139-02
4.7399+05	5.1753-05	2.4530+01	1.7231-02
3.6200+05	5.4321-05	1.9667+01	2,1811-02
2.7125+05	5.3511-05	1,4515+01	2.2445-02
2.0029+05	5.9721-05	1,1962+01	2.9148-02
1.5394+05	6.6722-05	1.0271+01	3.6418-02
1.1794+05	7.9209-05	9.3419+00	3.6418-02
9.0133+04	8.8581-05	7,9931+00	4.6474-02
6.9293+04	1.0078-04	6.9830+00	4,5980-02
5.3520+04	1.1609-04	6.2131+00	5.4074-02
4.1720+04	1.2484-04	5.2093÷00	5,5902-02
3.2775+04	2.0426-04	6.6946+00	6,3628-02
2.6004+04	4.2270-04	1.0992+01	6.0746-02
2.0004+04	2.5787-04	5.2874+00	5,8222-02
1.6143+04	3.4236-04	5.5267+00	6.2137-02
1.2732+04	4.3640-04	5.5562+00	5.8621-02
1.0033+04	3.4141-04	3.4424+00	7.6923-02
8.0292+03	8.3904-04	6,7368+00	7.1247-02

KUN 131 3-19-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 15 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

WEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ē,)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.1944-05	2.6507+01	1.9178-02
1.6015+00	1.7263-05	2.8682+01	1.8437-02
1.2903+06	2,4867-05	3,2086+01	1.8055-02
9.3676+05	3,9231-05	3.6750+01	1.2222-02
6.4729+05	6.3238-05	4.0933+01	1.1997-02
4.7399+05	8.6830-05	4.1157+01	1.3413-02
3.6206+05	9,4568-05	3.4239+01	1.6642-02
2.7125+05	9.9767-05	2.7062+01	1.6590-02
2.0029+05	1.1147-04	2.2326+01	2.1313-02
1.5394+05	1.2374-04	1.9049+01	2.7013-02
1.1794+05	1.2640-04	1.4908-01	2,9283-02
9.0133+04	1.5289-04	1.3780+01	3.5529-02
6.9295+04	1.4622-04	1.0132+01	3.8462-02
5.3520+04	2.2105-04	1.1831+01	3,9778-02
4.1728+04	2.1294-04	8.8856+00	4.3121-02
3.2775+04	3.4614-04	1.1345+01	4.7576-02
2.6004+04	3.9697-04	1.0323+01	5.8763-02
2.0504+04	4.3541-04	8,9276+00	4.4614-02
1.0143+04	5.7987-04	9.3608+00	4.8269-02
1.27:2+04	6.5459-04	8.3342+00	4.8679-02
1.0053+04	8.1467-04	8.2143+00	5.0912-02

RUN T32 3-19-06

COMPUTED 6-26-69

INTERMEDIATE NEUTRON SPECTRUM FOR EMPTY LH2 DEWAR AT 15 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

HEUTRON ENERGY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(ビバ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.6361-05	3,6310+01	2.7542-02
1.6615+06	2.6492-05	4.4016+01	2,4903-02
1.2903+06	4.1581-05	5,3652+01	2.3312-02
9.3076+05	7.1919-05	6.7371+01	1.5079-02
6.4729+05	1.2516-04	8.1015+01	1.4245-02
4.7399+05	1.8072-04	8.5659+01	1.5602-02
3.6206+05	2.0780-04	7.5236+01	1,9035-02
2.7125+05	2.1973-04	5,9602+01	1.9248-02
2.0029+05	2.4182-04	4.8434+01	2.5823-02
1.5394+05	2.6231-04	4.0380+01	3.4448-02
1.1794+05	2,5482-04	3,0053+01	4.0775-02
9.0133+04	2.2340-04	2.0136+01	6.8904-02
6.9295+04	1.7205-04	1.1922+01	9.9403-02
5.3520+04	1.7082-04	9.1423+00	1.5523-01

RUN T34 3-19-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 5 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.1224-05	2.4909+01	2.0562-02
1.6615+06	9.0043-06	1.4961+01	2.6525-02
1.2903+06	9.1241-06	1.1773+01	3.0945-02
9.3676+05	8.0994-06	7.5872+00	2.7633-02
6.4729+05	7.5152-06	4.8645+00	3.5861-02
4.7399+05	7.8788-06	3.7345+00	4.6048-02
3.6206+05	8.334-06	3,1982+00	5.6024-02
2.7125+05	9.6784-06	2,6253+00	5.5140-02
2.0029+05	8.3118-06	1.6648+00	8.0090-02

TABLE E64

COMPUTED 6-25-69

RUH T37 3-19-06

INTERMEDIATE NEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 5 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	1.7968-05	3,9876+01	1.6054-02
1.6615+06	1.5375-05	2.5546+01	2.0064-02
1.2903+00	1.5195-05	1.9606+01	2.3714-02
9.3676+05	1.4902-05	1.3960+01	2.0152-02
6.4729+45	1.4496-05	9.3831+00	2.5554-02
4.7399+05	1.5581-05	7.3852+00	3.2168-02
3.6206+05	1.5991-05	5.7897+00	4.1225-02
2.7125+05	1.4420-05	3.9114+00	4.4473-02
2.0029+05	1.6578-05	3.3204+00	5.6469-02
1.5394+05	2.0715-05	3.1889+00	6.7481-02
1.1794+05	1.3780-05	2.2149+00	7.7429-02
9.0133+04	2.1198-05	1.9106+00	9,5001-02
6.9295+04	3.3913-05	2.3500+00	8.1379-02
5.2329+04	1.7742-05	9.2842-01	1.5394-01
3.9998+04	1.6186-05	6.4741-01	1.7100-01
3.1022+04	5.1468-05	1,5966+00	1.4374-01
2.4332+04	3.0573-05	7.4390-01	1.7303-01

COMPUTED 6-25-69

KUN 138 3-19-66

INTERMEDIATE NEUTRON SPECIRUM FOR 7.0-INCHES LH2 AT 5 DEGREES USING A DETECTOR COMPOSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROP = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(モッ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	3,7881-05	8.4069+01	1.1386-02
1.6015+06	3.5364-05	6.0751+01	1.3419-02
1.2903+06	3.6872-05	4.7576+01	1.5721-02
9.3070+05	4.0105-05	3.7569+01	1.2721-02
6.4729+05	4.1194-05	2.6664+01	1,5715-02
4.7399+05	4.3075-05	2.0417+01	1.9960-02
3.5206+05	4.0324-05	1.4600+01	2.6958-02
2.7125+05	3,8531-05	1.0452+01	2,3061-02
2.0029+05	4.0178-05	8.0473+00	3.7987-02
1.5394+05	4.2063-05	6.4752+00	4.8679-02
1.1794+05	5.3738-05	6.3379+00	4.7673-02
9.0133+04	7.5391-05	6.8403+00	5.3452-02
6.9295+04	6.5402-05	4.6013+00	6.0302-02
5.3520+04	1.1890-04	6.3635+00	5.7073-02
4.1723+04	1.2337-04	5.1480+00	5,9976-02
3.2775+04	1.0778-04	5.4990+00	7.0888-02
2.60.4+04	3.3027-04	9.8885+00	6.5233-02
2.0000++04	2.3259-04	4.7690+00	6.3888-02
1.61+3+04	3.3438-04	5.3979+00	6.7884-02
1.2732+04	4.0086-04	6.1223+00	5,9976-02
1.0033+04	6,0766-04	5.1270+00	6.5653-02

KUN 139 3-20-06

COMPUTED 6-25-69

INTERMEDIATE NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 5 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ELLERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E.)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	6.3007-05	1.3983+02	1.0180-02
1.0015+00	6.3979-05	1.0630+02	1.1720-02
1.2903+06	7.1352-05	9,2065+01	1.3077-02
9.3070+05	8.2228-05	7,7028+01	1.0331-02
6.4729+05	9.6217-05	6,2280+01	1.1951-02
4.7399+05	1.0369-04	4.9148+01	1.5033-02
3.6206+05	9.0210-05	3,2661+01	2,1011-02
2.7125+05	8.2735-05	2,2442+01	2.2345-02
2.0029+05	8.6166-05	1.7258+01	3.0072-02
1.5394+05	8.8236-05	1.3583+01	3.9442-02
1.1794+05	9.8555-05	1.1624+01	4.0879-02
9.0133+04	1,2775-04	1.1514+01	4.8314-02
6.9295+04	1.2750-04	8.8351+00	5,1031-02
5.3520+04	1.3208-04	9.7449+00	5.4636-02
4.17:0+04	2.1661-04	9.0387+00	5,2228-02
3.2775+04	3.3961-04	1.1131+01	6.3423-02
2.6004+04	5.4201-04	1.4094+01	6.5067-02
2.03(4+04	4.5213-04	9.2705+00	5.4011-02
1.6143+04	7.7213-04	1.2464+01	5.2030-02
1.2732+04	6.1396-04	7.8169+00	6.1663-02
1.00:3+04	9.3353-04	9.4128+00	5.9804-02
8.0292+03	1.7790-03	1.4284+01	6.1757-02
6.3844+03	2.3274-03	1.4859+01	5.7890-02
5.0445+03	2.2053-03	1.1124+01	5.8461-02
4.0034+03	2.9155-03	1.1672+01	5,7054-02
3.1979+03	3.0388-03	9.7178+00	6.0366-02
2.5453+03	4.6162-03	1.1750+01	5.3513-02

TABLE E67

COMPUTED 6-25-69

NUN 140 3-20-66

INTERMEDIATE NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 5 DEGREES USING A DETECTOR COMPUSED OF A D4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESOLUTION FOR GROUPING = .20 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2193+06	6,6963-05	1.4861+02	1.3061-02
1.6015+06	8.1334-05	1.3514+02	1.3707-02
1.2903+06	1.0398-04	1.3417+02	1.4254-02
9.3676+05	1.3856-04	1.2980+02	1.0475-02
6.4729+05	1.8546-04	1.2005+02	1,1305-02
4.7399+05	2.2414-04	1.0624+02	1.3389-02
3.6206+05	2.1746-04	7.8734+01	1.7707-02
2.7125+05	2.0569-04	5.5793+01	1.8544-02
2.0029+05	2.0227-04	4.0513+01	2,5598-02
1.5394+05	2.0782-04	3,1992+01	3.3518-02
1.1794+05	2.0542-04	2.4227+01	3.6840-02
9.0133+04	2.4770-04	2.2326+01	4.5464-02
6.9295+04	2.1242-04	1.4720+01	5.1400-02
5,3520+04	4.0307-04	2.1572+01	4.7864-02
4.1728+04	3.8274-04	1.5971+01	5.2044-02
3.2775+04	6.0674-04	1.9886+01	5,6870-02
2.6004+04	1.0092-03	2.6243+01	5,9245-02
2.0504+04	7.5372-04	1.5454+01	5.3255-02
1.6143+04	9.9425-04	1,6050+01	5,9623-02
1.2732+04	1.2670-03	1.6131+01	5,6344-02
1.00a3+04	1.8698-03	1.8853+01	5.4579-02
8.0292+03	2.7694-03	2,2236+01	6.3577-02
6.3044+03	3.6300-03	2.3175+01	6,0623-02
5.0443+03	4.6895-03	2,3655+01	5.2472-02
4.0034+03	5.1318-03	2,0545+01	5.6174-02

## NUN T41 3-20-06

## TABLE E68

COMPUTED 6-26-69

INTERMEDIATE NEUTRON SPECTRUM FOR EMPTY LH2 DEWAR AT 5 DEGREES USING A DETECTOR COMPUSED OF A B4C-PARAFFIN DISK AND A NE-226 SCINTILLATOR ENERGY RESILUTION FOR GROUPING = .20 LARGEST FRACTIONAL EPROP = .20

NEUTRON ENERGY	HEUTRON FLUX	FLUX*ENERGY	RELATIVE
(L.)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2143+06	1,3428-04	2.9801+02	1.4053-02
1.6615+06	1.8080-04	3.0040+02	1.4035-02
1.29.3+00	2.5300-04	3,2652+02	1.3979-02
9.3070+05	3.9741-04	3,7228+02	9,5155-03
0.4127+05	6.0293-04	3,9027+02	9.6834-03
4.7399+05	8.1769-04	3.8758+02	1.0986-02
3.6206+05	8.3149-04	3,1915+02	1.3931-02
2.71; 5+05	5.9388-04	2.4382+02	1.4408-02
2.00, 9+05	9.4354-04	1.8998+02	1.9874-02
1.5394+05	9.9949-04	1.5386+02	2.7165-02
1.1794+05	9.5729-04	1.1290+02	3.3093-02
9.01:3+04	7.3398-04	6.6156+01	6.4617-02
6.9295+04	4.9731-04	3.4461+01	1.2473-01

RUNS R1.K2 2-20-66

1-3-69

THERMAL NEWTRON SPECTRUM FOR 13.0-INCHES LH2 AT 37 DEGREES USING A BADK OF 32 VERTICAL BF3 DETECTORS ENERGY RESCLUTION FOR GROUPING = .10 LARGEST FRACTIONAL ERROR = .20

HEUTRON ENLROY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(E)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.9551+01	4.7553-03	1,4052-01	1.3600-01
5.8504+00	1.3709-02	8.0333-02	1.7443-01
2.8905+00	3 1325-02	9.0736-02	1 9061-01
2.0500+00	4 9775-02	1 0250-01	1 5751-01
1 54 7400	7 6603-02	1 1797-01	1 3001-01
1.3337+00	7 3554-02	9 7757-02	1 6804-01
1.1931+00	0 5399-02	0.0931-02	1 4804-01
9.5215-01	9.3398-02	9.0031-02	1.4094-01
1.1139-01	1.1209-01	8.7138-02	1.0303-01
6.4072-01	1.0464-01	6.7673-02	1.9525-01
5.4641-01	1.4403-01	7.8699-02	1.5435-01
4.6770-01	2.2330-01	1.0445-01	1.2014-01
4.0493-01	2.3114-01	9.3596-02	1.2219-01
3.5398-01	2.6520-01	9.3875-02	1.1708-01
3.1207-01	2.1440-01	6.6908-02	1.6028-01
2.7/19-01	3.2424-01	8.9876-02	1.1516-01
2.47.4-01	4.2481-01	1.0528-01	1.0243-01
2.1224-01	4.5307-01	9.6160-02	7.6070-02
1.7515-01	4.1989-01	7.3544-02	9.5818-02
1.4699-01	4,8165-01	7.0798-02	1.0239-01
1.2512-01	6.4352-01	8.0517-02	8.0728-02
1.0778-01	7.0864-01	7.6377-02	8.1716-02
9.3022-02	7.0057-01	6.5729-02	9.8002-02
8.2400-02	7.1742-01	5,9120-02	1.0633-01
7.2953-02	7.9420-01	5.7939-02	1.0703-01
6.5038-02	1.0528+00	6.8472-02	9.3503-02
5.6879-02	1.2339+00	7.0183-02	7.4452-02
4.8918-02	1.7586+00	8.6027-02	5.7204-02
4.2019-02	2.2702+00	9.6527-02	5.4755-02
3.73.13-02	2.5358+00	9.4593-02	5.6139-02
3.2995-02	3.7132+00	1.2252-01	4.8426-02
2.9395-02	4.3185+00	1.4163-01	4.6064-02
2.5914-42	7.6048+00	1,9707-01	3.2659-02
2.2037-02	1.0913+01	2.4704-01	3.0291-02
1.9954-02	1.7809+01	3,5536-01	2.5180-02
1.7730-02	3.8062+01	6.7484-01	1.8502-02
1.5868=02	7.0934+01	1,1256+00	1.4481-02
1.4112-112	1,1826+02	1.6689+00	1.0849-02
1.2471-02	1,9696+02	2,4563+00	9.1535-03
1.1004-02	2.4878+02	2.7600+00	8.8861-03
9.92	2.7186+02	2.6990+00	9.2182-03
8.8451-03	2-8610+02	2.5306+00	8.9723-03
7.8507-03	3.0856+02	2,4224+00	9.5071-03
7.0101-03	3,1800+02	2.2305+00	1.0202-02

ENERGY RESULUTION FOR GROUPING = .10 LARGEST FRACTIONAL ERROR = .20 FLUX\*ENERGY RELATIVE HEUTRON LHERGY NEUTRON FLUX (RELATIVE UNITS) (RELATIVE UNITS) UNCERTAINTY (E.) 9.6608-03 6.2510-63 3.2484+02 2.0306+00 9.8819-03 5.55.7-03 1.8269+00 3.2866+02 1.7454+00 4.9750-03 1.0235-02 3.5084+02 8.7702-03 4.4471-03 3.8907+02 1.7302+00 9.2258-03 3.9700-03 3.8801+02 1.5407+00 9.3653-03 3.5442-03 4.0821+02 1.4468+00 4.1571+02 1.3148+00 1.0193-02 3.1627-03 1.0575-02 2.0235-03 4.2217+02 1.1920+00 1.1551-02 2.5218-03 4.3102+02 1.0869+00 1.1949-02 4.4582+02 1.0051+00 2.2505-03 1.3135-02 4.4935+02 9.0643-01 2.0172-03 1.3859-02 1.8073-03 4.4957+02 8.1251-01 7.0719-01 1.5675-02 1.6211-03 4.3624+02 1.4564-03 4.2790+02 6.2331-01 1.6750-02 4.0071+02 5.2297-01 1.8063-02 1.3051-03 3.8767+02 4.5257-01 1.9316-02 1.1674-03 3.0321+02 3.8014-01 2.1951-02 1.0460-03 9.4057-04 3.3754+02 3.1748-01 2.4192-02 2.6892-02 3.0924+02 2.6114-01 3.4440-04 2.9996-02 7.5770-04 2.9058+02 2.1262-01 3.3606-02 6.79.8-04 2.5011+02 1.7004-01

1.3472-01

2.2081+02

RUNS K1, R2 2-20-60 CONTINUED

6.1010-04

THERMAL NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BE3 DETECTORS

TABLE E69

4-3-69

3.7429-02

KUNS K4, K3 2-20-60

4-3-69

THERMAL DELITRON SPECTRUM FOR 10.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BES DETECTORS ENERGY RESOLUTION FOR GROUPING = .10 LARCEST FRACTIONAL EPROR = .20

LEUTRON ENERGY	GEUTRON FLUX	FLUX*ENERGY	PELATIVE
(に、)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2,9061401	4.9491-03	1 4625-01	1 3992-01
7 4472+01	1 765-02	1 2979-01	1 6337-01
1.372400	2 6377-02	1 1621-01	1 7447-01
4.3725400	5.0471-02	1,1621-01	1 3171-01
2.39.0+00	6 4930-02	1 3370 01	1.3701-01
2.0092+00	1 0471-01	1.6112-01	1.0777-01
1 1931-00	1. 453-01	1.7/100-01	0.6706-02
1.1921+00	1.4055-01	1 5320-01	1 0735-01
9.5215-01	2.0520-01	1.6030-01	0.7757-02
7.7739-01	2.0620-01	1.0030-01	9,7757-02
6.40/2-01	2.2038-01	1.4640-01	1.0352-01
5.4011-01	2.3196-01	1.2075-01	1,1309-01
4.0770-01	2.5657-01	1.2409-01	1.1586-01
4.04.3-01	3.6339-01	1.4/15-01	9.1560-02
3.5398-01	5.1094-01	1.1007-01	1.103/-01
3.12.7-01	4.1755-01	1.3030-01	1.0364-01
2.7719-01	4.1037-01	1.1375-01	1.0787-01
2.4/44-01	4.8027-01	1.1903-01	9.9124-02
2.1224-01	5.5919-01	1.3991-01	6.4350-02
1.7515-01	7.5432-01	1.3212-01	6.6/41-02
1.4699-01	8.8692-01	1.3037-01	6.7764-02
1.2012-01	1.1014+00	1.3781-01	6,0815-02
1.0/73-01	8.4685-01	9.1273-02	8.0226-02
9.3822-02	1.0185+00	9.5558-02	7.6555-02
8.2400-02	1.2405+00	1.0222-01	7.4797-02
1.53-2-05	1.0919+00	7.9657-02	9.4511-02
6.5030-02	1.2922+00	8.4042-02	9.1225-02
5.0079-12	1.2/13+00	1.0644-01	5.9024-02
4.8910-02	2.0207+00	9.8849-02	6.5804-02
4.2019-02	2.7149+00	1.1543-01	5.8603-02
3.7303-02	3.009+00	1.4178-01	5.2277-02
3.2995-02	5.3803+00	1.7752-01	4.6125-02
2.9395-02	6.7892+00	1.9957-01	4.3349-02
2.5914-02	1.0321+01	2,6746-01	3.2899-02
2.2037-02	1.6675+01	3.7747-01	2.7805-02
1.9954-02	2.7987+01	5,5845-01	2.3564-02
1.7730-02	5.6590+01	1.0033+00	1.7710-02
1.58-0-02	1.0255+02	1.6273+00	1.4171-02
1.4112-02	1.6556+02	2.3364+00	1.0820-02
1.2471-02	2.7083+02	3.3775+00	9.2717-03
1.1094-02	3.3374+02	3.7025+00	9.0805-03
9.9278-03	3.6583+02	3.6319+00	9.4246-03
8.04:1-03	3.8952+02	3,4453+00	9.1170-03
7.8367-03	4.1020+02	3.2675+00	9.6878-03

CONTINULU			
THERMAL NEUTRON SEA BANK OF 12 VERT	PECTRUM FOR 10.5-1 ICAL BE3 DETECTORS	NCHES LH2 AT 37 D	EGREES USING
ENERGY RESOLUTION	FOR GROUPING = .1	U LARGEST FRAC	TIONAL ERROR = .20
NEUTRON ENERGY	NEUTRON FLUX		RELATIVE
(E))	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
7.0141-03	4.1461+02	2,9081+00	1.0564-02
6.2010-03	4.45/5+02	2.7864+00	9.7751-05
5.5587-03	4.5050402	2.3570+00	1 0426-02
4.4471-03	5.3435+02	2.3763+00	8.8599-03
3.9708-03	5.2537+02	2.0861+00	9.4177-03
3.5442=03	5.5303+02	1,9600+00	9.5039-03
3.1027-03	5.6781+02	1.7956+00	1.0336-02
2.8235-03	5,9158+02	1.6703+00	1.0574-02
2.5210-03	6.0835+02	1.5341+00	1.1489-02
2.2545-03	6,2379+02	1.4063+00	1.1957-02
2.0172-03	6.2507+02	1.2609+00	1.3170-02
1.8073-03	6.2181+02	1.1238+00	1.3953-02
1.6211-03	6,2595+02	1.0147+00	1.5355-02
1.4564-03	5.9131+02	8.6118-01	1.6749-02
1.3051-03	5.5895+02	7.2949-01	1.8161-02
1.1074-03	5.3922+02	6.2949-01	1.9582-02
1.0465-03	5.0891+02	5.3263-01	2.2227-02
9.4057-04	4.7584+02	4.4756-01	2.4506-02
8.4446-04	4.3994+02	3.7151-01	2.7253-02
7.5778-04	4.0051+02	3.0350-01	3.0572-02
6.7988-04	3.6096+02	2.4541-01	3.4460-02
6.1010-04	3,2025+02	1,9538-01	3.8976-02

TABLE ETO

RUNS 14+13 2-20-60

4-3-69

RUNS K5, RO 2-20-65

0-3-69

THERMAL NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL RF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(ビッ)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
4.4266+01	5.6329-03	2.4935-01	1.0590-01
1.4836+01	1.3468-02	2.7399-01	1.0947-01
7.3472+00	3.8537-02	2.8314-01	1,0472-01
4.3725+00	7.0919-02	3.4507-01	8.0759-02
2.8960+00	1.1774-01	3.4105-01	7.8016-02
2.0592+00	1.4215-01	2,9272-01	8.6454-02
1.5357+00	2.0884-01	3.2134-01	7.7697-02
1.1931+00	2.8751-01	3.4303-01	7.0601-02
9.5213-01	3,2361-01	3.0812-01	7.4908-02
7.7739-01	3.3815-01	3.0174-01	7.4797-02
6.4072-01	4.9433-01	3.1969-01	7.0158-02
5.4041-01	5.8042-01	3.1715-01	6.8216-02
4.6776-01	6.5308-01	3.0548-01	6.7009-02
4.0493-01	5.9703-01	2.4176-01	8.4235-02
3.5398-01	6.9457-01	2.4586-01	7.7856-02
3.1207-01	8.1578-01	2.5458-01	7.2923-02
2.7719-01	9.0263-01	2.5020-01	7.5615-02
2.4784-01	1.2098+00	2.9984-01	6.7418-02
2.2292-01	1.3479+00	3.0047-01	6.4343-02
2.0157-01	1.5185+00	3.0608-01	6.5500-02
1.8315-01	1.7342+00	3,2678-01	6.0897-02
1.6715-01	1.6784+00	2.8054-01	6.7976-02
1.5315-01	1.6359+00	2.5054-01	7.4694-02
1.40: 4-01	1.1952+00	1.6833-01	9.7436-02
1.2995-01	1.5709+00	2.0414-01	8.1324-02
1.2028-01	1.7935+00	2.1572-01	8.2960-02
1.1165-01	1.6665+00	1.8606-01	8.7953-02
1.0392-01	1.7729+00	1.8424-01	8.8129-02
9.6961-02	1.3953+00	1.8377-01	9.2243-02
9.0052-02	1.6328+00	1.4807-01	1.0421-01
8.4992-02	1.9960+00	1.6964-01	9.6224-02
7.9020-02	2.5892+00	2.0667-01	8.2751-02
7.5107-02	2.1772+00	1.6352-01	1.0347-01
7.0800-02	2.4871+00	1.7609-01	9.3534-02
6.6351-12	2.7103+00	1.8119-01	9.5225-02
6.3224-02	2.8382+00	1.7944-01	9.2457-02
5.901.3-42	3.3534+00	2.0081-01	8.5372-02
5.6801-02	3.2853+00	1.8661-01	9.6415-02
5.2632-02	3.5996+00	1.8945-01	6.6815-02
4.7720-02	4.0468+00	1.9311-01	6.4130-02
4.3460-42	5.3322+00	2.3177-01	5.6968-02
3.9759-02	5.8373+00	2.3407-01	5.7847-02
3.0508-02	8.0395+00	2.9533-01	5,1259-02

LUNIINULD 7.0-INCHES LH2 AT 37 DEGREES USING THERMAL NEUTRON SPECTRUM FOR A BANK OF 32 VERTICAL BE3 DETECTORS LARGEST FRACTIONAL ERROR = .20 ENERGY RESOLUTION FOR GROUPING = .05 NEUTRON FLUX RELATIVE FLUX\*ENERGY NEUTRON ENERGY (RELATIVE UNITS) (RELATIVE UNITS) UNCERTAINTY (EV) 5.0914-02 9.4226+00 3.1700-01 3.3042-02 4.4424-02 1.3164+01 4.0943-01 3.11,2-02 1.6759+01 4.8335-01 4.0948-02 2.8841-02 5.4096-01 3.9527-02 2.0170+01 2.6820-02 3.6338-02 2.5287+01 6.3235-01 2.5007-02 3.3609-02 7.6700-01 3.2313+01 2.3375-02 9.6586-01 3.0053-02 4.4103+01 2.1900-02 2.9236-02 5.2065+01 1.0706+00 2.0562-02 2.6067-02 1.3721+00 1.9340-02 7.0922+01 2.3547-02 9.4324+01 1.7201+00 1.8230-02 2.0377-02 1.3627+02 2.3470+00 1.7223-02 2,9386+00 1.3034+02 1.8360-02 1.6295-02 1.7321-02 2.1761+02 3,3601+00 1.5441-02 2.7541+02 4.0356+00 1.6013-02 1.4053-02 1.2600-02 3.2513+02 4.4712+00 1.3752-02 1.1750-02 5.3395+00 1.27:9-02 4.1816+02 1.1463-02 4.8682+02 5.7863+00 1.1800-02 1.1595-02 5.2885+02 5.8639+00 1.1080-02 1.1985-02 1.0367-02 5.4618+02 5.6622+00 1.2330-02 5.5552+00 9.7110-03 5.7201+02 1.2590-02 6.0384+02 5.5051+00 9.1168-03 1.2933-02 6.2934+02 5.3955+00 8.5733-03 1.3524-02 6.3289+02 5.1122+00 8.0770-03 1.3908-02 6.5446+02 4.9895+00 7.0230-03 6.6623+02 1.4419-02 4.8011+00 7.2063-03 1.4856-02 6.7335+02 4.5935+00 6.8219-03 1.5150-02 4.3320+00 6.4076-03 6.6980+02 1.2602-02 7.2519+02 4.4155+00 6.0837-03 6.8500+02 1.3031-02 5.6939-03 3.9003+00 7.0533+02 3,7638+00 1.3638-02 5.3362-03 3.7218+00 1.3845-02 7.4272+02 5.0110-03 8.1170+02 3.8271+00 1.2726-02 4.7149-03 1.2546-02 8.2771+02 3.6786+00 4.4443-03 1.2892-02 7.9864+02 3.3513+00 4.1963-03 1.3240-02 3.90 5-03 8.2060+02 3.2566+00 1.3605-02 3.7587-03 8.4987+02 3.1944+00 1.4185-02 8.5690+02 3.0548+00 3.5650-03 1.3135-02 2.9633+00 3.3017-03 8.8069+02 2.7749+00 1.3855-02 3.1010-03 8.7784+02 1.4234-02 2.971.3-03 2.7492+00 9.2402+02 1.4896-02 9.3275+02 2.6167+00 2.8054-03 1.5576-02 2.6497-03 9.4033+02 2.4916+00

RUNS 15, RU 2-20-65

2.5065-03

2.4024+00

1.6177-02

9.5848+02

4-3-69

## TABLE E71

CONTINUEL			
THERMAL NE. TRUN SI	PECTRUM FOR 7.0-3	INCHES LH2 AT 37 D	EGREES USING
ENERGY RESOLUTION	FUR GROUPING = .	5 LARGEST FRAC	TIONAL ERROR = .20
NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.3/47-03	9.5968+02	2.2790+00	1.6962-02
2.2530-03	9.9282+02	2.2368+00	1.7464-02
2.1293-03	9.7750+02	2.0819+00	1.6892-02
2.0063-03	1.0084+03	2.0232+00	1,7571-02
1.6933-03	9.7184+02	1.8400+00	1.8775-02
1.7890-03	9.8588+02	1.7643+00	1.9628-02
1.6941-03	9,9383+02	1.6836+00	2.0480-02
1.6061-03	1.0120+03	1.6254+00	2.1275-02
1.51-4-03	9.7182+02	1.4756+00	2.1182-02
1.4310-03	9.7701+02	1.3987+00	2.2288-02
1.35, 0-03	9.6059+02	1.3068+00	2.3622-02
1.27.9-03	9,5977+02	1.2274+00	2.4675-02
1.2110-03	9.4803+02	1.1486+00	2.6078-02
1.1494-03	9.3450+02	1.0741+00	2.7550-02
1.0881-03	9.1065+02	9.9741-01	2.7341-02
1.0270-03	8.9472+02	9.1959-01	2,9125-02
9.72+3-04	8.7470+02	8.5058-01	3.0980-02
9.2141-04	8.5129+02	7.8439-01	3.3005-02
8.7430-04	8.2836+02	7.2424-01	3.5113-02
0.2814-04	8.0225+02	6.6438-01	3.5345-02
7.8310-04	7.7429+02	6,0635-01	3.7862-02
7.4100-04	7.4297+02	5.5101-01	4.0630-02
7.0330-04	7.1597+02	5.0360-01	4.3439-02
0.0015-04	6.8436+02	4.5589-01	4.4256-02
6.30 4-04	6.5203+02	4.1080-01	4.7624-02
5.9079-04	6.1079+02	3.6809-01	5.1276-02
5.0010-04	5.8134+02	3,2938-01	5.5062-02

E-79

4-3-69

TABLE E71

10N5 15, KU 2-20-66 Ċ

RUNS 1.8187 2-20-60

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESULTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

HEUTKON ENERGY	NEUTRON FLUX	FLUX*EIJERGY	RELATIVE
( = 1, )	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
4.4200+01	7.1965-03	3,1856-01	8.8753-02
1.4030+01	2.7567-02	4.0898-01	7,6827-02
7.3472+00	5,6235-02	4.1317-01	7.5219-02
4.3725+00	1.1704-01	5.1176-01	5.9180-02
2.891,0+00	1.7272-01	5.0030-01	5.8721-02
2.0592+00	2,5539-01	5.2590-01	5.4019-02
1.5387+00	3.0201-01	4.6470-01	6.0406-02
1.1931+00	4.1282-01	4.9254-01	5.2289-02
9.5213-01	5.2061-01	4.9569-01	5,3609-02
7.7739-01	6.0786-01	4.7254-01	5.4765-02
6.4072-01	6.9868-01	4.5185-01	5,3603-02
5.4041-01	8,4918-01	4.6400-01	5.1980-02
4.6776-01	9.1158-01	4.2640-01	5.4700-02
4.0493-01	9.1746-01	3.7151-01	6.1639-02
3.5398-01	1,1522+00	4.0786-01	5.4128-02
3.12:7-01	1.2006+00	3.7467-01	5.9590-02
2.7719-01	1.5333+00	4.2502-01	5.2391-02
2.47:4-01	1.5177+00	3.7615-01	5.8797-02
2.2292-01	2.0537+00	4.5781-01	5.1128-02
2.0157-01	2.0730+00	4.1785-01	5.4305-02
1.8315-01	2.3928+00	4.3824-01	5.1466-02
1.6715-01	2.5700+00	4.2958-01	5.2229-02
1.5315-01	2.4499+00	3.7520-01	5.6925-02
1.40.4-01	2,5840+00	3.6393-01	5.8349-02
1.2995-01	2.6034+00	3.3831-01	6.1013-02
1.1547-01	2,7381+00	3,1754-01	6.8354-02
1.0392-01	2,3637+00	2.4564-01	7.4100-02
9.0961-02	3,0026+00	2.9114-01	6,7636-02
9.00.2-02	3.1900+00	2.8928-01	6.9156-02
8.4992-02	3,1039+00	2.6381-01	7.3171-02
7.9820-02	3.6854+00	2.9417-01	6.9940-02
7.5107-02	3,2458+00	2.4378-01	7.4860-02
7.08.0-02	3.4496+00	2.4423-01	7.7521-02
6.6851-02	3,6691+00	2.4528-01	7.7725-02
6.3224-02	4.5991+00	2.9077-01	6.9086-02
5.98:3-02	4.6028+00	2.7563-01	7.0896-02
5.6801-02	5.9379+00	3.3728-01	6,1119-02
5.2032-02	5.6194+00	2,9576-01	4,9236-02
4.7720-02	6.9466+00	3.3149-01	4.6431-02
4.3400-02	8.4894+00	3,6900-01	4,4095-02
3.9759-02	9.4922+00	3.7740-01	4.4868-02
3.6578-02	1.1777+01	4.2995-01	4.1766-02
3.3042-02	1.4079+01	4.7365-01	4.0292-02

TABLE E72

RUNS REPR7 2-20-65 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
3.1102-02	1.8840+01	5.8596-01	3.6693-02
2.8841-02	2.7831+01	8.0267-01	3,1004-02
2.6820-02	3,0053+01	8,0602-01	3.1493-02
2.5007-02	3, 9721+01	9.6830-01	2 9148-02
2.3375-02	4 9204+01	1 1269+00	2 7242-02
2.1900-02	5 0630+01	1 3050+00	2 5902-02
2.15.0-02	7 1349401	1 5200+00	2.0702-02
2.0362-02	0 4 407 - 01	1 9260.00	2.4215-02
1.9346-02	9.4367+01	1.8260+00	2.2317-02
1.8236-02	1.2/6/+02	2.3282+00	2.0087-02
1.7223-02	1.7559+02	2.9915+00	1.7859-02
1.6295-02	2.21 +02	3.6121+00	1.6444-02
1.5441-02	2.61 9402	4.1133+00	1.5588-02
1.4053-02	3.022.5+02	4.4378+00	1.5188-02
1.3752-02	3.4833+02	4.7902+00	1.2116-02
1.2769-02	4.5289+02	5.7830+00	1.1228-02
1.1880-02	5.1540+02	6.1260+00	1.1104-02
1.1088-02	5.3714+02	5,9558+00	1.1450-02
1.0367-02	5.7552+02	5,9664+00	1.1615-02
9.7118-03	6.0580+02	5.8834+00	1.1897-02
9.1168-03	6.1918+02	5.6449+00	1.2355-02
8.5733-03	6.4454+02	5.5258+00	1.2707-02
8.0176-03	6.8866+02	5.5627+00	1.2879-02
7.6238-03	6.9035+02	5.2631+00	1.3461-02
7.2063-03	7.3587+02	5.3029+00	1.3653-02
6.8219-03	7.3283+02	4,9993+00	1.4174-02
6.4676-03	7.2596+02	4.6952+00	1.4463-02
6.08.7-03	7.6119+02	4.6347+00	1.2232-02
5.6939-03	7.4982+02	4.2694+00	1.2401-02
5.3362-03	7.6807+02	4.0986+00	1.2986-02
5.0110-03	8,1618+02	4.0899+00	1.3132-02
4.7149-03	8.7463+02	4,1238+00	1,2206-02
4.444.5-03	8.7768+02	3,9007+00	1.2114-02
4.19.5-03	8.4281+02	3.5367+00	1.2490-02
3.9665-03	8-5086+02	3,3766+00	1.2943-02
3.7587-03	8.7841+02	3,3017+00	1.3328-02
3.5650-03	9.9143+02	3 1779+00	1.3816-02
3.3647-03	9 0911+02	3 0500+00	1 2842-02
3.1610-03	9.2293102	2 9174400	1.3450-02
2.975 -03	9 6485+02	2 8707+00	1 3839-02
2.7/55-05	1 0032402	2 010/400	1 //20/-02
2.8054-03	1.0032403	2.5144700	1 4762-02
2.0497-03	1.0349+03	2. 1422400	1 5569-02
2.5065-03	1.0256+05	2,5/0/+00	1.5568-02
2.3/47-03	1.0159+05	4125+00	1.6404-02

4-3-69

TABLE E72

RUNS R8,R7 2-20-66 CUNTINUED

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
2.2530-03	1.0359+03	2.3339+00	1.7039-02
2.1298-03	1.0733+03	2.2859+00	1.6021-02
2.0063-03	1.0488+03	2.1042+00	1.7152-02
1.8933-03	1.0989+03	2.0805+00	1.7550-02
1.7896-03	1,0888+03	1.9485+00	1.8519-02
1.6941-03	1.0869+03	1.8413+00	1.9498-02
1.6061-03	1.0947+03	1.7582+00	2.0380-02
1.5184-03	1.0825+03	1.6437+00	1,9968-02
1.4316-03	1.0691+03	1.5305+00	2.1202-02
1.3520-03	1.1013+03	1,4890+00	2,1991-02
1.2789-03	1.0888+03	1.3925+00	2.3064-02
1.2116-03	1.0768+03	1.3047+00	2.4362-02
1.1494-03	1.0633+03	1.2222+00	2.5717-02
1.0881-03	1.0440+03	1,1360+00	2.5513-02
1.0278-03	1.0220+03	1.0504+00	2.7143-02
9.7243-04	9.9926+02	9.7171-01	2.8875-02
9.2141-04	9.7546+02	8,9880-01	3.0722-02
8.7430-04	9.5116+02	8.3160-01	3.2659-02
8.2814-04	°.2136+02	7.6302-01	3,2880-02
7.8310-04	8,9105+02	6,9778-01	3.5199-02
7.4163-04	8.6002+02	6.3782-01	3,7682-02
7.0338-04	8.2858+02	5.8281-01	4.0310-02
6.6615-04	7.9331+02	5.2846-01	4.1064-02
6.3004-04	7.5577+02	4,7617-01	4.4223-02
5.9079-04	7.1924+02	4.2924⇒01	4.7528-02
5.6610-04	e.7671+02	3.8309-01	5,1156-02

4-3-69

RUNS R9, R10 2-20-55

4-3-69

THERMAL NEGTRON SPECTRUM FOR 2.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BE3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*EMERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
4.4266+01	1.1523-02	5,1008-01	5.8015-02
1.4830401	3.7294-02	5.5329-01	5.9157-02
7.3472+00	8.7031-02	6.3943-01	5.0552-02
4.3725+00	1.3469-01	5.8893-01	5.4650-02
2.89-6+00	2.1591-01	6.2830-01	4.9565-02
2.0592+00	2.9785-01	6.1333-01	4.7870-02
1.5367+00	3.8808-01	5.9714-01	4.7515-02
1,1931+00	4.9902-01	5,9538-01	4.6917-02
9.5213-01	6.2649-01	5.9650-01	4.4865-02
7.7739-01	7.4522-01	5.7933-01	4.5806-02
6.4672-01	9.3617-01	6.0544-01	4.1845-02
5.4641-01	1.0047+00	5.4898-01	4.6850-02
4.6776-01	1.0402+00	4-8656-01	4.9028-02
4.0493-01	1,1733+09	4.7510-01	4.9260-02
3.5390-01	1.3284+00	4.7023-01	4.9554-02
3-1207-01	1.4146+00	4.4145-01	5.0456-02
2.7719-01	1.7502+00	4.8514-01	4.7658-02
2.47:4-01	2.1201+00	5.2545-01	4.3496-02
2.2292-01	2.6071+00	5.8117-01	4.1896-02
2.0157-01	2.0067+00	5.3753-01	4.4452-02
1.8315-01	2.9116+00	5.3326-01	4.4858-02
1.6715-01	3,0531+00	5,1033-01	4.4198-02
1.5315-01	3.2191+00	4.9301-01	4.6295-02
1.4084-01	3.3121+00	4.6648-01	4.7514-02
1.2995-01	3.2699+00	4.2492-01	5.0168-02
1.2028-01	3.5137+00	4.2263-01	5.0729-02
1.1165-01	3.6912+00	4.1212-01	5.2385-02
1.0392-01	3.3993+00	3.5326-01	5.7381-02
9.6961-02	3.6499+00	3,5390-01	5.5903-02
9.0002-02	3.4773+00	3.1533-01	6.4638-02
8.4992-02	4.3450+00	3,6929-01	5.5407-02
7.9020-02	4.3788+00	3,4952-01	5.7854-02
7.5107-02	4.0935+00	3.0745-01	6.4554-02
7.0800-02	4.6855+00	3.3173-01	6.1975-02
6.6851-02	4.6633+00	3.1175-01	6.1306-02
6.3224-02	5.0646+60	3.2020-01	6.4934-02
5.9853-02	6.0847+00	3,6437-01	5,7847-02
5.6001-02	6.3817+00	3.6249-01	5.7802-02
5.2032-02	6,9156+00	3,6398-01	4.2634-02
4.7720-12	8.0984+00	3.8646-01	4.1097-02
4.3460-02	1.0338+01	4.5152-01	3.7446-02
3.9759-02	1.3058+01	5,1917-01	3.4909-02
3.0208-02	1.5464+01	5.6456-01	3.4282-02

RUNS R9, R10 2-20-66 TABLE E73 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
3.3042-02	1.7111+01	5.7565-01	3.4288-02
3.1102-02	2,2749+01	7.0754-01	3.0892-02
2.8841-02	2,9855+01	8.6105-01	2.8432-02
2.6820-02	3.5849+01	9.6147-01	2,7145-02
2.5007-02	4.4842+01	1,1214+00	2.5451-02
2.3375-02	5.2983+01	1.2385+00	2.4569-02
2.1900-02	6.4661+01	1.4161+00	2,3222-02
2.0562-02	7.9143+01	1.6273+00	2.1914-02
1.9346-02	9.4333+01	1.8250+00	2.1062-02
1.8236-02	1,2411+02	2.2633+00	1,9142-02
1.7223-02	1,7022+02	2,9317+00	1,6961-02
1.6295-02	2.0040+02	3.2655+00	1,6315-02
1.5441-02	2.3334+02	3.6030+00	1.5618-02
1.4653-02	2.6559+02	3.8917+00	1.5266-02
1.3752-02	3.0236+02	4,1581+00	1,2219-02
1.2769-02	3.6185+02	4.6205+00	1.1804-02
1.1856-02	4.1196+02	4.8966+00	1.1657-02
1.1068-02	4.3289+02	4.7999+00	1.1966-02
1.0367-02	4.6537+02	4.8245+00	1.2133-02
9.7118-03	4.7781+02	4.6404+00	1.2591-02
9.1168-03	4,9685+02	4.5297+00	1.2975-02
8.5733-03	5.2755+02	4.5228+00	1.3192-02
8.0776-03	5.5334+02	4.4697+00	1.3507-02
7.6238-03	5.7865+02	4.4115+00	1.3830-02
7.2063-03	5.8732+02	4,2324+00	1.4372-02
6.8219-03	6.0874+02	4.1528+00	1.4616-02
6.4676-03	6.1465+02	3.9753+00	1.4776-02
6.08A7-03	6.4064+02	3,9007+00	1.2535-02
5.6939-03	5.9941+02	3.4130+00	1.3025-02
5.3362-03	6.5030+02	3,4701+00	1.3302-02
5.0110-03	6.7641+02	3.3895+00	1.3550-02
4.7149-03	7.0932+02	3.3444+00	1.2737-02
4.444.3-03	6.9017+02	3.0673+00	1.2838-02
4.1963-03	6.6419+02	2.7871+00	1.3208-02
3.9685-03	6.7090+02	2.6625+00	1.3689-02
3.7587-03	7.1343+02	2.6816+00	1.3895-02
3.5650=03	7,1823+02	2.5605+00	1.4474-02
3.3647-03	7.1347+02	2,4006+00	1.3642-02
3.1610-03	7.5503+02	2.3866+00	1.3972-02
2.9753-03	7.7847+02	2.3162+00	1.4494-02
2.8054-03	7.8959+02	2.2151+00	1.5125-02
2.6497-03	8-1179+02	2,1510+00	1.5647-02
2.5065-03	8-3829+02	2,1012+00	1.6201-02
2.0000-00	O O O O C O C O C		

THERMAL NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 37 DEGREES USING A BANK OF 32 VENTICAL BE3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROP = .20 NEUTRON ENERGY RELATIVE NEUTRON FLUX FLUX\*ENERGY (RELATIVE UNITS) (RELATIVE UNITS) UNCERTAINTY (EV) 2.3747-03 8.5139+02 2.0218+00 1.6838-02 8.8006+02 1.9828+00 1.7349-02 2.2530-03 1.6643-02 8.7971+02 1.8736+00 2.1298-03 8.8754+02 1.7458-02 1.7807+00 2.0063-03 1.8731-02 1.8933-03 8.5674+02 1.6221+00 8.8606+02 1.5857+00 1.9341-02 1.7896-03 2.0256-02 1.6941-03 8.9360+02 1.5138+00 1.6061-03 8.8600+02 1.4230+002.1303-02 1.5184-03 8,9600+02 1.3605+00 2.0618-02 8.9296+02 1.2784+00 2.1909-02 1.4310-03 8.9144+02 2.3018-02 1.3520-03 1,2052+00 9.0071+02 1.1519+00 2.3781-02 1.2789-03 2.5060-02 1.2110-03 8.9468+02 1.0840+00 2.6432-02 8.8460+02 1.0168+00 1.1494-03 2.6139-02 8.7367+02 9.5064-01 1.0881-03 2.7749-02 8.5850+02 8.8237-01 1.0278-03 2.9465-02 9.7243-04 8.4189+02 8,1868-01 3.1266-02 9.2141-04 8.2574+02 7.6085-01 3.3149-02 8.7430-04 8.0881+02 7.0714-01 7.8678+02 6.5156-01 3.3288-02 8.2814-04 7.6503+02 5.9909-01 3.5519-02 7.8310-04 3.7935-02 7.4089+02 5.4947-01 7.4163-04 4.0517-02 7.1489+02 5.0284-01 7.0338-04 4.5970-01 4.1069-02 6.6015-04 6,9009+02

6.6141+02

6.3157+02

5,9882+02

RUNS R9,R10 2-20-66 CONTINUED

6.3004-04

5.9679-04

5.6010-04

4.1671-01

3.7691-01

3.3899-01

4-3-69

4.4051-02

4.7199-02

5.0537-02

TABLE E73

RKNS T1, T2 2-22-66

4-3-69

THERMAL NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .10 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
3.0003+00	6.1019-02	1.8308-01	1.9445-01
1.3923+00	9.3118-02	1.2965-01	1,9650-01
7.7346-01	1.9215-01	1.4862-01	1,7823-01
5.4761-01	2.4748-01	1.3552-01	1.7184-01
4.1874-01	3.0088-01	1.2599-01	1.8855-01
3.3945-01	3.6137-01	1.2267-01	1.8298-01
2.8074-01	3.8193-01	1.0722-01	1.9377-01
2.4084-01	5,1873-01	1.2493-01	1.7293-01
2.0160-01	4.6804-01	9.4357-02	1.6825-01
1.6757-01	8.6111-01	1.4430-01	1.1383-01
1.4637-01	7.4839-01	1.0954-01	1.4946-01
1.2895-01	1.0585+00	1.3649-01	1.1275-01
1.1447-01	1.0400+00	1.1905-01	1.3589-01
1.0229-01	1.1551+00	1,1816-01	1.3331-01
9.0819-02	8.8620-01	8.0484-02	1.6708-01
8.0161-02	1.1727+00	9.4005-02	1.3762-01
7.1275-02	1.2852+00	9.1603-02	1.3534-01
6.3788-02	1.6098+00	1.0269-01	1.2124-01
5.6853-02	1.5937+00	9,0607-02	1.2418-01
5.0489-02	1.8915+00	9.5500-02	1.1542-01
4.5137-02	2.2016+00	9,9374-02	1.0360-01
4.0258-02	2.8989+00	1.1670-01	7.9742-02
3.5832-02	3.2864+00	1.1776-01	7.8700-02
3.2101-02	3.5805+00	1.1494-01	8.6667-02
2.8724-02	5.2324+00	1.5030-01	6.0286-02
2.5678-02	6.8742+00	1.7652-01	5.5489-02
2.2954-02	8.45/1+00	1.9412-01	5.0181-02
2.0519-02	1.1982+01	2.4586-01	4.7114-02
1.8357-02	1.8422+01	3.3817-01	3.59/3-02
1.6438-02	3.0492+01	5.0123-01	2.8/61-02
1.4/37-02	5.0851+01	7.4939-01	2.25/2-02
1.3219-02	7.3081+01	9.6606-01	2.0595-02
1.1865-02	1.0585+02	1.2559+00	1.6226-02
1.0611-02	1.3208+02	1.4079+00	1.6220-02
9.5008-03	1.4836+02	1.4095+00	1 6535-02
8.5217-03	1.6651+02	1.4189+00	1.6535-02
7.0293-03	1.8293+02	1 3006+00	1 7651-02
6 1570 03	1 0250+02	1 1851+00	1 7205-02
6.1034-03	2 0307+02	1 1223+00	1 7043-02
5.5266-03	2.0307+02	1 0636+00	1 6958-02
4.9007-03	2.0721402	1 0116+00	1 5565-02
3 00/1-03	2 3556+02	G 4130_01	1 5474-02
3.3764-03	2.3330+02	7.4139=01	1.07/4-02

TABLE E74

KKNS T1, T2 2-22-66 CUNTINUED

THERMAL NEUTRON SPECTRUM FOR 13.0-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .10 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
3.5885-03	2.5291+02	9,0757-01	1.5977-02
3.2242-03	2.5961+02	8.3703-01	1.6839-02
2.8992-03	2.7707+02	8.0328-01	1.7434-02
2.6040-03	2.8355+02	7.3836-01	1.8615-02
2.3369-03	2.9435+02	6.8787-01	1.9457-02
2.1007-03	3,1286+02	6.5722-01	2.0120-02
1.8880-03	3.0629+02	5.7828-01	2.1798-02
1.6969-03	3.0946+02	5.2512-01	2.3954-02
1.5258-03	2,9706+02	4.5325-01	2.5613-02
1.3706-03	2.7690+02	3.7952-01	3.0328-02
1.2305-03	2,9573+02	3.6390-01	3.0282-02
1.1045-03	2.6750+02	2.9545-01	3.4781-02
9.9159-04	2,5307+02	2.5094-01	4.1954-02
8.9057-04	2.3808+02	2.1203-01	4.5196-02
8.0035-04	2.0455+02	1.6371-01	5.6531-02
7.1965-04	1.8082+02	1.3016-01	6.9326-02
6.4740-04	1,5019+02	9.7233-02	9.0561-02
5.8174-04	1,1192+02	6.5108-02	1.5564-01

4-3-69

RUNS 14,13 2-22-66

4-3-69

THERMAL NEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .10 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
•			
6.7179+00	4.1298-02	2.7744-01	1.7707-01
3.0003+00	6.9553-02 -	2.0868-01	1.7535-01
1.6674+00	1.2212-01	2.0362-01	1.7586-01
1.1173+00	1.7728-01	1.9807-01	1.5997-01
8.6186-01	2.7293-01	2.3523-01	1.6985-01
7.3981-01	2.8596-01	2.1156-01	1.8129-01
6.4198-01	2,9432-01	1.8895-01	1.9911-01
5.6234-01	3.5072-01	1,9722-01	1.7712-01
4.8257-01	3.7427-01	1.8061-01	1.5314-01
3.9649-01	3.2328-01	1.2818-01	1.8451-01
3.3080-01	4.9296-01	1.6307-01	1.5035-01
2.8694-01	4.9833-01	1.4299-01	1.6980-01
2.5126-01	4.9987-01	1.2560-01	1,8003-01
2.2184-01	6.2275-01	1,3815-01	1.5652-01
1.9730-01	7.9241-01	1.5634-01	1.3327-01
1.7662-01	6.5330-01	1,1539-01	1.8622-01
1.5643-01	1.0776+00	1.6357-01	1.0752-01
1.3725-01	1.2095+00	1.6000-01	1,0315-01
1.2139-01	8.7348-01	1.0603-01	1.6706-01
1.0812-01	1.3843+00	1.4967-01	1.0449-01
9.6922-02	1,2520+00	1,2135-01	1.2771-01
8.6315-02	1.4199+00	1.2256-01	1.1604-01
7.6417-02	1,7802+00	1.3604-01	9.8875-02
6.8130-02	1.7600+00	1.1991-01	1.1146-01
6.0497-02	1.8941+00	1.1459-01	9.5718-02
5.3530-02	2.3505+00	1.2582-01	9,2328-02
4.7701-02	2.6834+00	1.2800-01	8.7910-02
4.2776-02	3.0310+00	1.2965-01	8.4469-02
3.8266-02	3.8406+00	1.4696-01	8.0291-02
3.4157-02	5.0543+00	1.7264-01	6.7754-02
3.0458-02	5.1541+00	1.5698-01	7.1666-02
2.7136-02	8.5725+00	2.3262-01	5,1295-02
2.4336-02	1.0933+01	2.6607-01	5.0479-02
2.1820-02	1.4315+01	3.1235-01	4.3216-02
1.9562-02	2.0691+01	4.0476-01	3.8093-02
1.7549-02	3.3098+01	5.8084-01	3.0307-02
1.5675-02	6.0420+01	9.4708-01	2,2939-02
1.4017-02	9.5044+01	1.3322+00	1,9820-02
1.2545-02	1.3718+02	1.7209+00	1,6999-02
1.1237-02	1.8479+02	2.0765+00	1.5850-02
1.0076-02	2.1491+02	2,1654+00	1.5357-02
9.0104-03	2.3365+02	2.1053+00	1.5422-02
8.0719-03	2.4512+02	1.9786+00	1.6534-02

E-88

LARGEST FRACTIONAL ERROR = .20 ENERGY RESOLUTION FOR GROUPING = .10 FLUX\*ENERGY NEUTRON ENERGY NEUTRON FLUX RELATIVE (EV) (RELATIVE UNITS) (RELATIVE UNITS) UNCERTAINTY 1.9162+00 1.6617-02 7.2469-03 2.6441+02 2.7406+02 1.7804+00 1.6710-02 6.4963-03 2.3162+02 1.6385+00 1.6126-02 5.8183-03 1.6357-02 3.0314+02 1.5788+00 5.2083-03 3.1378+02 1.4629+00 1.5389-02 4.6622-03 1.4320-02 4.1747-03 3.5044+02 1.4630+00 1.4815-02 3.6733+02 1.3739+00 3.7401-03 1.2225+00 3.3531-03 3.6460+02 1.5939-02 3.9054+02 1.1751+00 1.6567-02 3.0089-03 1.1371+00 1.7056-02 4.2066+02 2.7031-03 1.8049-02 2.4263-03 4.2006+02 1.0192+00 4.2924+02 9.3437-01 1.9384-02 2.1760-03 4.4139+02 8.6354-01 2.0575-02 1.9564-03 2.1526-02 4.5583+02 8.0149-01 1.7563-03 2.3947-02 4.2251+02 6,6676-01 1.5781-03 2.5844-02 1.4173-03 4.3720+62 6.1964-01 2.9249-02 5.0428-01 1.2741-03 3.9579+02 1.1448-03 3,9192+02 4.4867-01 3.1674-02 3.8151+02 3.9242-01 3.5105-02 1.0286-03 9.2445-04 3.1668+02 2.9275-01 4.3502-02 4.9007-02 2.9150+02 2.4230-01 8.3121-04 2,1899-01 5.3082-02 2.9314+02 7.4704-04 6.6028-02 6.7128-04 2.4625+02 1.6530-01

1.4088-01

2.3351+02

RUNS 14,13 2-22-66 CUNTINUED

6.0332-04

THERMAL NEUTRON SPECTRUM FOR 10.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BE3 DETECTORS

4-3-69

7.6139-02

RUNS 15, T6 2-22-66

4-3-69

THERMAL NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 78 DEGREES US A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY (EV)	NEUTRON FLUX (RELATIVE UNITS)	FLUX*ENERGY (RELATIVE UNITS)	RELATIVE UNCERTAINTY
2.0529+01	1.9918-02	4.0890-01	1,9396-01
1.0615+01	3.7893-02	4.0223-01	1.7735-01
6.4805+00	6.0661-02	3,9311-01	1.6620-01
4.3672+00	8.2041-02	3,5829-01	1.7580-01
2.9361+00	1.0717-01	3.1466-01	1.5322-01
2.0855+00	1.6350-01	3.4098-01	1.5686-01
1.7428+00	2.1726-01	3.7864-01	1.8352-01
1.5618+00	2.5820-01	4.0326-01	1.7403-01
1.3413+00	2.0693-01	2.7756-01	1.7658-01
1.1105+00	3.2208-01	3.5767-01	1.2654-01
9.7308-01	4.6998-01	4.5733-01	1.3962-01
8.6186-01	3.5187-01	3.0326-01	1.4259-01
7.1452-01	3.5435-01	2.5319-01	1.3588-01
6.0019-01	4.5992-01	2.7604-01	1.4189-01
5.4435-01	6,1194-01	3.3311-01	1.5819-01
5.1160-01	6.6575-01	3.4060-01	1.5756-01
4.6006-01	6.2388-01	2.9201-01	1,2399-01
4.2931-01	9.1272-01	3,9184-01	1.3131-01
4.0626-01	7.2343-01	2,9390-01	1.7211-01
3.7520-01	5.7292-01	2,1496-01	1.6261-01
3.3882-01	6.9378-01	2.3507-01	1.4075-01
3.0749-01	8.6219-01	2.6511-01	1.2769-01
2.8031-01	8.9879-01	2.5194-01	1.2402-01
2.5658-01	9,7585-01	2,5038-01	1.2354-01
2.3574-01	1.1351+00	2.6759-01	1.2403-01
2.1734-01	1.1645+00	2.5309-01	1.1611-01
2.0101-01	1.3458+00	2.7052-01	1.0939-01
1.8645-01	1.2331+00	2,2991-01	1.3363-01
1.7343-01	1.3391+00	2.3224-01	1.2819-01
1.6172-01	1,5096+00	2.4413-01	1.2361-01
1.5115-01	1.5393+00	2.3267-01	1.3067-01
1.4159-01	1.9112+00	2.7061-01	1.0771-01
1.3291-01	2.1393+00	2.8433-01	1.0493-01
1.2500-01	2.1547+00	2.6934-01	1.1110-01
1.1778-01	1.8907+00	2.2269-01	1.2/05-01
1.1116-01	2.3090+00	2.5667-01	1.0664-01
1.0509-01	1.9040+00	2.0009-01	1.3575-01
9.9498-02	2.2271+00	2.2159-01	1.1269-01
9.4346-02	2.5194+00	2.3/70-01	1.1021-01
8.8467-02	1.9768+00	1.7488-01	1.21/3-01
8.20A8-U2	2,9810+00	2.4470-01	8.6930-02
7.6374-02	3.0098+00	2.298/-01	9.3053-02
7.1238-02	2,291/+00	1.6326-01	1,1550-01

RUNS T5+T6 2-22-66 TABLE E76 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
6.6602-02	2.8548+00	1.9014-01	1.0355-01
6.2404-02	3.3586+00	2.0959-01	9,9696-02
5.8590-02	3.5986+00	2.1084-01	9,2356-02
5.5116-02	3.4583+00	1.9061-01	1.0059-01
5.1943-02	4.1955+00	2.1793-01	9,3400-02
4.9030-02	4.8811+00	2.3935-01	9,0651-02
4.6366-02	4.1593+00	1.9285-01	1,1171-01
4.3908-02	5.4013+00	2.3716-01	9.3544-02
4.1643-02	5,5655+00	2.3176-01	9.7585-02
3.9220-02	6.2175+00	2.4385-01	7.8769-02
3.6697-02	6.8749+00	2.5229-01	7.6987-02
3.4410-02	7.7738+00	2.6750-01	7.6543-02
3.2332-02	9.8208+00	3.1753-01	6.5241-02
3.0437-02	9.6644+00	2.9416-01	7,0903-02
2.8706-02	1.1531+01	3.3101-01	6.6287-02
2.7120-02	1.4924+01	4.0474-01	6.1942-02
2.5063-02	1.7380+01	4,4602-01	5.8545-02
2.4323-02	1.6640+01	4.0473-01	6.2606-02
2.3086-02	2.3813+01	5.4975-01	5.3581-02
2.1808-02	2.7054+01	5.8999-01	4.5820-02
2.0509-02	3.2212+01	6.6064-01	4.4011-02
1.9324-02	4.1017+01	7.9261-01	4.0786-02
1.8241-02	5.5054+01	1.0042+00	3.6317-02
1.7250-02	6.3155+01	2.0894+00	3.5513-02
1.6341-02	9.0495+01	1.4788+00	3.0764-02
1.5503-02	1.2135+02	1.8813+00	2.7391-02
1.4654-02	1.4569+02	2.1349+00	2,3806-02
1.3805-02	1.8502+02	2.5542+00	2.2070-02
1.3023-02	2.2341+02	2.9095+00	2.08/9-02
1.2305-02	2.58/9+02	3.1844+00	2.0219-02
1.1644-02	3.1497+02	3.6675+00	1 9094-02
1.1032-02	3.2828+02	3.6216+00	1.9400-02
1.0467-02	3.4786+02	3.5411+00	1 0103-02
9.9013-03	3.7922+02	3.7548+00	1.0103-02
9.3407-03	3.8487402	3.5950+00	1 9176-02
8.8264-03	4.0043402	3 50/6+00	1 9737-02
8.3518-03	4.1360+02	3.5107+00	2.0011-02
7.5167-03	4 4459402	3.3400+00	2.0889-02
7.1176-07	4.6138+02	3,2821+00	2.0021-02
6.7215-03	4.0100402	3,1829+00	2.0402-02
6.3609-03	4.4028+02	2.8006+00	2.1435-02
6.0296-03	4.6516+02	2.8043+00	2.0739-02

RUNS T5,T6 2-22-66 CONTINUED TABLE E76

THERMAL NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
5.7214-03	4,7823+02	2,7361+00	2,0508-02
5.4201-03	5.0786+02	2.7527+00	1,9664-02
5.1260-03	5.2287+02	2.6802+00	2.0355-02
4.8553-03	4.9330+02	2.3951+00	2.0921-02
4.6056-03	5.0712+02	2.3356+00	1.9734-02
4.3625-03	5.6245+02	2.4537+00	1.8003-02
4.1267-03	5,6232+02	2.3205+00	1.8305-02
3.9094-03	5.8869+02	2.3014+00	1.8654-02
3.7088-03	5,7182+02	2,1208+00	1.9822-02
3.5144-03	6.1463+02	2,1601+00	1,9096-02
3.3265-03	6,1770+02	2.0548+00	1,9909-02
3.1533-03	6,0801+02	1.9172+00	2.0977-02
2.9932-03	6,3483+02	1.9002+00	2.1466-02
2.8386-03	6,5007+02	1.8453+00	2.1303-02
2.6897-03	6,5272+02	1.7556+00	2.2251-02
2.5522-03	6.5033+02	1.6598+00	2.3440-02
2.4199-03	7.1238+02	1.7239+00	2.2329-02
2.2928-03	6.8998+02	1.5820+00	2.3827-02
2.1756-03	6.8316+02	1.4863+00	2.5151-02
2.0631-03	7.5102+02	1.5494+00	2.4280-02
1.9553-03	6.7296+02	1.3158+00	2.6833-02
1.8558-03	7.1614+02	1.3290+00	2.7248-02
1.7605-03	6.8309+02	1.2026+00	2.8521-02
1.6695-03	6.7177+02	1.1215+00	2,9866-02
1.5826-03	6,9963+02	1.1072+00	3.0050-02
1.4998-03	7.2371+02	1.0854+00	3.0744-02
1.4234-03	6.7065+02	9.5460-01	3,3383-02
1.3506-03	6.9353+02	9.3668-01	3,3010-02
1.2812-03	6,7351+02	8.8290-01	3,5004-02
1.2152-03	6.804/+02	7 2200-01	3,0400-02
1.1525-03	6.2732402	6 6355-01	4 1444-02
1.0930-03	6 0633402	6 3953-01	4 3507-02
1.0366-03	5.0033+02 5.07/(3.00	5,2052-01	4.5007-02
9.8310-04	5.8743402	4 9415-01	5.0986-02
9.0241-04	5 1128+02	4.5220-01	5.3602-02
0.3903-04	4.0994.502	4.1946-01	5.7071-02
7.9607-04	4.3529+02	3.4652-01	6.3655-02
7.5542-04	4.5103+02	3.4072-01	6.6951-02
7.1701-04	4.3717+02	3,1346-01	6.8884-02
6.7993-04	3.8494+02	2.6173-01	7.7183-02
6.4495-04	3.7707+02	2.4319-01	8.4495-02
6.1195-04	4.0857+02	2.5002-01	8.2127-02

4-3-69

RUNS 15, T6 2-22-66 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 7.0-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
5.8082-04	3,8556+02	2.2395-01	9.0969-02
5.5146-04	2,9785+02	1.6425-01	1.0706-01

RUNS T8, T7 2-22-66

## TABLE E77

4-3-69

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.6646+01	4,9230-02	8,1948-01	1.7743-01
1.2074+01	7.1103-02	8.5850-01	1.6009-01
8.1679+00	8.2318-02	6.7237-01	1.4263-01
5.2680+00	1.3204-01	6.9559-01	1.2561-01
3.6794+00	1.7064-01	6.2785-01	1.3359-01
2.7146+00	2.0722-01	5.6252-01	1.3626-01
2.0855+00	2.9458-01	6.1435-01	1,2006-01
1.6523+00	2.6347-01	4.3533-01	1.5728-01
1.4075+00	3,9253-01	5,5249-01	1.6933-01
1.2750+00	5,5359-01	7.0583-01	1.3752-01
1.1604+00	5.4281-01	6.2988-01	1.4797-01
1.0605+00	5.2602-01	5.5784-01	1.5935-01
9.7308-01	7.5999-01	7.3953-01	1.2158-01
8.9592-01	6.8137-01	6.1045-01	1,4298-01
8.2781-01	5.8140-01	4.8129-01	1.7097-01
7.6696-01	7.4208-01	5.6915-01	1.4562-01
7.1267-01	8,2921-01	5,9095-01	1.3866-01
6.6393-01	1.0620+00	7.0509-01	1,2039-01
6.2003-01	9.9271-01	6.1551-01	1.2879-01
5.8034-01	1.0374+00	6.0204-01	1.3541-01
5.4435-01	1.0282+00	5.5970-01	1.4051-01
5.1160-01	1,1619+00	5,9443-01	1.3255-01
4.8172-01	1,2509+00	6.0258-01	1.1834-01
4.5439-01	1.4970+00	6.8022-01	1.1474-01
4.2931-01	1.3409+00	5.7566-01	1.2932-01
4.0626-01	1,3231+00	5.3752-01	1.3141-01
3.8501-01	1.4506+00	5.5850-01	1.3253-01
3,5632-01	1,4697+00	5,2368-01	9,6078-02
3.2259-01	1.7857+00	5,7605-01	8,7691-02
2.9343-01	1.5644+00	4.5904-01	1.0214-01
2.6805-01	1.9024+00	5.0994-01	9,2322-02
2.4583-01	1.9402+00	4,7696-01	1.0395-01
2.2626-01	2.4277+00	5.4929-01	8.7614-02
2.0893-01	2.5890+00	5.4092-01	8.4904-02
1.9353-01	2,5036+00	4.8452-01	9,2978-02
1.7976-01	3.3087+00	5.9477-01	8,1152-02
1.6742-01	3,6670+00	6.1393-01	7,9505-02
1.5630-01	3.5594+00	5,5633-01	8.7095-02
1.4625-01	3.3740+00	4.9345-01	9.0712-02
1.3715-01	3.7033+00	5,0791-01	9.0414-02
1.2886-01	3.9345+00	5,0700-01	8.7546-02
1.2131-01	4.3965+00	5.3334-01	8.6485-02
1.1440-01	4.6026+00	5.2654-01	8.7640-02

RUNS 18,17 2-22-66 TABLE E77 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUXFENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
,			
1 0806-01	4 5269+00	4 8918-01	9 1354-02
1.0000-01	4.0637.00	4 3590-01	0 5670-02
1.0223-01	4.2637+00	4.3585-01	9.50/0-02
9.6871-02	4.7562+00	4.60/4-01	9.5240-02
9.1919-02	5.2695+00	4.8437-01	9.1131-02
8.6261-02	4.8216+00	4,1592-01	8,4443-02
8.0114-02	5.3996+00	4.3258-01	7.9809-02
7.4602-02	4.8467+00	3.6157-01	8.9407-02
6.9641-02	5.5432+00	3,8603-01	8.3004-02
6.5157-02	5.8645+00	3.8211-01	8,9606-02
6.1092-02	6.3045+00	3.8515-01	8.6544-02
5.7396-02	7.0559+00	4.0498-01	8.4357-02
5.4027-02	8,1249+00	4.3896-01	7.8773-02
5 09/1-02	7 9657+00	4.0582-01	8.4448-02
5.0948-02	9 5655+00	4.0302-01	0 3450-02
4.8121-02	8.000	4 0637-01	0.0400-02
4.5524-02	8.9200+00	4.0007-01	7 0114-02
4.3133-02	1.0814+01	4.0044-01	0.0554-02
4.0926-02	9.5118+00	3.8928-01	9.0554-02
3.8565-02	1.2914+01	4.9803-01	6.5030-02
3.6104-02	1.5433+01	5.5719-01	6.2200-02
3.3872-02	1.6674+01	5.6478-01	6,3128-02
3.1842-02	1.8520+01	5.8971-01	6.3032-02
2.9990-02	2.1760+01	6.5258-01	5,9023-02
2.8297-02	2.9101+01	8.2347-01	5.3427-02
2.6744-02	2.9844+01	7,9815-01	5.5317-02
2.5317-02	3.5459+01	8.9772-01	5,1656-02
2.4005-02	4.1282+01	9,9097-01	4,9284-02
2.2649-02	5.4946+01	1.2445+00	9.0048-02
2.1273-02	5.7576+01	1.2248+00	4.1252-02
2.0022-02	7,1121+01	1,4240+00	3.8246-02
1.8800-02	9,0853+01	1.7153+00	3.5615-02
1.7830-02	1 2123+02	2,1620+00	3,1938-02
1.6877-02	1 5466+02	2 6102+00	2 9447-02
1.6677-02	1.5400402	2.0102+00	2 6915-02
1.5998-02	1.9/32+02	3,1567+00	2.0715-02
1.5109-02	2.64/3+02	3.9998+00	2.2095-02
1.4220-02	3,2438+02	4.612/+00	2.0901-02
1.3406-02	3,6875+02	4.9435+00	2.0427-02
1.2657-02	4.3964+02	5.5645+00	1.9501-02
1.1968-02	5.3487+02	6.4013+00	1.8477-02
1.1332-02	5,9708+02	6,7661+00	1.8193-02
1.0744-02	6,1821+02	6,6420+00	1.8594-02
1.0157-02	6,5002+02	6,6023+00	1.7497-02
9.5749-03	6.8318+02	6.5414+00	1.7844-02
9.0418-03	6.9368+02	6.2721+00	1,8510-02

4-3-69

\_. \_ \_ \_ . \_

E-95

RUNS T8, T7 2-22-66 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
8.5502-03	7,1231+02	6.0904+00	1.9084-02
8.0985-03	7.6678+02	6.2098+00	1.9230-02
7.6818-03	7.7059+02	5,9195+00	1,9994-02
7.2695-113	7.9571+02	5.7844+00	1,9262-02
6.8646-03	8.2933+02	5.6930+00	1,9555-02
6.4927-03	7.6848+02	4.9895+00	2.0709-02
6.1501-03	8,1191+02	4,9933+00	2.0201-02
5.8338-03	8.3204+02	4.8540+00	1.9864-02
5.5238-03	8.3075+02	4.5889+00	1.9357-02
5.2213-03	8.4767+02	4.4259+00	2.0284-02
4.9431-03	8.5248+02	4.2139+00	2.0644-02
4.6866-03	8.3866+02	3,9305+00	1.9886-02
4.4372-03	9.7549+02	4.3284+00	1.7519-02
4.1954-03	9.2355+02	3.8747+00	1.8159-02
3.9728-03	9.4739+02	3,7638+00	1.8675-02
3.7673-03	1.0022+03	3,7756+00	1.8974-02
3.5684-03	9.8928+02	3,5301+00	1.9065-02
3.3762-03	1.0197+03	3.4427+00	1.9627-02
3.1991-03	1.0270+03	3.2855+00	2.0500-02
3.0356-03	1.0663+03	3.2369+00	2.1018-02
2.8778-03	1,1159+03	3.2113+00	2.0573-02
2.7258-03	1.0935+03	2.9807+00	2.1787-02
2.5855-03	1.1128+03	2.8771+00	2.2571-02
2.4507-03	1,1498+03	2.8178+00	2.2361-02
2.3213-03	1.1313+03	2.6261+00	2.3679-02
2.2018-03	1.1518+03	2.5360+00	2.4554-02
2.0873-03	1.2049+03	2.5150+00	2.4302-02
1.9777-03	1.1911+03	2.3556+00	2.5595-02
1.8765-03	1.2008+03	2.2533+00	2.6692-02
1.7796-03	1.2063+03	2.1467+00	2.7044-02
1.6871-03	1.1763+03	1.9845+00	2.8658-02
1.6016-03	1.1917+03	1.9086+00	2.9818-02
1.5199-03	1.2218+03	1.8570+00	2.9942-02
1.4420-03	1.2169+03	1.7548+00	3.1602-02
1.3677-03	1.1938+03	1.6328+00	3.2332-02
1.2970-03	1.1845+03	1.5363+00	3,4415-02
1.2298-03	1.0539+03	1.2961+00	3.7049-02
1.1060-03	1.09/1+03	1.2/92+00	5./911-02
1.1055-03	1.0002+03	1.1124+00	4.1012-02
1.04A1-03	1.0010+03	1.112/+00	4.1045-02
9.93/3-04	9.555/402	9.4750-01	4 8931-02
9.4223-04	9.0051402	7 7740-01	5 0918-02
8.9351-04	8./005+02	/.//40-01	0,0,10-05

4-3-69

KUNS T8,T7 2-22-66 CUNTINUED

THERMAL NEUTRON SPECTRUM FOR 4.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
			5 ( 001 00
8.4740-04	8.0940+02	6.8589-01	5.6094-02
8.03A1-04	8.8079+02	7.0799-01	5.5973-02
7.6258-04	7.3799+02	5.6278-01	6.3522-02
7.2362-04	7.6947+02	5.5680-01	6.3525-02
6.8680-04	7.5125+02	5,1596-01	6.8992-02
6.5201-04	6.5281+02	4.2564-01	7.7682-02
6.1848-04	7.0255+02	4.3451-01	7.8420-02
5.8686-04	6,9913+02	4,1029-01	8.1218-02
5.5704-04	5.7986+02	3,2301-01	9.8534-02

RUNS 19, T10 2-22-66

TABLE E78

THERMAL NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX (RELATIVE UNITS)	FLUX*ENERGY (RELATIVE UNITS)	RELATIVE UNCERTAINTY
1.2710+02	8.4956-03	1,0798+00	1.6050-01
3,1811+01	3.2547-02	1.0354+00	1.3090-01
1.4360+01	5.1421-02	7.3841-01	1.5132-01
9.1557+00	9.0583-02	8.2935-01	1.7632-01
7.1800+00	1,1527-01	8.2764-01	1.6774-01
5.7810+00	1.7330-01	1,0018+00	1.3255-01
4.7550+00	1.6147-01	7.6779-01	1.7368-01
3.9795+00	1.9946-01	7.9375-01	1.6311-01
3.3792+00	1.9683-01	6.6513-01	1.9144-01
2.9051+00	3.3909-01	9.8509-01	1.2299-01
2.5241+00	3.7090-01	9.3619-01	1.2361-01
2.2137+00	6.2562-01	1.3849+00	6.5803-02
1.9572+00	4.2528-01	8,3236-01	1.3577-01
1.7428+00	5,9887-01	1.0437+00	1.0573-01
1.5018+00	5.9958-01	9,3642-01	1.1519-01
1.4075+00	6.5654-01	9.2408-01	1.1364-01
1.2750+00	5.6839-01	7.2470-01	1.4378-01
1.1604+00	1.1360+00	1.3182+00	7.0036-02
1.0605+00	9.0393-01	9,5862-01	1.0472-01
9.7308-01	6.1080-01	5.9436-01	1,6134-01
8.9592-01	8,5065-01	7.6211-01	1.2410-01
8.2791-01	1.0238+00	8.4751-01	1.1421-01
7.6090-01	1.1002+00	8.4381-01	1.0920-01
7.1267-01	8.9519-01	6,3798-01	1.4497-01
6.6393-01	1.2796+00	8.4956-01	1.0946-01
6.2003-01	1.2856+00	7.9711-01	1.1208-01
5.8034-01	1.2194+00	7.0767-01	1.2437-01
5.4435-01	1.0170+00	5.5360-01	1.5995-01
5.1160-01	1.8168+00	9.2947-01	7.9635-02
4.8172-01	1.3407+00	6.4584-01	1.4294-01
4.5439-01	2.1657+00	9.8407-01	8.0025-02
4.2931-01	1.7967+00	/./134-01	1.1085-01
4.0626-01	1.2012+00	4.8800-01 5.5806-01	1 3903-01
3.8501-01	1.4518+00	5.5090-01	7 1134-02
3.5032-01	2.1221+00	6 4257-01	9 9014-02
3.2259-01	1 0898+00	5 5452-01	9.9368-02
2.5343-01	3 0538+00	9 1857-01	6 1012-02
2.4563-01	2.6484+00	6.5106-01	7.2171-02
2.2026-01	2.5726+00	5.8208-01	8.8572-02
2.0803-01	3.0128±00	6.2946-01	8.5546-02
1.9353-01	3.4816+00	6.7379-01	7.9449-02
1.7976-01	3,5538+00	6.3883-01	8.2381-02

E-98

.

RUNS T9, T10 2-22-66 CONTINUED

THERMAL NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(Ev)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.6742-01	3.6003+00	6.0276-01	8.3792-02
1.5630-01	4.5589+00	7.1256-01	7.1466-02
1.4625-01	4.4246+00	6.4710-01	8.3847-02
1.3715-01	4.7003+00	6.4465-01	7.7101-02
1.2886-01	4.5031+00	5.8027-01	8,4281-02
1.2131-01	4.2687+00	5.1784-01	9.4287-02
1.1440-01	5.1342+00	5.8735-01	8.2796-02
1.0806-01	5.7438+00	6.2068-01	7.8546-02
1.0223-01	4.8912+00	5.0003-01	9,9328-02
9.6871-02	6.0664+00	5.8766-01	8.0928-02
9.1919-02	5.5607+00	5.1113-01	8.7321-02
8.6261-02	5.2875+00	4.5611-01	8.5649-02
8.0114-02	5.0774+00	4.0677-01	9.1207-02
7.4002-02	6.4842+00	4.8373-01	7.8080-02
6.9641-02	5.9864+00	4.1690-01	8.8055-02
6.5157-02	6.5488+00	4.2070-01	8.0042-02
6.1092-02	8.4631+00	5.1703-01	7.0202-02
5.7396-02	7.9763+00	4.5781-01	7.9282-02
5.4027-02	8.6943+00	4.6973-01	7.1643-02
5.0946-02	1.0435+01	5.5162-01	7.1043-02
4.8121-02	9.0746+00	4.3008-01	7 7972-02
4.5524-02	1.1160+01	5.0805-01	7 2997-02
4.5133-02	1.4058+01		7 4949-02
4.0920-02	1,4350401	5.5077-01	6 6742-02
3.8565-02	1 7105+01	6 1756-01	6 0011-02
3.3070 02	2 0150+01	6 9252-01	5 8830-02
5.5872-02	2.0150+01	6.660%-01	6 1253-02
3.1842-02	2.0942+01	8 4050-01	5.2896=02
2. 3390-02	3 0051+01	8.5035-01	5.4916-02
2.6744-02	3 4018+01	9.0978-01	5.3224-02
2.5317-02	3.5850+01	9.0761-01	5.3988-02
2.4005-02	4.5694+01	1,0969+00	4.8941-02
2.2649-02	5.3470+01	1,2110+00	4.2378-02
2.1273-112	6.7321+01	1,4321+00	3,9361-02
2.0022-02	8.2526+01	1.6523+00	3.6769-02
1.8880-02	1.0208+02	1,9273+00	3.4507-02
1.7834-02	1.1899+02	2.1221+00	3.2873-02
1.6877-02	1.6907+02	2.8534+00	2.8774-02
1.5998-02	2.2795+02	3.6467+00	2.5677-02
1.5109-02	2.7176+02	4.1060+00	2.2271-02
1.4220-02	3.3873+02	4.8167+00	2.1005-02
1.3400-02	3,9520+02	5,2981+00	2.0176-02
-			

4-3-69

E-99
RUNS T9, T10 2-22-66 CONTINUED 4-3-69

THERMAL NEUTRON SP	PECTRUM FOR 2.5-IN	CHES LH2 AT 78 DE	GREES USING
ENERGY RESOLUTION	FOR GROUPING = .0	5 LARGEST FRAC	TIONAL ERROR = .20
NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
	(RECATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.2657-02	4.5810+02	5,7982+00	1.9576-02
1.1968-02	5.5216+02	6.6083+00	1.8543-02
1.1332-02	5.9359+02	6.7266+00	1.8655-02
1.0744-02	6.3/19+02	6.8460+00	1.8730-02
	6 (170+02	6.4121+00	1.8142-02
9.5749-03	6.0925+02	6 3134+00	1.8365-02
9.0418-03	7.1986+02	6.1549+00	1 9453-02
8.0995-03	7,1171+02	5.7638+00	2.0357-02
7.6818-03	7.5384+02	5,7908+00	2.0606-02
7.2695-03	7.9095+02	5.7498+00	1,9840-02
6.8646-03	8,1502+02	5-5948+00	2.0210-02
6.4927-03	7.6547+02	4.9700+00	2.1255-02
6.1501-03	8.1118+02	4.9888+00	2.0765-02
5.8338-03	8.1994+02	4.7834+00	2.0483-02
5.5238-03	8.2892+02	4.5788+00	1.9825-02
5.2213-03	8.6008+02	4.4907+00	2.0642-02
4.9431-03	8.5706+02	4,2365+00	2.1093-02
4.6866-03	8.3674+02	3,9215+00	2.0426-02
4.4372-03	9.3670+02	4.1563+00	1.8227-02
4.1954-03	8.9321+02	3.7474+00	1.8904-02
3.9728-03	9.2160+02	3.6613+00	1.9366-02
3.7673-03	9.4000+02	3.5413+00	2.0080-02
3,5084-03	9.6036+02	3.4269+00	1.9778-02
3.1901-03	9.8041402	3 2803+00	2.0534-02
3.0356-03	1.0400+03	3,1570+00	2.1793-02
2.8778-03	1.0707+03	3,0813+00	2.1607-02
2.7258-03	1.0791+03	2,9414+00	2.2537-02
2.5855-03	1.0914+03	2.8218+00	2.3423-02
2.4507-03	1.1407+03	2,7955+00	2.3112-02
2.3213-03	1.1597+03	2.6920+00	2.4147-02
2.2018-03	1.1331+03	2,4949+00	2.5450-02
2.0873-03	1.1052+03	2.3069+00	2,6021-02
1.9777-03	1.2157+03	2,4043+00	2.5920-02
1.8765-03	1.1601+03	2.1769+00	2.8073-02
1.7796-03	1.1360+03	2.0216+00	2.8618-02
1.6871-03	1,1654+03	1.9661+00	2,9620-02
1.6016-03	1.2590+03	1.9844+00	2.9800-02
1.5199-03	1 1930+03	1 7207+00	3.1043-02
1.3677-07	1 1209+03	1 5331+00	3 4248-02
1.2970-03	1,1303+03	1,4660+00	3.5708-02

RUNS T9, 10 2-22-66 TABLE E78 CONTINUED

4-3-69

THERMAL NEUTRON SPECTRUM FOR 2.5-INCHES LH2 AT 78 DEGREES USING A BANK OF 32 VERTICAL BF3 DETECTORS ENERGY RESOLUTION FOR GROUPING = .05 LARGEST FRACTIONAL ERROR = .20

NEUTRON ENERGY	NEUTRON FLUX	FLUX*ENERGY	RELATIVE
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)	UNCERTAINTY
1.2298-03	1.0226+03	1.2576+00	3.8340-02
1.1060-03	1.0683+03	1.2456+00	3,9339-02
1.1055-03	1.1456+03	1.2665+00	3.9173-02
1.0431-03	1.0065+03	1.0549+00	4.4263-02
9.9373-04	9.5686+02	9.5086-01	4.5962-02
9.4223-04	8.6832+02	8.1816-01	5.1971-02
8.9351-04	9.1386+02	8.1654-01	5.1233-02
8.4740-04	9.1441+02	7.7487-01	5.3068-02
8.03a1-04	8.1298+02	6.5348-01	5.8197-02
7.6258-04	7.6551+02	5.8376-01	6.4161-02
7.2362-04	6,7362+02	4.8744-01	7.1764-02
6.8680-04	7.7919+02	5.3515-01	6.8151-02
6.5201-04	7.2431+02	4.7226-01	7.4809-02
6.1848-04	6.7860+02	4.1974-01	8.0100-02
5.8686-04	5.6748+02	3.3303-01	9.5678-02
5.5704-04	6.3997+02	3,5649-01	9.4360-02

E-101

OSR MONTE CARLO RESULTS FOR A 2.5-INCH THICK JESS OF LIQUID HYDROGEN NORMALIZED TO ONE SOURCE NEUTRON FOR THE UNCOLLIDED FLUX AT 0 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1.06-12	1.38-05
1.20+07	1.68-12	1.85-05
1.00+07	5.11-12	4.60-05
8.00+06	1.31-11	9.20-05
6.00+06	3.80-11	1.90-04
4.00+06	6.86-11	2.40-04
3.00+0 ,	1.36-10	3.40-04
2.00+06	1.95-10	3-41-04
1.50+06	2.98-10	3.72-04
1.00+06	3.83-10	3.45-04
8.00+05	3,71-10	2.60-04
6.00+05	3.70-10	1.85-04
4.00+05	2.37-10	7.10-05
2.00+05	7.87-11	1.18-05
1.00+05	2.70-11	1.97-06
4.60+04	9.47-12	3.22-07
2.20+04	2.17-12	3.48-08
1.00+04	4.66-11	3.40-07

USR MUNTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE SOURCE NEUTRON FOR THE COLLIDED FLUX AT O DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1 4	0 (0 15	
1.40+07	8.62-15	1.12-07
1.20+07	1.91-14	2.10-07
1.00+07	4.50-14	4.05-07
8.00+06	1.86-13	1.30-06
6.00+06	4.40-13	2.20-06
4.00+06	5.77-13	2.02-06
3.00+06	9.60-13	2.40-06
2.00+06	1.09-12	1.91-06
1.50+06	2.56-12	3.20-06
1.00+06	2.68-12	2.41-06
8.00+05	4.57-12	3.20-06
6.00+05	3.60-12	1.80-06
4.00+05	1.83-12	5.50-07
2.00+05	2.40-12	3.60-07
1.00+05	2.81-12	2.05-07
4.60+04	3.82-12	1.30-07
2.20+04	8.25-12	1.32-07
1.00+04	1.92-11	1.40-07
4.60+03	3.85-11	1.31-07
2.20+03	8.56-11	1.37-07

E-103

OSR MONTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE SOURCE NEUTRON FOR AN ANGLE OF 37 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1.38-17	1-80-10
1.20+07	1.27=16	1.40-09
1.00+07	4.89-16	4.40-09
R.00+06	2 96-15	2.00-09
6.00+06	5 40-15	2.00-08
6.00+06	5.40-15	2.70-08
4.00+06	2.14-14	7.50-08
3.00+06	5.80-14	1.45-07
2.00+06	9.71-14	1.70-07
1.50+06	2.20-13	2.75-07
1.00+06	3.78-13	3.40-07
8.00+05	5.71-13	4.00-07
6.00+05	1.14-12	5.70-07
4.00+05	1.60-12	4.80-07
2.00+05	1.81-12	2.71-07
1.00+05	1.78-12	1.30-07
4.60+04	2.64-12	8.97-08
2.20+04	6.06-12	9.70-08
1.00+04	1,10-11	8-00-08
4.50+03	2.65-11	9-00-08
2.20+03	5.94-11	9-50-08
		7.00 00

OSR MUNTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 78 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	2.00-18	2.60-11
1.20+07	1.09-17	1.20-10
1.00+07	7.44-17	6.70-10
8.00+06	2.14-16	1.50-09
6.00+06	8.00-16	4.00-09
4.00+06	2.03-15	7.10-09
3.00+06	4.44-15	1.11-08
2.00+06	8,63-15	1.51-08
1.50+06	1.77-14	2.21-08
1.00+06	3.18-14	2.86-08
8.00+05	7.84-14	5.49-08
6.00+05	7.20-14	3.60-08

E-105

05R MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR THE UNCOLLIDED FLUX AT O DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	8.08-13	1.05-05
1.20+07	1.45-12	1.60-05
1.00+07	3.89-12	3.50-05
8.00+06	1.00-11	7.00-05
6.00+05	2.58-11	1.29-04
4.00+06	4.60-11	1.61-04
3.00+06	6.96-11	1.74-04
2.00+06	8.69-11	1.52-04
1.50+06	1.24-10	1.55-04
1.00+06	1.33-10	1.20-04
8.00+05	1.01-10	7.10-05
6.00+05	8.80-11	4.40-05
4.00+05	3.67-11	1.10-05
2.00+05	5.67-12	8.50-07
1.00+05	1.06-12	7.75-08
4.60+04	1.79-13	6.08-09
2.20+04	7.56-14	1.21-09
1.00+04	5.89-14	4.30-10

OSR MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR THE COLLIDED FLUX AT 0 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	6.08-15	7.90-08
1.20+07	1.27-14	1.40-07
1.00+07	4.44-14	4-00-07
8.00+06	1.43-13	1.00-06
6.00+06	2.82-13	1.41-06
4.00+06	4.51-13	1.58-06
3.00+06	6.40-13	1.60-06
2.00+06	8.86-13	1.55-06
1.50+06	1.00-12	1.25-06
1.00+06	1,33-12	1.20-06
8.00+05	1,41-12	9.90-07
6.00+05	1.90-12	9.50-07
4.00+05	1.33-12	4.00-07
2.00+05	2.07-12	3.11-07
1.00+05	3.15-12	2.30-07
4.60+04	4.71-12	1.60-07
2.20+04	8.75-12	1.40-07
1.00+04	1.52-11	1.11-07
4.60+03	2.86-11	9.71-08
2.20+03	5.87-11	9.40-08

5R MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN ORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 37 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	3.23-17	4.20-10
1.20+07	7.27-17	8-00-10
1.00+07	3.00-16	2.70-09
8.00+06	1.11-15	7-80-09
6.00+06	6.40-15	3-20-08
4.00+06	2.03-14	7.10-08
3.00+06	4.08-14	1.02-07
2.00+06	1.03-13	1.81-07
1.50+06	1.78-13	2.22-07
1.00+06	2.87-13	2.58-07
8.00+05	4.21-13	2.95-07
6.00+05	4.56-13	2.28-07
4.00+05	6.37-13	1.91-07
2.00+05	1.07-12	1.60-07
1.00+05	1.89-12	1.38-07
4.60+04	3.56-12	1.21-07
2.20+04	5,50-12	8.80-08
1.00+04	1.12-11	8.20-08
4.60+03	2.35-11	8.00-08
2.20+03	4,69-11	7.50-08

05R MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 78 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	6,90-19	8.97-12
1.20+07	1.77-18	1.95-11
1.00+07	3.67-18	3.30-11
8.00+06	1.30-17	9.12-11
6.00+06	2.18-16	1.09-09
4.00+06	3.49-16	1.22-09
3.00+06	1.04-15	2.60-09
2.00+06	2,23-15	3.90-09
1.50+06	4.34-15	5.42-09
1.00+06	9.33-15	8.40-09
8.00+05	1.20-14	8.40-09
6.00+05	1.58-14	7.90-09
4.00+05	4.73-14	1.42-08
2.00+05	1.30-13	1.95-08
1.00+05	5.75-13	4.20-03
4.60+04	9.56-13	3.25-08
2.20+04	4,19-12	6.70-08
1.00+04	8.15-12	5.95-08

05R MONTE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR THE UNCOLLIDED FLUX AT 0 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	4.32-13	5.62-06
1.20+07	6.18-13	6.80-06
1.00+07	1.58-12	1.42-05
8.00+06	3.01-12	2.11-05
6.00+06	5.60-12	2.80-05
4.00+06	6.14-12	2.15-05
3.00+06	6.04-12	1.51-05
2.00+06	3.95-12	6.92-06
1.50+06	3.10-12	3.88-06
1.00+06	1.66-12	1.49-06
8.00+05	6.57-13	4.60-07
6.00+05	2.20-13	1.16-07
4.00+05	1.60-14	4.80-09
2.00+05	1.73-16	2.60-11

OSR MONTE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR THE COLLIDED FLUX AT 0 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	4.93-15	6.41-08
1.20+07	8.64-15	9.50-08
1.00+07	2.33-14	2.10-07
3.00+06	5.86-14	4.10-07
6.00+06	1.16-13	5.80-07
4.00+06	1.43-13	5.00-07
3.00+06	1.48-13	3.70-07
2.00+06	1.43-13	2.50-07
1.50+06	1.68-13	2.10-07
1.00+06	2.01-13	1.81-07
8.00+05	1.36-13	9.50-08
6.00+05	1.54-13	7.70-08
4.00+05	1.37-13	4.10-08
2.00+05	1.72-13	2.58-08
1.00+05	1.78-13	1.30-08
4.60+04	3.24-13	1.10-08
2.20+04	4.63-13	7.41-09
1.00+04	8,36-13	6.10-09
4.60+03	1.85-12	6.30-09
2.20+03	3.62-12	5.80-09

E-111

05R MONTE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 37 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	7.00-18	9.10-11
1.20+07	1,84-17	2.02-10
1.00+07	1.00-16	9.00-10
8.00+06	5.71-16	4.00-09
6.00+06	3,80-15	1.90-08
4.00+06	1.20-14	4.20-08
3.00+06	1.96-14	4.90-08
2.00+06	2,06-14	3.60-08
1.50+06	3.54-14	4.42-08
1.00+06	4.22-14	3.80-08
8.00+05	5.86-14	4.10-08
6.00+05	7.20-14	3.60-08
4.00+05	7.27-14	2.18-08
2.00+05	1.07-13	1.61-08
1.00+05	1.26-13	9.20-09
4.60+04	2.59-13	8.80-09
2.20+04	4.06-13	6.50-09
1.00+04	8.25-13	6.02-09
4.60+03	1.70-12	5.78-09
2.20+03	3.87-12	6.20-09

OFR MONTE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 78 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1,12-18	1.45-11
1.20+07	2,55-18	2.80-11
1.00+07	1.20-17	1.08-10
8.00+06	1.84-17	1.29-10
6.00+06	6.30-17	3.15-10
4.00+06	1.29-16	4.50-10
3.00+06	5.20-16	1.30-09
2.00+06	1.26-15	2.20-09
1.50+06	2.42-15	3.03-09
1.00+06	5.22-15	4.70-09
8.00+05	6.14-15	4.30-09
6.00+05	1.70-14	8.50-09
4.00+05	3.23-14	9.70-09
2.00+05	2.93-14	4.40-09

05R MONTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 5 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	2.15-15	2.80-08
1.20+07	5.55-15	6.10-08
1.00+07	1.33-14	1.20-07
8.00+06	6.43-14	4.50-07
6.00+06	1.16-13	5.80-07
4.00+06	2.49-13	8.70-07
3.00+06	4.04-13	1.01-06
2.00+06	6.29-13	1.10-06
1.50+06	1.60-12	2.00-06
1.00+06	1.03-12	9.30-07
8.00+05	2.29-13	1:60-07
6.00+05	1.70-12	8.50-07
4.00+05	8.33-13	2.50-07

05R MONTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 15 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1•4J+07	7.50-17	9.75-10
1•20+07	6.33-16	6.96-09
1•00+07	7.36-16	6.62-09
8.00+06	8.63-16	6.04-09
6.00+06	1.58-14	7.92-08
4.00+06	2.61-14	9.14-08
3.00+06	1.05-13	2.63-07
2.00+06	1.27-13	2.23-07
1.50+06	2.42-13	3.02-07
1.00+06	3.50-13	3.15-07
8.00+05	2.14-12	1.50-06
6.00+05	9.18-13	4.59-07

OSR MONTE CARLO RESULTS FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 53 DEGREES

UPPER ENERGY	NEUTRON F	LUX	FLUX*ENE	RGY
(EV)	(RELATIVE L	JNITS)	(RELATIVE	UNITS)
1.40+07	8.08-16	1	1.05-1	0
1.20+07	2.18-17		2.40-1	0
1.00+07	6.78-17	,	6.10-1	0
8.00+06	6.86-17	7	4.80-1	0
6.00+06	5,20-16	5	2.60-0	9
4.00+06	2.51-15	5	8.80-0	9
3.00+06	9.60-15	5	2.40-0	8
2.00+06	7.43-15	5	1.30-0	8
1.50+06	2.16-14		2.70-0	8
1.00+06	6.44-14	•	5.80-0	8
8.00+05	8.57-14		6.00-0	8
6.00+05 4.00+05	4 33-13		9.00-0	8
2.00+05	1.07-12		1.60-0	7

.

USR MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 5 DEGREES

	stand the second second second	-
UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1.77-15	2.30-08
1.20+07	3.45-15	3.80-08
1.00+07	1.07-14	9.60-08
8.00+06	3.14-14	2.20-07
b.00+06	8.60-14	4.30-07
4.00+06	2.03-13	7.10-07
3.00+06	3.00-13	7.50-07
5×00+06	3.60-13	6.30-07
1.50+06	4.64-13	5.80-07
1.00+06	3.67-13	3.30-07
8.0u+05	5.00-13	3.50-07
6.0u+05	5.60-13	2.80-07

OSR MONTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 15 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EJ)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1.00-16	1.30-09
1.20+07	2.64-16	2.90-09
1.00+07	5.22-15	4.70-08
8.00+06	9.57-15	6.70-08
6.00+06	2.40-14	1.20-07
4.00+06	7.14-14	2.50-07
3.00+06	1.04-13	2.60-07
2.00+06	1.83-13	3.20-07
1.50+06	2.72-13	3.40-07
1.00+06	4.67-13	4.20-07
8.00+05	5.00-13	3.50-07
6.00+05	7.60-13	3.80-07
4.00+05	1.50-12	4.50-07

USR MUNTE CARLO RESULTS FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 53 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	1.85-18	2.40-11
1.20+07	6.82-18	7.50-11
1.00+07	5.56-17	5.00-10
8.0u+u6	1,36-16	9.50-10
0.00+06	9.80-16	4.90-09
4.00+06	1.94-15	6.80-09
3.00+06	2.60-15	6.50-09
2.00+06	2.17-15	3.80-09
1.50+06	2.64-15	3.30-09
1.00+06	6.78-15	6.10-09
8.00+05	7.14-15	5.00-09
6.00+05	9.20-15	4.60-09

OSR MUNTE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 5 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
1.40+07	2.05-15	2.67-08
1.20+07	2.80-15	3.08-08
1.00+07	7.70-15	6.93-08
8.00+06	2.70-14	1.89-07
6.00+06	1.14-13	5.70-07
4.01+06	7.71-14	2.70-07
3.00+06	5.64-14	1.41-07
2.00+06	9.49-14	1.66-07
1.50+06	9.84-14	1.23-07
1.00+06	7.00-14	6.30-08
8.00+05	7.91-14	5.54-08
0.00+05	7.00-14	3.50-08
4.00+05	9.67-14	2.90-08

USR MONIE CARLO RESULTS FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN NORMALIZED TO ONE NEUTRON FOR AN ANGLE OF 15 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS)	(RELATIVE UNITS)
		the second state of the se
1.40+07	2.52-16	3.27-09
1.20+07	1.24-15	1.36-08
1.00+07	4.67-15	4.20-08
8.00+06	1.27-14	8.90-08
0.00+06	2.32-15	1.16-08
4.00+06	4.91-14	1.72-07
3.00+06	3.92-14	9.80-08
2.00+06	6.86-14	1.20-07
1.50+06	1.39-13	1.74-07
1.00+06	1.88-14	1.69-08
8.00+05	3.10-14	2.17-08
6.00+05	5.66-14	2.83-08
4.00+05	4.07-14	1.22-08

# RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER LNERGY	NEUTRON	FLUX	FLUX*ENE	RGY
(EV)	(RELATIVE	UNITS)	(RELATIVE	UNITS)
9.50-01	8.83-0	2	7.95-0	2
8.50-01	9.74-0	2	7.79-0	2
7.50-01	1.14-0	1	7.97-0	2
6.50-01	1.32-0	1	7.89-0	2
5.50-01	1.61-0	1	8.03-0	2
4.50-01	2.09-0	î	8.38-0	2
3.50-01	2.48-0	1	7.82-0	2
2.80=01	2 08-0	1	7.75-0	2
2.40-01	3 44-0	1	7 7/100	2
2.10-01	3 9/ -0	1	7.69-0	2
2.10-01	4 07-0	1	7.08-0	2
1.40-01	5 82-0	1	7.42-0	2
1.15-01	5.02-0	1	1.72-0	2
1.05-01	6 45-0	4 .	6.75-0	2
1.05-01	7 10-0	1	5.24-0	2
8.85-02	7.18-0	1	5.98-0	2
7.80-02	7.85-0	1	5./5-0	2
6.85-02	9.00-0	1	5.62-0	2
5.65-02	1.15+0	0	5./5-0	2
4.35-02	1.63+0	0	6.46-0	2
3.57-02	2.18+0	0	7.30-0	2
3.13-02	2.86+0	0	8.37-0	2
2.73-02	4.01+0	0	1.00-0	1
2.25-02	6.63+0	0	1.36-0	1
1.84-02	1.00+0	1	1.73-0	1
1.62-02	1.29+0	1	1.96-0	1
1.43-02	1.65+0	1	2.16-0	1
1.13-02	2.17+0	1	2.32-0	1
9.60-03	2.63+0	1	2.38-0	1
8.45-03	2.98+0	1	2.36-0	1
7.40-03	3.33+0	1	2.31-0	1
6.45-03	3.72+0	1	2.25-0	1
5.65-03	4.13+0	1	2.14-0	1
4.70-03	4.58+0	1	1.96-0	1
3.85-03	4.79+0	1	1.74-0	1
3.40-03	5.07+0	1	1.59-0	1
2.85-03	5.16+0	1	1.34-0	1
2.35-03	5.06+0	1	1.09-0	1
1.95-03	4.95+0	1	8.79-0	2
1.60-03	4.53+0	1	6.79-0	2
1.40-03	4.61+0	1	5.88-0	2
1.10-03	4.30+0	1	4.40-0	2
9.00-04	3.81+0	1	3.05-0	2
7.00-04	3.42+0	1	2.05-0	2

E-122

CONTI JULL

RESULIS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER ENERGY NEUTRON FLUX FLUX\*ENERGY (EV) (RELATIVE UNITS) (RELATIVE UNITS)

5.00-04	2.94+01	1.17-02
3.00-04	2.40+01	4.80-03
1.00-04	1.17+01	5.84-04

RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

UPPER ENERGY	NEUTRON	FLUX	FLUX*ENERGY	
(Ey)	(RELATIVE	UNITS)	(RELATIVE UNITS)	
9.50-01	8.89-0	2	8.00-02	
8.50-01	9.79-0	2	7.83-02	
7.50-01	1.14-0	01	8.01-02	
6.50-01	1.32-0	01	7.92-02	
5.50-01	1.61-0	01	8.07-02	
4.50-01	2.11-0	01	8.43-02	
3.50-01	2.49-0	01	7.84-02	
2.80-01	2.99-0	01	7.79-02	
2.40-01	3.46-0	01	7.79-02	
2.10-01	3.97-0	01	7.74-02	
1.80-01	4.87-0	01	7.80-02	
1.40-01	5.87-0	01	7.48-02	
1.15-01	6.17-0	01	6.79-02	
1.05-01	6.47-0	01	6.26-02	
8.83-02	7.20-0	1	5.99-02	
7.83-02	7.87-0	1	5.76-02	
6.80-02	9.02-0	1	5.64-02	
5.65-02	1.16+0	0	5.79-02	
4.30-02	1.66+0	0	6.56-02	
3.57-02	2.25+0	0	7.53-02	
3.13-02	3.00+0	0	8.80-02	
2.73-02	4.38+0	0	1.09-01	
2.25-02	8.05+0	0	1.65-01	
1.84-02	1.42+0	1	2.45-01	
1.62-02	2.02+0	1	3.08-01	
1.43-02	2.81+0	1	3.67-01	
1.13-02	3.93+0	1	4.21-01	
9.60-03	4.88+0	1	4.40-01	
8.40-03	5.51+0	1	4.37-01	
7.40-03	6.10+0	1	4.22-01	
6.45-03	6.72+0	1	4.07-01	
5.65-03	7.35+0	1	3.80-01	
4.70-03	7.96+0	1	3.40-01	
3.80-03	8.12+0	1	2.94-01	
3.40-03	8.46+0	1	2.64-01	
2.85-03	8.37+0	1	2.18-01	
2.35-03	7.98+0	1	1.71-01	
1.95-03	7.59+0	1	1.35-01	
1.60-03	6.73+0	1	1.01-01	
1.40-03	6.73+0	1	8.57-02	
1.103	6.09+0	1	6.24-02	
9.00-04	5.21+0	1	4.17-02	
7.00-04	4.51+0	1	2.70-02	

CONTI, ULD

RESULTS OF IDE TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 2.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

(EV) (RELATIVE UNITS) (RELATIVE UNITS)

5.00-04	3.70+01	1.48-02
3.00-04	2.85+01	5.70-03
1.00-04	1.33+01	6.64-04

RESULTS OF IDE TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER ENERGY	NEUTRON FLUX	FLUX*ENERGY
(EV)	(RELATIVE UNITS	S) (RELATIVE UNITS)
9.50-01	5.93-02	5.34-02
8.50-01	6.60-02	5.28-02
7.50-01	7.81-02	5.46-02
6.50-01	9.15-02	5.49-02
5.50-01	1.14-01	5.69-02
4.50-01	1.52-01	6.07-02
3.50-01	1.82-01	5.72-02
2.80-01	2,22-01	5.76-02
2.40-01	2.59-01	5.84-02
2.10-01	2,99-01	5.84-02
1.80-01	3.71-01	5.93-02
1.40-01	4.48-01	5.72-02
1.15-01	4.70-01	5.17-02
1.05-01	4.92-01	4.76-02
8.85-02	5.45-01	4.54-02
7.80-02	5.95-01	4.36-02
6.85-02	6.83-01	4.27-02
5.65-02	8.79-01	4.39-02
4.35-02	1.27+00	5.04-02
3.57-02	1.76+00	5.89-02
3.13-02	2.40+00	7.03-02
2.73-02	3.60+00	8.95-02
2.25-02	6.87+00	1.40-01
1.84-02	1,21+01	2.10-01
1.62-02	1.70+01	2.60-01
1.43-02	2.33+01	3.04-01
1.18-02	3.20+01	3.42-01
9.60-03	3,96+01	3.57-01
8.45-03	4.49+01	3.56-01
7.40-03	4.99+01	3.45-01
6.45-03	5.53+01	3.35-01
5.65-03	6.08+01	3.15-01
4.70-03	6.64+01	2.84-01
3.85-03	6.84+01	2.48-01
3.40-03	7.15+01	2.24-01
2.85-03	7.14+01	1.86-01
2.35-03	6.86+01	1.47-01
1.95-03	6.57+01	1.17-01
1.60-03	5.88+01	8.82-02
1.40-03	5,90+01	7.52-02
1.15-03	5.38+01	5.51-02
9.00-04	4.63+01	3.71-02
7.00-04	4.03+01	2.42-02

CONTI JUED

RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER ENERGY	NEUTRON	FLUX	FLUX*EN	RGY
(EV)	(RELATIVE	UNITS)	(RELATIVE	UNITS)

5.00-04	3.32+01	1.33-02
3.00-04	2,56+01	5.11-03
1.00-04	1.18+01	5.92-04

RESULTS OF IDE TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

UPPER ENERGY	NEUTRON	FLUX	FLUX*EN	RGY
(EV)	(RELATIVE	UNITS)	(RELATIVE	UNITS)
9.50-01	5.42-0	02	4.88-0	20
8.50-01	6.00-0	2	4.80-0	2
7.50-01	7.05-0	)2	4.93-0	2
6.50-01	8.20-0	2	4.92-0	2
5.50-01	1.01-0	01	5.05-0	2
4.50-01	1.33-0	01	5.34-0	2
3.50-01	1.59-0	)1	5.02-0	2
2.80-01	1.93-0	1	5.03-0	2
2.40-01	2.25-0	1	5.07-0	2
2.10-01	2.60-0	)1	5.06-0	2
1.80-01	3.21-0	1	5.14-0	2
1.40-01	3.89-0	1	4.96-0	2
1.15-01	4.11-0	1	4.52-0	2
1.05-01	4.32-0	1	4.18-0	2
8.80-02	4.81-0	1	4.00-0	2
7.80-02	5.27-0	1	3.86-0	2
6.80-02	6.05-0	1	3.78-0	2
5.65-02	7.79-0	1	3.90-0	2(
4.35-02	1.12+0	0	4.45-0	2
3.57-02	1.54+0	0	5.15-0	2
3.13-02	2.08+0	0	6.09-0	)2
2.75-02	3.09+0	0	7.69-0	)2
2.25-02	5.96+0	0	1.22-0	)1
1.84-02	1.13+0	1	1.96-0	1
1.62-02	1.71+0	1	2.61-0	)1
1.43-02	2.52+0	1	3.28-0	1
1.13-02	3.71+0	1	3.97-0	)1
9.60-03	4.73+0	1	4.27-0	)1
8.45-03	5.38+0	1	4.26-0	1
7.40-03	5.96+0	1	4.13-0	1
6.45-03	6.58+0	1	3.98-0	1
5.65-03	7.18+0	1	3.72-0	1
4.70-03	7.75+0	1	3.31-0	1
3.85-03	7.87+0	1	2.85-0	1
3.40-03	8.17+0	1	2.55-0	1
2.85-03	8.03+0	1	2.09-0	1
2.35-03	7.58+0	1	1.63-0	1
1.95-03	7.16+0	1	1.27-0	1
1.60-03	6.30+0	1	9.45-0	2
1.40-03	6.26+0	1	7.98-0	2
1.15-03	5.62+0	1	5.76-0	2
9.00-04	4.76+0	1	3.81-0	2
7.00-04	4.07+0	1	2.44-0	2

CONTI JULD

RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 4.5-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

UPPER ENERGY	PER ENERGY NEUTRON FLUX		FLUX*ENERGY	
(EV)	(RELATIVE	UNITS)	(RELATIVE	UNITS)

5.00-04	3.30+01	1.32-02
3.00-04	2.50+01	5.01-03
1.00-04	1.15+01	5.77-04

RESULTS OF IDE TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER ENERGY	NEUTRON FL	UX FLU	JX*ENE	RGY
(EV)	(RELATIVE UN	ITS) (RELA	TIVE	UNITS)
0.50-01	3 50-07	-	20-0	
9.50-01	3.58-03		0.22-0	3
8.50-01	4.02-03		5.22-0	3
7.50-01	4.80-03		5.36-0	5
6.50-01	5.70-03		3.42-0	5
5.50-01	7.20-03		5.60-0	3
4.50-01	9.78-03		5.91-0	3
3.50-01	1.18-02	-	5. 72-0	3
2.80-01	1.46-02	-	5.80-0	3
2.40-01	1.73-02		5.89-0	3
2.10-01	2.01-02		5.92-0	3
1.80-01	2.51-02	4	.02-0	3
1.40-01	3.05-02		.89-0	3
1.15-01	5.19-02	3	.50-0	3
1.05-01	3.31-02	- 3	.21-0	3
8.85-02	3.66-02	1.1.2.2.00	.05-0	3
7.80-02	4.16-02		.05-0	3
6.85-02	4.68-02	2	.93-0	3
5.65-02	5.73-02	2	.86-0	3
4.35-02	7.49-02	2	.97-0	3
3.57-02	1.04-01	3	.47-0	3
3.13-02	1.43-01	4	.19-0	3
2.73-02	2.09-01	5	.21-0	3
2.25-02	3.50-01	7	.16-0	3
1.84-02	8.19-01	1	•42-0	2
1.62-02	2.05+00	3	.13-0	2
1.43-02	3.99+00	5	.21-0	2
1.18-02	7.38+00	7	.90-0	2
9.60-03	1.25+01	1	.13-0	1
8.45-03	1.67+01	1	.33-0	1
7.40-03	1.96+01	1	.36-0.	1
6.45-03	2.21+01	- 1	.33-0	1
5.65-03	2.49+01	1	.29-0	1
4.70-03	2.80+01	1	.20-0	1
3.85-03	2.90+01	1	.05-0	1
3.40-03	2.83+01	8	.85-0	2
2.85-03	2,99+01	7	.78-0	2
2.35-03	2.86+01	6	.16-0	2
1.95-03	2.61+01	4	.63-02	2
1.60-03	2.32+01	3	.48-02	2
1.40-03	1,95+01	2	.48-02	2
1.15-03	1.99+01	2	.04-02	2
9.00-04	1.77+01	1	.42-02	2
7.00-04	1.49+01	8	.91-03	3

CONT1 UED

RESULTS OF IDE TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 38 DEGREES

UPPER ENERGY NEUTRON FLUX FLUX\*ENERGY (EV) (RELATIVE UNITS) (RELATIVE UNITS)

5.00-04	1.36+01	5.45-03
3.00-04	1.38+01	2.77-03
1.00-04	1.95+01	9.73-04

RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

UPPER ENERGY	NEUTRON	FLUX	FLUX*EN	ERGY
(EV)	(RELATIVE	UNITS)	(RELATIVE	UNITS)
			_	
9.50-01	3.01-0	3	2.70-	03
8.50-01	3.34-0	3	2.67-	03
7.50-01	3,95-0	3	2.76-	03
6.50-01	4.62-0	3	2.77-	03
5.50-01	5.73-0	3	2.87-	03
4.50-01	7.64-0	3	3.06-	03
3.50-01	9.20-0	3	2.90-	03
2.80-01	1.12-0	2	2.92-	03
2.40-01	1.32-0	2	2.97-	03
2.10-01	1.53-0	2	2.98-	03
1.80-01	1.90-0	2	3.03-	03
1.40-01	2.32-0	2	2.95-	03
1.15-01	2.45-0	2	2.70-	03
1.05-01	2.58-0	2	2.49-	03
8.85-02	2.87-0	2	2.39-	03
7.80-02	3.15-0	2	2.31-	03
6.85-02	3.62-0	2	2.26-	03
5.65-02	4.68-0	2	2.34-	03
4.35-02	6.79-0	2	2.69-	03
3.57-02	9.37-0	2	3.14-1	03
3.13-02	1.28-0	1	3.76-	03
2.73-02	1.96-0	1	4.87-	03
2.25-02	4.09-0	1	8.37-	03
1.84-02	9.03-0	1	1.56-	02
1.62-02	1.57+0	0	2.40-	02
1.43-02	2.70+0	0	3.52-	02
1.18-02	4.80+0	0	5.13-0	02
9.60-03	6.94+0	0	6.26-	02
8.45-03	8.36+0	0	6.62-	02
7.40-03	9,67+0	0	6.70-0	02
6.45-03	1.10+0	1	6.65-	SC
5.65-03	1.23+0	1	6.34-0	02
4.70-03	1.35+0	1	5.76-0	02
3.85-03	1.38+0	1	5.00-0	02
3.40-03	1.43+0	1	4.47-0	2
2.85-03	1.40+0	1	3.64-0	202
2.35-03	1.31+0	1	2.82-0	202
1.95-03	1.23+0	1	2.18-0	202
1.60-03	1.07+0	1	1.60-0	2
1.40-03	1.05+0	1	1.54-0	2
1.15-03	9.31+0	0	9.54-0	3
9.00-04	7.76+0	0	6.21-0	13
7.00-04	6,55+0	0	3.93-0	3

CONTI ULD

RESULTS OF IDF TRANSPORT CALCULATIONS OF THERMAL NEUTRON FLUX FOR A 13.0-INCH THICKNESS OF LIQUID HYDROGEN AT AN ANGLE OF 77.4 DEGREES

JPPER ENERGY	NEUTRON	FLUX	FLUX*ENERGY
(EV)	(RELATIVE	UNITS)	(RELATIVE UNITS)

5.00-04	5.21+00	2.09-03
3.00-04	3.88+00	7.76-04
1.00-04	1.76+00	8.81-05